# UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

# FORM 8-K

## CURRENT REPORT

Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

Date of Report (Date of earliest event reported): February 5, 2024

# AYALA PHARMACEUTICALS, INC.

(Exact name of registrant as specified in its charter)

Delaware	001-36138	02-0563870
(State or other jurisdiction	(Commission	(IRS Employer
of incorporation)	File Number)	Identification No.)
9 Deer Park Drive, Suite K-1		
Monmouth Junction, NJ		08852
(Address of principal executive offices)		(Zip Code)

Registrant's telephone number, including area code: (609) 452-9813

(Former name or former address, if changed since last report.)

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions (*see* General Instruction A.2. below):

□ Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)

□ Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)

□ Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))

□ Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

Securities registered pursuant to Section 12(b) of the Act: None

Indicate by check mark whether the registrant is an emerging growth company as defined in as defined in Rule 405 of the Securities Act of 1933 (§230.405 of this chapter) or Rule 12b-2 of the Securities Exchange Act of 1934 (§240.12b-2 of this chapter).

Emerging growth company  $\Box$ 

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act.  $\Box$ 

#### Item 1.01. Entry into a Material Definitive Agreement.

#### The Asset Purchase Agreement

On February 5, 2024, (the "Execution Date") Ayala Pharmaceuticals, Inc., a Delaware corporation (the "Company"), and Immunome, Inc. ("Purchaser"), entered into an Asset Purchase Agreement (the "Asset Purchase Agreement") pursuant to which Purchaser will acquire certain of the Company's assets and liabilities related to its AL101 and AL102 programs (the "Asset Sale"), which constitute substantially all of the Company's assets. Pursuant to the Asset Purchase Agreement, Purchaser will pay to the Company an aggregate purchase price of \$20,000,000 in cash, subject to certain adjustments (the "Closing"), 2,175,489 shares (the "Shares") of Purchaser's common stock, \$0.0001 par value (the "Common Stock"), issuable at Closing and up to \$37,500,000 in cash due upon the achievement of certain development and commercial milestone events set forth in the Asset Purchase Agreement.

The Asset Purchase Agreement, the Asset Sale and the other transactions contemplated by the Asset Purchase Agreement have been approved by the board of directors of the Company (the "Board") and is subject to receipt of approval of the stockholders of Ayala. Pursuant to the Asset Purchase Agreement, holders of more than a majority of the issued and outstanding shares of capital stock of the Company (the "Consenting Shareholders") are to execute and deliver to the Company, within three business days after the Execution Date, a written consent (the "Written Consent") authorizing, approving and adopting the Asset Purchase Agreement and the transactions contemplated thereby, including the Asset Sale and other transactions contemplated by the Asset Purchase Agreement and the related transaction documents as defined therein.

The Asset Purchase Agreement contains customary representations, warranties, conditions and covenants, including covenants (i) concerning the conduct of business by the Company prior to the Closing, (ii) prohibiting the Company and its representatives from soliciting, initiating or knowingly inducing, encouraging or facilitating any competing acquisition proposal, (iii) prohibiting the Company and its controlled affiliates from competing with Purchaser for five years following the Closing in certain fields, and (iv) restricting the Company's ability to make distributions to stockholders, dissolve or wind up its business or file for bankruptcy for six months following the Closing.

The Company will prepare and file an information statement in respect of the Asset Sale and the other transactions contemplated by the Asset Purchase Agreement (the "Information Statement") with the U.S. Securities and Exchange Commission (the "SEC") in accordance with Rule 14c-2 promulgated under the Exchange Act of 1934, as amended.

Each party's obligation to consummate the Asset Sale is also conditioned upon certain other customary closing conditions, including the accuracy of the other party's representations and warranties as of the Closing, subject, in certain instances, to certain materiality and other thresholds, the performance by the other party of its obligations and covenants under the Asset Purchase Agreement in all material respects, the lapse of at least twenty (20) calendar from the date the Company mails the definitive Information Statement to its stockholders, receipt of certain third party consents, the delivery of certain related ancillary documents by the other party, and the absence of any injunction or other legal prohibitions preventing consummation of the Asset Sale.

The representations, warranties and covenants of the parties do not survive the Closing. Purchaser has the right to recover certain liabilities that exceed specified thresholds by set off against future milestone payments.

The Purchase Agreement also provides that until the six-month anniversary of the Closing, the Company will hold and not sell 50% of the Shares, subject to certain exceptions. Further, the Company has agreed, subject to certain exceptions, that until the one-year anniversary of the Closing, any transfer of the Shares by the Company that exceed 15% of the average daily trading volume of the Common Stock over the five-trading day period ending on the trading day immediately prior to such trading date shall be made pursuant to a block trade or other disposition through a market participant designated by Purchaser.

Purchaser has agreed to use its commercially reasonable efforts to (x) file a resale registration statement with the SEC registering the Shares for resale on or before the date that is seven days following the earlier of (i) April 1, 2024 and (ii) the date Purchaser files its annual report on Form 10-K for the year ended December 31, 2023 and (y) cause such resale registration statement to be declared effective as soon as practicable after the filing thereof but no later than 90 calendar days after the filing thereof or by five trading days from when the Company is notified that the SEC will not review the resale registration statement or that it will not be subject to further review.

The Asset Purchase Agreement also provides that if after six months from closing, the Company adopts a plan of dissolution, it may distribute all of Shares it still holds and its contractual rights to the contingent milestone payments through contingent value rights.

#### Support Agreement

In connection with the execution of the Asset Purchase Agreement, certain of the Company's officers, directors and stockholders who collectively are the record or beneficial holders of more than a majority of the issued and outstanding shares of capital stock of the Company entered into a stockholder support agreement in favor of Purchaser (the "Support Agreement") providing, among other things, that such officers, directors and stockholders will, among other things, (i) deliver the Written Consent; (ii) vote against any proposal made in opposition to, or in competition with, the Purchase Agreement or the Asset Purchase and (iii) vote against any acquisition proposal involving a third party. The Written Consent must be duly executed and delivered to the Company by the stockholders who are party to the Support Agreement by 11:59 p.m. Eastern Time on February 8, 2024.

The Purchase Agreement and form of Support Agreement have been included as exhibits hereto solely to provide investors with information regarding their terms. Neither is intended to be a source of financial, business or operational information about the Company or Purchaser. The representations, warranties and covenants contained in the Purchase Agreement and Support Agreement were made only for the purposes of the Purchase Agreement and Support Agreement as of the dates specified therein and solely for the benefit of the parties thereto. In addition, the representations, warranties and covenants contained in the Purchase Agreement and Support Agreement may be subject to qualifications and limitations agreed upon by the parties in connection with negotiating the terms thereof, including the Company's representations, warranties and covenants in the Purchase Agreement being qualified by disclosure schedules made for the purpose of allocating contractual risk amongst the parties as opposed to establishing such matters as facts, and may further be subject to certain standards of materiality applicable to the parties that differ from those applicable to investors. As a result, investors should not rely on the representations, warranties and covenants included in the Purchase Agreement, or any descriptions thereof, as characterizations of the actual state of facts or condition of the Company or Purchaser and each of their respective business. Moreover, information concerning the subject matter of the representations and warranties may change after the date of the Purchase Agreement and Support Agreement, which subsequent information may or may not be fully reflected in public disclosures.

The foregoing description of the terms of the Purchase Agreement and Ayala Support Agreement are not complete and are qualified in their entirety by reference to the Purchase Agreement and form of Ayala Support Agreement, copies of which are filed as Exhibit 2.1 and Exhibit 10.1 to this Current Report on Form 8-K and incorporated herein by reference.

# Item 5.02 Departure of Directors or Certain Officers; Election of Directors; Appointment of Certain Officers; Compensatory Arrangements of Certain Officers.

On February 5, 2024, the Company's Compensation Committee approved the payment of transaction bonuses, payable in cash and contingent upon closing of the Asset Purchase to each of the following executives officers of the Company in the applicable respective amounts: Mr. Ken Berlin, President and Chief Executive Officer, \$1,600,000; Mr. Roy Golan, Chief Financial Officer, \$450,000; and Dr. Andres Gutierrez, Chief Medical Officer and Executive Vice President, \$800,000.

## Item 7.01. Regulation FD Disclosure.

On February 6, 2024, the Company and Purchaser issued a joint press release announcing the execution of the Asset Purchase Agreement. A copy of that press release is furnished as Exhibit 99.1 to this Current Report on Form 8-K and incorporated herein by reference.

Following the consummation of the Asset Purchase, the Board expects to review alternatives for the Company's cash and public company status.

In accordance with General Instruction B.2 of Form 8-K, the foregoing information, including Exhibit 99.1, shall not be deemed "filed" for the purposes of Section 18 of the Exchange Act or otherwise subject to the liabilities of that section, nor shall such information, including Exhibit 99.1, be deemed incorporated by reference into any filing under the Securities Act of 1933, as amended, or the Exchange Act, except as shall be expressly set forth by specific reference in such filing.

## Item 9.01. Financial Statements and Exhibits.

(d) Exhibits.

Exhibit No.	Description
2.1+†	Asset Purchase Agreement, dated as of February 5, 2024, by and between Ayala Pharmaceuticals, Inc. and Immunome, Inc.
10.1	Form of Support Agreement, dated February 5, 2024.
99.1	Press release dated February 6, 2024
104	Cover Page Interactive Data File (embedded within the Inline XBRL document)

+ Certain schedules and exhibits to this agreement have been omitted in accordance with Item 601(a)(5) of Regulation S-K. A copy of any omitted schedule and/or exhibit will be furnished to the Securities and Exchange Commission on request.

† Certain confidential information contained in this document, marked by \*\*\*, has been omitted pursuant to Item 601(b)(10)(iv) of Regulation S-K because it is both (i) not material and (ii) the type of information that the registrant treats as private or confidential.

#### **Forward-Looking Statements**

Certain statements contained in this Form 8-K may be considered forward-looking statements that involve a number of risks and uncertainties, including statements regarding the future conduct of our studies and the potential efficacy and success of product candidates. Forward-looking statements generally include statements that are predictive in nature and depend upon or refer to future events or conditions, and include words such as "may," "will," "should," "would," "expect," "anticipate," "plan," "likely," "believe," "estimate," "project," "intend," and other similar expressions among others. Statements that are not historical facts are forward-looking statements. Forward-looking statements are based on current beliefs and assumptions that are subject to risks and uncertainties and are not guarantees of future performance. Actual results could differ materially from those contained in any forward-looking statement as a result of various factors, including, without limitation: the risk that the Asset Sale will not be completed; potential litigation relating to the proposed transaction that could be instituted against the Company, Purchaser or their respective directors; possible disruptions from the proposed transaction that could harm our and/or Purchaser's respective businesses; the ability of the Company and Purchaser to retain, attract and hire key personnel; potential adverse reactions or changes to relationships with customers, employees, suppliers or other parties resulting from the announcement or completion of the proposed transaction; the Company's levels of available cash and our need to raise additional capital; legislative, regulatory and economic developments; unpredictability and severity of catastrophic events, including, but not limited to, acts of terrorism or outbreak of war or hostilities, as well as management's response to any of the aforementioned factors; and such other factors as are set forth in the Company's periodic public filings with the SEC, including but not limited to those described under the heading "Risk Factors" in the Form 10-K for the fiscal year ended December 31, 2022 of Old Ayala, Inc. (f/k/a Ayala Pharmaceuticals, Inc.) and the Form 10-K for the fiscal year ended October 31, 2022 of Ayala Pharmaceuticals, Inc. (f/k/a Advaxis, Inc.) ("Ayala" or "we," "us" or "our"), and such entities' periodic public filings with the SEC. Except as required by applicable law, we undertake no obligation to revise or update any forward-looking statement, or to make any other forward-looking statements, whether as a result of new information, future events or otherwise.

#### Additional Information and Where to Find It

In connection with the proposed Asset Sale, the Company intends to file relevant materials with the SEC, including the Information Statement in preliminary and definitive form. The Company's stockholders are strongly advised to read all relevant documents filed by the Company with the SEC, including the Information Statement, because they will contain important information about the proposed transaction. These documents will be available at no charge on the SEC's website at www.sec.gov. In addition, documents will also be available without charge by visiting the the Company's website at www.ayalapharma.com.

#### SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

February 6, 2024

## AYALA PHARMACEUTICALS, INC.

By: /s/ Kenneth A. Berlin

Name: Kenneth A. Berlin Title: President and Chief Executive Officer

# ASSET PURCHASE AGREEMENT

by and between

Ayala Pharmaceuticals, Inc.

and

IMMUNOME, INC.

Dated as of February 5, 2024

# TABLE OF CONTENTS

Page

1.	DEFINITI	ONS.	1
2.	PURCHAS	se and Sale of Assets	20
	2.1	Acquired Assets.	20
	2.2	Assumed Liabilities	21
	2.3	Excluded Liabilities	21
	2.4	Assets Incapable of Transfer	21
3.	CONSIDE	RATION	21
	3.1	Consideration	21
	3.2	Closing Consideration.	22
	3.3	Milestone Consideration	22
	3.4	Withholding	25
	3.5	Legends	25
	3.6	Lock Up	27
	3.7	Orderly Market	27
	3.8	Rule 144 Reporting.	27
4.	CLOSING		28
	4.1	Closing	28
	4.2	Deliveries at Closing.	28
	4.3	Possession	29
5.	Represe	NTATIONS AND WARRANTIES	29
	5.1	Representations and Warranties of Seller	29
	5.2	Representations and Warranties of Purchaser	45
6.	Additio	NAL AGREEMENTS	49
	6.1	Public Disclosures	49
	6.2	Confidentiality	49
	6.3	Conduct Pending the Closing	49
	6.4	Further Assurances and Cooperation; Non-Compete.	50
	6.5	Tax Matters	51
	6.6	Seller Stockholder Approval	52
	6.7	Non-Solicitation.	54

# TABLE OF CONTENTS (continued)

			Page
	6.8	Deposit	54
	6.9	Restrictions on Dissolution and Winding Up	55
	6.10	Tail Insurance	55
	6.11	Transition Services	55
	6.12	Employee Matters	55
7.	Condit	TIONS PRECEDENT TO OBLIGATIONS OF PURCHASER.	56
	7.1	Accuracy of Representations and Warranties	56
	7.2	Performance of Covenants	56
	7.3	No Restraints	56
	7.4	No Material Adverse Effect	56
	7.5	No Governmental Litigation	56
	7.6	Closing Certificate	56
	7.7	Seller Closing Deliverables	56
	7.8	Seller Information Statement	57
8.	Condit	TIONS PRECEDENT TO OBLIGATION OF SELLER	57
	8.1	Accuracy of Representations	57
	8.2	Performance of Covenants	57
	8.3	No Restraints	57
	8.4	Closing Certificate	57
	8.5	Purchaser Closing Deliverables	57
9.	SURVIV	/AL.	57
	9.1	Survival of Representations; Warranties and Covenants	57
10.	REGIST	TRATION RIGHTS.	57
	10.1	Definitions	57
	10.2	Registration Procedures.	58
	10.3	Rule 415; Cutback	60
	10.4	Registration Right Indemnification.	61
	10.5	Prospectus Suspension	62
11.	Termin	NATION	62
	11.1	Termination of the Agreement	62
	11.2	Effect of Termination	64

## TABLE OF CONTENTS (continued)

			Page
12.	MISCELL	ANIFOLIS	64
12.	IVIISCEEL	Antous	01
	12.1	Governing Law; Jurisdiction	64
	12.2	Specific Performance	64
	12.3	WAIVER OF JURY TRIAL	64
	12.4	Entire Agreement; Severability	65
	12.5	Incorporation by Reference	65
	12.6	Amendments and Waivers	65
	12.7	Notices	65
	12.8	No Assignment; Binding Effect	66
	12.9	Third Person Beneficiaries	67
	12.10	Relationship of the Parties	67
	12.11	Headings; Interpretation	67
	12.12	Counterparts; Signatures	67
	12.13	Expenses	67
Exhibits			
Exhibit .	A	– Support Agreement	
Exhibit 1	В	<ul> <li>Assignment and Assumption Agreement</li> </ul>	
Exhibit	С	– Bill of Sale	
Exhibit 1	D	– CVR Agreement	
Exhibit 1	E	<ul> <li>IP and Patent Assignment Agreement</li> </ul>	
Exhibit	F	<ul> <li>Stockholder Written Consent</li> </ul>	
Schedul	ÆS		
Schedul	e A	<ul> <li>Requisite Holders/Consenting Stockholders</li> </ul>	
Schedul		– Compounds	
Schedul		<ul> <li>Product Patents</li> </ul>	
Schedul		– Dropped Patents	
Schedul		<ul> <li>Knowledge Parties</li> </ul>	
Schedul		<ul> <li>Qualified CMC Expenses</li> </ul>	
	e 2.1(a)(i		
Schedul		<ul> <li>Certain Assumed Liabilities</li> </ul>	
	e 4.2(a)(v)		
Schedul		– Transition Services	

-iii-

#### ASSET PURCHASE AGREEMENT

THIS ASSET PURCHASE AGREEMENT (this "Agreement") is made as of February 5, 2024 (the "Agreement Date"), by and between Ayala Pharmaceuticals, Inc., a Delaware corporation ("Seller"), and Immunome, Inc., a Delaware corporation ("Purchaser"). Seller and Purchaser are sometimes referred to herein individually as a "Party" and collectively as the "Parties".

#### RECITALS

A. The Parties intend to provide for the purchase by Purchaser of the Acquired Assets, including the assumption of certain Liabilities relating to the Acquired Assets, all on the terms and subject to the conditions set forth in this Agreement.

B. Subject to, and in accordance with, this Agreement, the Parties desire to enter into (i) an Assignment and Assumption Agreement and (ii) a Bill of Sale.

C. Concurrently with the execution and delivery of this Agreement and as a condition and inducement to Purchaser's willingness to enter into this Agreement, the Persons identified as "Requisite Holders" on Schedule A (such equity holders, the "Requisite Holders") have entered into Support Agreements, dated as of the date of this Agreement, in substantially the form attached hereto as Exhibit A (the "Support Agreement"), pursuant to which the Requisite Holders have, subject to the terms and conditions set forth therein, agreed to vote all their shares of Seller's capital stock in favor of the Transactions.

Now, THEREFORE, in consideration of the premises and of the respective representations, warranties, covenants, agreements, and conditions contained herein, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, and intending to be legally bound hereby, the Parties agree as follows:

#### AGREEMENT

#### **1. DEFINITIONS.**

In this Agreement and any Exhibit, Disclosure Schedule, and Schedule attached hereto, the following terms have the meanings specified or referred to in this Section 1 and shall be equally applicable to both the singular and plural forms.

"Acquired Assets" has the meaning set forth in Section 2.1(a).

#### "Acquired Business Contracts" has the meaning set forth in Section 2.1(a)(ii).

"Affiliate" means, as to any Person, any other Person that, directly or indirectly through one or more intermediaries, controls, is controlled by, or is under common control with that Person, but only for so long as such control exists. For purposes of this definition, "control" (including, with correlative meanings, the terms "controlled by" and "under common control with"), as used with respect to any Person or group of Persons, means possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of the Person, whether (a) through direct or indirect beneficial ownership of at least 50% (or such lesser percentage which is the maximum allowed to be owned by a foreign entity in a particular jurisdiction) of the voting stock or other ownership interest in such corporation or other entity, or (b) by contract.

"Agreement" has the meaning set forth in the preamble.

"Allocation Schedule" has the meaning set forth in Section 6.5(a).

"Annual Net Sales" means, with respect to a Product, total worldwide Net Sales of such Product during a calendar year.

"Anti-Corruption Laws" means the Foreign Corrupt Practices Act of 1977, the Anti-Kickback Act of 1986 or any applicable Laws of similar effect, and the related regulations and published interpretations thereunder.

"Assignment Agreements" means any agreement with Third Parties (other than an Inventor) assigning right, title, or interest to Seller Patents.

"Assignment and Assumption Agreement" means the Assignment and Assumption Agreement in substantially the form attached hereto as Exhibit B.

"Assumed Liabilities" has the meaning set forth in Section 2.2.

"Bill of Sale" means the Bill of Sale in substantially the form attached hereto as Exhibit C.

"BMS" means Bristol-Myers Squibb Company or its successors or assigns to the BMS License.

"BMS License" means the License Agreement dated November 29, 2017, between Seller and BMS, as amended.

"Business Day" means any day that is not a Saturday, a Sunday or other day on which banks are required or authorized by Law to be closed in the Commonwealth of Pennsylvania or the State of New Jersey.

"Closing" has the meaning set forth in Section 4.1.

"Closing Consideration" has the meaning set forth in Section 3.2(b).

"Closing Date" has the meaning set forth in <u>Section 4.1</u>.

"Closing Cash Payment" has the meaning set forth in Section 3.2(a).

"Code" means the Internal Revenue Code of 1986.

"Combination Product" means a product in which one or more active ingredients that are not Products are sold in combination with, in addition to, or in a bundle with, a Product. Such other active ingredient(s) are referred to as the "Other Product(s)".

"Commercialization" means the conduct of all activities undertaken in preparation for and following Regulatory Approval relating to the promotion, sale (including receiving, accepting, and filling product orders), marketing, and distribution (including importing, exporting, transporting, customs clearance, warehousing, invoicing, handling, and delivering) of a product, including sales force, detailing, advertising, market research, market access (including price and reimbursement activities) and sales force training. "Commercialize" and "Commercializing" have correlative meanings.

"Competing Product" has the meaning set forth in Section 6.4(c).

"Compounds" means the gamma secretase inhibitors known as AL101 and AL102, as set forth on Schedule B.

"Consenting Stockholder" means the Persons identified as "Consenting Stockholders" on Schedule A.

"Conflict" has the meaning set forth in Section 5.1(c).

"Consideration" has the meaning set forth in Section 3.1.

"Contract" means any legally binding written or oral contract, agreement, or instrument, including supply contracts, licenses, understandings or commitments, customer agreements, subcontracts, leases of personal property, notes, guarantees, pledges, and conditional sales agreements to which the Person referred to is a party or by which any of its assets are bound.

"Control" or "Controlled by" means, with respect to any Intellectual Property Rights, the possession by a party of the ability (whether by ownership, license or other right) to grant access to, or a license or sublicense of, such Intellectual Property Rights without violating the terms of any agreement or other arrangement with any Third Party.

"Controlled Group" means Seller and any trade or business, whether or not incorporated, which is treated together with Seller as a single employer under Section 4001(b)(1) of ERISA or Sections 414(b), (c), (m), or (o) of the Code.

"Cut Back Shares" has the meaning set forth in Section 10.3.

"CVR" means a contingent value right issued under the CVR Agreement.

"CVR Agreement" means the Contingent Value Right Agreement in substantially the form attached hereto as Exhibit D.

"Deposit" means \$4,000,000.00 in immediately available funds, which amount was paid by Purchaser to Seller on December 23, 2023.

"Development" means all development activities for a pharmaceutical or biological product (whether alone or for use together, or in combination, with another active agent or ingredient as a combination product or combination therapy) that are directed to obtaining Regulatory Approval(s) of such product and lifecycle management of such product in any country in the world, including all non-clinical, preclinical, and clinical studies and trials of such product; toxicology, pharmacokinetic, and pharmacological studies; statistical analyses; assay development; protocol design and development; the preparation, filing, and prosecution of any Regulatory Approval Application for such product; development activities directed to label expansion and/or obtaining Regulatory Approval for one or more additional indications following initial Regulatory Approval; development activities conducted after receipt of Regulatory Approval; and all regulatory affairs related to any of the foregoing. "Develop", "Developing" and "Developed" and similar variations have correlative meanings.

#### "Development Milestone Event" has the meaning set forth in Section 3.3(a).

"Diligent Efforts" means, with respect to the efforts to be expended by Purchaser with respect to any task, objective, activity or decision to be undertaken with respect to the Development of a Product, the carrying out of such Development activities with a level of effort and resources consistent with those efforts that Purchaser and its controlled Affiliates devote to the Development of a pharmaceutical product owned by them or to which they have rights at a similar stage of development, and of similar scientific data and validation, market potential, profit potential and strategic value and based on conditions then prevailing. Such efforts may take into account, without limitation, scientific data (including safety and efficacy data), Regulatory Authority-approved labeling, product profile, the competitiveness of alternative products, pricing and reimbursement for such Product in a country relative to other markets, the likely timing of such Product's entry into the market, the proprietary position of such Product, the anticipated profitability of such Product and the other products in Purchaser's and its Affiliates' portfolio (including other products under development), the likelihood of Regulatory Approval and other relevant scientific, technical and commercial factors. Diligent Efforts shall be determined on a country by country basis and indication by indication basis and it is anticipated that the level of effort will change over time reflecting the change in conditions.

#### "Disclosure Schedules" has the meaning set forth in Section 5.1.

"Distributor" means any Third Party wholesaler or distributor engaged for the sale of Product provided that such wholesaler or distributor does not make any royalty, milestone, profit share or other payment to Purchaser and its Affiliates based on such wholesaler's or distributor's sale of Product.

#### "DOJ" means the Antitrust Division of the U.S. Department of Justice.

"Domain Names" means domain names, uniform resource locators, other names and locators associated with the Internet, and applications or registrations for the foregoing.

"Dropped Patents" means each of the following, whether or not pending, issued, expired, withdrawn, rejected, canceled, abandoned, or closed: (a) the Patents listed on Schedule C-1, and (b) any renewals, divisions, continuations (in whole or in part), or requests for continued examination of any of such patents, certificates of invention, and patent applications, any and all patents or certificates of invention issuing thereon, and any and all reissuances, reexaminations, extensions, divisions, renewals, substitutions, confirmations, registrations, revalidations, revisions, and additions of or to any of the Patents listed on Schedule C-1. "Effectiveness Deadline" has the meaning set forth in Section 10.2(b).

"Employee Benefit Plan" means (i) all "employee benefit plans" (as defined in Section 3(3) of ERISA), (ii) all other employee benefit plans, policies, agreements or arrangements, (iii) all employment, individual consulting, executive compensation, or other compensation agreements, or bonus or other incentive compensation, stock purchase, equity or equity-based compensation, deferred compensation, change in control, retention, severance, sick leave, vacation, recreation, retirement, pension, loans, salary continuation, health, medical, dental, vision, accident, disability, cafeteria, life insurance and educational assistance plans, policies, agreements or arrangements, and (iv) any collective bargaining agreement or union contract, in each case, whether written or unwritten and whether or not subject to ERISA, that are sponsored or maintained by Seller or any member of the Controlled Group for the benefit of current or former employees or current or former consultants, independent contractors or directors of Seller or any of its Subsidiaries has any Liability.

"Employee Liabilities" means any and all Liabilities, whenever or however arising, including all costs and expenses relating thereto arising under Contract, Law or Permit, Order or any award of any arbitrator of any kind relating to any Employee Benefit Plan, employment agreement, natural person consulting or independent contractor agreement, or otherwise relating to an employee or other natural person service provider and his or her potential or actual service or employment with Seller or any member of the Controlled Group.

"Enforceability Exception" has the meaning set forth in Section 5.1(b).

"ERISA" means the Employee Retirement Income Security Act of 1974.

"Evaluation Material" has the meaning set forth in Section 5.1(bb).

"Excess Liabilities" means (a) the excess (if any) of (i) the aggregate amount of Specified Liabilities over (ii) \$4,000,000, *plus* (b) the excess (if any) of (i) the sum of (A) the Qualified CMC Expenses reimbursed to Seller as of Closing and (B) any Qualified CMC Expenses assumed and paid by Purchaser after the Closing, over (ii) \$1,000,000.

"Exchange Act" means the Securities Exchange Act of 1934.

"Excluded Assets" has the meaning set forth in Section 2.1(b).

"Excluded Liabilities" means all Liabilities of Seller and its Affiliates not expressly included in the definition of Assumed Liabilities, including:

(a) except as expressly included as an Assumed Liability, all Liabilities arising from Seller's ownership or use of the Acquired Assets or the conduct of its business, including the Product Operations, as of or prior to Closing;

(b) all Liabilities relating to Excluded Assets;

(c) all Seller Tax Liabilities;

(d) all Liabilities of Seller or any of its Affiliates for product liability, infringement, or misappropriation arising from preclinical or clinical trials of a Product or Product Operations conducted by or for Seller on or prior to Closing;

(e) all Employee Liabilities;

(f) any Liabilities that arose or were incurred under any Acquired Business Contract on or prior to Closing or that relate to any failure to perform, improper performance, warranty or other breach, default, or violation by Seller on or prior to the Closing;

(g) except as expressly included as an Assumed Liability, all Liabilities for invoices, bills, accounts payable, or other trade payables due or owned by Seller or any of its Affiliates to any Third Party relating to or arising from the Development, manufacture (or having manufactured), packaging, importation, marketing, or distribution of a Compound or Product prior to the Closing;

(h) any Liability arising out of or resulting from non-compliance with any Law by Seller or its Affiliates with respect to the Product Operations or the Acquired Assets;

(i) any Liability of Seller or its Affiliates to pay any fees or commissions to any broker, finder, or agent with respect to this Agreement or any of the Transactions or the consummation thereof;

(j) any Liability arising out of or resulting from any liquidation, dissolution or winding up of Seller, including any Legal Proceeding, bankruptcy filing, assignment for the benefit of creditors or similar restructuring or reorganization involving Seller; and

(k) all Excess Liabilities.

"Exclusivity Agreement" means the Exclusivity Agreement, dated December 22, 2023, as amended, by and between Purchaser and Seller.

"FDA" means the U.S. Food and Drug Administration, or any successor agency thereto.

"FDC Act" means the U.S. Federal Food, Drug, and Cosmetic Act of 1938.

"Filed SEC Documents" has the meaning set forth in Section 5.1.

"Filing Date" has the meaning set forth in Section 10.2(a).

"Fraud" means, with respect to any Party hereto, the making of a representation and warranty set forth in <u>Article 5</u>, with actual knowledge that such representation and warranty was false when made, with the intent that the Purchaser (in the case of the Seller) or the Seller (in the case of the Purchaser) rely thereon to its detriment.

"FTC" means the U.S. Federal Trade Commission or any successor agency thereto.

"Fundamental Representations" means the representations and warranties contained in <u>Sections 5.1(a)</u> (*Organization and Existence*), <u>5.1(b)</u> (*Authority and Approval*), clause (i) of <u>5.1(c)</u> (*No Conflict*), <u>5.1(f)</u> (*Title*), <u>5.1(n)</u> (*No Brokers*) and <u>5.1(t)</u> (*Sufficiency of Acquired Assets*).

"GAAP" means U.S. generally accepted accounting principles, consistently applied.

"GCP" means good clinical practices, standards and procedures promulgated or endorsed by the FDA as set forth in the guidelines entitled "Guidance for Industry E6 Good Clinical Practice," including related regulatory requirements imposed by the FDA, in 21 C.F.R parts 11, 16, 50, 54, 56, 58, 312, 324, 320, 511, 514, 601, 812, and 814, and comparable regulatory standards, practices and procedures promulgated by any other Regulatory Authority, including applicable quality guidelines promulgated under the ICH, in each case, as amended from time to time and applicable to relevant activities hereunder.

"GLP" means good laboratory practices promulgated or endorsed by the FDA, as set forth in 21 C.F.R. Part 58, and comparable regulatory standards promulgated by any other Regulatory Authority, including applicable quality guidelines promulgated under the ICH, in each case, as amended from time to time and applicable to the relevant activities hereunder.

"GMP" means good manufacturing practices and standards for the production of drugs promulgated or endorsed by the FDA, as set forth in 21 C.F.R. Parts 210, 211, 600 through 680, 820, and 1271, as applicable, and comparable regulatory standards promulgated by any other Regulatory Authority, including applicable guidelines promulgated under the ICH, in each case, as amended from time to time and applicable to the relevant activities hereunder.

"Governmental Authority" means any federal, state, local, or any non-U.S. government, governmental, regulatory (including self-regulatory) or administrative authority, body, agency or commission or any court, tribunal, or judicial or arbitral body.

"HIPAA" means, collectively: (a) the Health Insurance Portability and Accountability Act of 1996; (b) the Health Information Technology for Economic and Clinical Health Act (Title XIII of the American Recovery and Reinvestment Act of 2009); and (c) the Omnibus Rule effective March 26, 2013 (78 Fed. Reg. 5566), and other implementing regulations at 45 C.F.R. Parts 160 and 164 and related binding guidance from the United States Department of Health and Human Services.

"HSR Act" means the Hart-Scott Rodino Antitrust Improvements Act of 1976, and rules and regulations promulgated thereunder.

"ICH" means the International Council for Harmonisation of Technical Requirements for Pharmaceuticals for Human Use.

"**IND**" means an investigational new drug application (including any amendment or supplement thereto) submitted to the FDA pursuant to U.S. 21 C.F.R. Part 312, or the comparable application submitted to the applicable Regulatory Authority, including any amendments thereto.

"Independent Expert" has the meaning set forth in Section 3.3(c)(iii).

"Information Statement" has the meaning set forth in Section 6.6(b).

"Insurance Policies" has the meaning set forth in Section 5.1(w).

"Intellectual Property Rights" means any and all of the following and any and all rights associated with the following, in any jurisdiction throughout the world, by whatever name or term known or designated, whether arising by operation of Law, Contract, or otherwise:

(a) Patents;

(b) Trademarks;

(c) Domain Names;

(d) all copyrights, mask work rights, moral rights, and common-law rights thereto, and all applications, registrations, and renewals in connection therewith throughout the world;

(e) all rights in databases and data collections;

(f) all Know-How;

world.

(g) all similar, corresponding, or equivalent rights to any of the foregoing anywhere in the world; and

(h) all rights to sue for past, present, or future infringement, violations, or misappropriation of any of the foregoing anywhere in the

"Inventor" means each of the named inventors of each of the Seller Patents, as well as any inventor who should be or should have been named on each of the Seller Patents.

"Inventor Assignment Agreement" means an agreement by Seller or its Affiliate with the respective Inventor assigning all right, title, and interest to the Seller Patents to Seller or its Affiliate. For clarity, any agreement with an employee or contractor of Seller or its Affiliate that provides for the general assignment of inventions made in the course of employment by or providing services to Seller or its Affiliate shall not be an "Inventor Assignment Agreement".

"Inventory" means all inventories of active pharmaceutical ingredient for the Compounds and Products, all drug product for the Compounds and Products, all raw and pack materials, work-in-process, finished goods, Products on stability, reference materials warehoused stock, supplies, consumables and packaging materials relating to the Product or manufacture thereof.

"IP and Patent Assignment Agreement" means the IP and Patent Assignment and Assumption Agreement in substantially the form attached hereto as Exhibit E.

"IQVIA Agreement" means the Master Services Agreement, dated July 21, 2020, by and between Old Ayala, Inc. (formerly Ayala Pharmaceuticals, Inc.), a Delaware corporation, and IQVIA Biotech LLC, as amended, together with that certain Amendment No. 2 to Initial Work Order Project Code: Al-DES-01/WZ95313, by and between Seller and IQVIA RDS Inc., to be executed after the date hereof.

"Key Personnel" has the meaning set forth in Section 6.12.

#### "Key Personnel Agreements" has the meaning set forth in Section 6.12.

"Know-How" means data, knowledge, trade secrets, know-how, discoveries, inventions and technical, business or other proprietary information, including materials, samples, manufacturing data, chemistry, manufacturing, and controls information, toxicological data, pharmacological data, data from preclinical, clinical or non-clinical testing or studies, assays, processes, protocols, procedures, methods, devices, designs, results, technology, platforms, formulations, regulatory filings, applications, approvals and associated correspondence, specifications, quality control testing data, customer lists, mailing lists and business plans. "Know-How" shall expressly exclude the subject matter of the Dropped Patents. Notwithstanding the foregoing, solely with respect to any representation by Seller pursuant to this Agreement (including but not limited to the representations set forth in <u>Section 5.1(m)</u>), "Know-How" shall expressly exclude subject matter that is solely claimed in the Dropped Patents. For clarity, subject matter that is claimed in both the Product Patents and the Dropped Patents shall be included within "Know-How."

"Knowledge". An individual shall be deemed to have "Knowledge" of a particular fact or other matter if such individual is actually aware of such fact or other matter, after reasonable investigation. Seller shall be deemed to have "Knowledge" of a particular fact or other matter if any individual set forth on Schedule D has Knowledge of such fact or other matter.

"Laws" means any United States federal, state and local, and any non-U.S., laws, statutes, regulations, rules, codes or ordinances enacted, adopted, issued or promulgated by any Governmental Authority (including those pertaining to electrical, building, zoning, environmental and occupational safety and health requirements) or common law, including GCP, GMP and GLP.

"Legal Proceeding" means any action, suit, charge, complaint, litigation, arbitration, proceeding (including any civil, criminal, administrative, investigative or appellate proceeding), hearing, inquiry, audit, examination or investigation commenced, brought, conducted or heard by or before, or otherwise involving, any court or other Governmental Authority or any arbitrator or arbitration panel.

"Liability" or "Liabilities" means, with respect to any Person, any debt, duty, liability or obligation of any kind (whether known or unknown, contingent, accrued, due or to become due, secured or unsecured, matured or otherwise), including accounts payable, royalties payable, and other reserves, accrued bonuses and commissions, accrued vacation and any other form of leave, termination payment obligations, employee expense obligations, Taxes (including penalties and interest), and all other liabilities and obligations of such Person or any of its Subsidiaries or Affiliates, regardless of whether such debts, duties, liabilities or obligations are required to be reflected on a balance sheet in accordance with GAAP.

"License Grants" means licenses, sublicenses, or other Contracts (whether royalty bearing or non-royalty bearing) under which rights in any Product IP have been granted to any Person by any of (a) Seller, (b) any Affiliate of Seller and (c) any other Person that, at the time of the execution of the license, sublicense, or other Contract, as applicable, is or was an Affiliate of Seller.

"Lien" means any lien, statutory lien, pledge, mortgage, security interest, claim, encumbrance, restriction on use or transfer, easement, right of way, option, conditional sale, or other title retention agreement of any kind or nature. In the case of an Intellectual Property Right that is subject to any inlicense or out-license, "Lien" includes the terms and conditions of such in-license or out-license, as applicable.

"Lock-Up Period" has the meaning set forth in Section 3.6.

"Lock-Up Shares" has the meaning set forth in Section 3.6.

"Look-Back Date" means the date that is three (3) years prior to the date of this Agreement.

"made available" means that Seller has provided to Purchaser that material in question, on before the second (2<sup>nd</sup>) Business Day prior to the date hereof, other than material that was delivered to Purchaser at the request of Purchaser made during such period.

#### "Mandatory Registration Statement" has the meaning set forth in Section 10.2(a).

"Material Adverse Effect" means any event, change, circumstance, occurrence, effect, result, or state of facts that, individually or in the aggregate, (a) is or would reasonably be expected to be materially adverse to the Acquired Assets, taken as a whole or materially increase the Assumed Liabilities or (b) materially impairs the ability of Seller to consummate, or prevents or materially delays, the Closing or would reasonably be expected to do so; *provided, however*, that any events, changes, circumstances, occurrences, effects, results, or states of facts resulting from the following items shall not be considered when determining whether a Material Adverse Effect has occurred: (i) changes in economic, political, regulatory, financial or capital market conditions generally or in the industries in which Seller operates; (ii) any acts of war, sabotage, terrorist activities or Laws or Orders imposed by a Governmental Authority associated with national security in response thereto; (iii) effects of weather or meteorological events, any epidemic, pandemic or disease outbreak, any public health emergency (as declared by the World Health Organization or the Health and Human Services Secretary of the United States) or other similar acts of God; (iv) any change of Law or accounting standards after the Agreement Date; (v) the announcement or pendency of this Agreement; and (vi) any failure by Seller to meet projections or forecasts or revenue or earnings predictions for any period (but, for the purposes of clarity, not the underlying cause of such failure); except in the case of each of clauses (i) through (iv), with respect to Seller, to the extent disproportionately affecting Seller relative to other similarly situated companies in the industries in which Seller operates.

"Milestone Payment" means any Development Milestone Payment or Net Sales Milestone Payment.

"Material Contracts" has the meaning set forth in Section 5.1(i)(i).

"Net Sales" means, with respect to any Product, the gross amounts invoiced by Purchaser and its Affiliates for sales of such Product to unaffiliated Third Parties, including to Distributors, less the following deductions provided to unaffiliated entities, each as actually allowed and taken and not reimbursed by any Third Party:

(a) trade, quantity and/or cash discounts, charge-back payments, allowances or rebates actually taken and allowed, including promotional or similar discounts or rebates and discounts or rebates to wholesalers or other distributors, buying groups, healthcare insurance carriers or other institutions, managed health care organizations, pharmacy benefit managers (or equivalents thereof), national, state/provincial, local, and other governments, their agencies and purchasers and reimbursors, or to trade customers (managed care and similar types of rebates and chargebacks);

(b) any payment actually made in respect to sales to any governmental authority in respect of any government subsidized program, including Medicare and Medicaid rebates;

(c) discounts provided in connection with coupon, voucher or similar patient programs;

(d) credits or allowances actually given or made with respect to Products by reason of rejection, defects, recalls, returns, rebates and retroactive price reductions;

(e) a reasonable allowance for bad debt (not to exceed 2% of Net Sales in the aggregate);

(f) any tax, tariff, duty or government charge (including any sales, value added, excise or similar tax or government charge, but excluding any income tax) levied on the sale, transportation or delivery of a Product and borne by the seller thereof, as adjusted for rebates and refunds, without reimbursement from any Third Party, including that portion of the annual fee on prescription drug manufacturers imposed by the Patient Protection and Affordable Care Act, Pub. L. No. 111-148 (as amended), that Purchaser or its Affiliates, as applicable, allocate to sales of such Product in accordance with Purchaser's or its Affiliates' standard policies and procedures consistently applied across its products, as applicable, in each case, to the extent non-creditable or refundable;

(g) any charges for freight, postage, shipping or transportation, or for insurance, in each case to the extent borne by Purchaser and/or its Affiliates; and

(h) any administrative fees paid to group purchasing organizations or managed care entities for the sale of Products.

Notwithstanding the foregoing, amounts received or invoiced by Purchaser or its Affiliates for the sale of Products among Purchaser and its Affiliates shall not be included in the computation of Net Sales hereunder. Net Sales shall be accounted for in accordance the selling party's standard practices in the relevant country of the sale and in accordance with GAAP.

Notwithstanding the foregoing, "Net Sales" shall not include any amounts invoiced for sales of Products supplied for use in clinical trials of Products, or under early access, compassionate use, named patient, indigent access, patient assistance or other reduced pricing programs.

Net Sales for a Combination Product shall be calculated as follows:

(i) If the Product and the Other Product(s) in such Combination Product each are sold separately in the same country during the same calendar year, Net Sales will be calculated by multiplying the total Net Sales (as described above) of the Combination Product by the fraction A/(A+B), where A is the public or list price in such country and calendar year of the Product sold separately in the same formulation and dosage, and B is the (sum of the) public or list price(s) in such country and calendar year of the Other Product(s) sold separately in the same formulation and dosage.

(ii) If the Product is sold independently of the Other Product(s) in such Combination Product in such country and calendar year, but the public or list price of the Other Product(s) cannot be determined, Net Sales will be calculated by multiplying the total Net Sales (as described above) of such Combination Product by the fraction A/C, where A is the public or list price in such country and calendar year of the Product sold independently and C is the public or list price of the Combination Product in such country.

(iii) If the Other Product(s) in such Combination Product are sold independently of the Product in such country and calendar year, but the public or list price of the Product cannot be determined, Net Sales will be calculated by multiplying the total Net Sales (as described above) of such Combination Product by the fraction [1-B/C], where B is the (sum of the) public or list price(s) in such country and calendar year of the Other Product(s) and C is the public or list price in such country and calendar year of the Combination Product.

(iv) If neither the Product nor the Other Product(s) in such Combination Product are sold separately in such country and calendar year, then Purchaser will determine a calculation of Net Sales for the Combination Product in good faith based on the relative values of the Product and the Other Product(s).

"NDA" means (a) a New Drug Application or Biologics License Application, as applicable, for any product requesting permission to market a drug or biologic, and all supplements or amendments thereto, filed pursuant to the requirements of the FDA, including all documents, data and other information concerning a product which are reasonably necessary for FDA approval to market a product in the United States and (b) all equivalents of the foregoing in any jurisdiction outside of the United States.

"Net Sales Information" has the meaning set forth in Section 3.3(c)(ii).

"Net Sales Report" has the meaning set forth in Section 3.3(c)(i).

"Nondisclosure Agreements" means any nondisclosure, confidentiality, or similar agreements in effect as of the date of this Agreement to which Seller is a party that primarily relate to the Acquired Assets or any Product IP, or any other Compound or Product.

"Non-Paying Party" has the meaning set forth in Section 6.5(b).

"Non-Scheduled License Grants" has the meaning set forth in Section 5.1(m)(ii).

"Non-Transferable Assets" has the meaning set forth in Section 2.4.

"Objection Notice" has the meaning set forth in Section 3.3(c)(iii).

"Opinion" has the meaning set forth in <u>Section 5.1(y)</u>.

"Order" means and includes any writ, law, rule, regulation, executive order or decree, judgment, injunction, ruling, or other order, whether temporary, preliminary, or permanent enacted, issued, promulgated, enforced, or entered into by any Governmental Authority.

"Orderly Trading Period" has the meaning set forth in Section 3.7.

"Ordinary Course of Business" means the ordinary course of business of Seller consistent with Seller's past custom and practice.

"Organizational Document" means (a) the articles or certificate of incorporation, association, or formation and the bylaws of a corporation; (b) the operating agreement, limited liability company agreement, or similar document governing a limited liability company; (c) any charter or similar document adopted or filed in connection with the creation, formation, or organization of a Person; and (d) any amendment to any of the foregoing.

"Outside Date" has the meaning set forth in Section 11.1(e).

"Party" or "Parties" has the meaning set forth in the preamble.

"Patent Documents" means all (a) prosecution files for each of the Seller Patents and any other Product Patents for which Seller or its Affiliate is responsible for filing or prosecution; (b) Assignment Agreements and all Inventor Assignment Agreements; and (c) documents, records, and files in the possession and Control of Seller, its counsel, or its agents with respect to (i) the conception and reduction to practice (and diligence in reduction to practice) of the inventions of any of the Seller Patents, or (ii) the filing, prosecution, registration, continuation, continuation-in-part, reissuance, correction, enforcement, defense, and maintenance of the Seller Patents and any other Product Patents for which Seller or its Affiliate is responsible for filing or prosecution.

"**Patents**" means (a) all patents, certificates of invention, applications for certificates of invention, priority patent applications, and patent applications, and (b) any renewals, divisions, continuations (in whole or in part), or requests for continued examination of any of such patents, certificates of invention, and patent applications, any and all patents or certificates of invention issuing thereon, and any and all reissuances, reexaminations, extensions, divisions, renewals, substitutions, confirmations, registrations, revalidations, revisions, and additions of or to any of the foregoing.

"Paying Party" has the meaning set forth in Section 6.5(b).

"**Permits**" means any license, permit, registration, listing, approval, qualification, letter, authorization, certificate of authority, qualification, NDA, IND, or similar document or authority issued or granted by any Regulatory Authority or pursuant to any Law.

"**Permitted Liens**" means (a) Liens for Taxes or similar governmental assessments and charges (i) not yet due and payable, or (ii) that are being contested in good faith through appropriate proceedings and for which adequate reserves have been established, (b) Liens referred to in the Acquired Business Contracts, (c) in the case of an Intellectual Property Right that is subject to any non-exclusive in-license or out-license, the terms and conditions of such in-license or out-license as apparent from the face of such license and the license is included in any Material Contract set forth on Section 5.1(i)(i) of the Disclosure Schedules, in any Contract set forth on Section 5.1(m)(i) or 5.1(m)(i) of the property which they encumber.

"Person" means any individual, corporation, partnership, joint venture, limited liability company, association, joint-stock company, trust, unincorporated organization or Governmental Authority.

"**Positive Topline Data**" means data showing that the RINGSIDE Part B Study met its primary endpoint of improving progression-free survival of patients with progressive desmoid tumors, as assessed by blinded independent central review, demonstrating a statistically significant improvement for AL102 over placebo as defined by the clinical protocol, specifically a hazard ratio of 0.35 or better.

"**Pre-Closing Tax Period**" means any taxable period ending on or before the Closing Date and, in the case of any Straddle Period, the portion of such period beginning before and ending on the Closing Date as determined by <u>Section 6.5(b)</u>.

"Product" means any product comprised of or containing either of the Compounds, in any form or formulation.

"Product Information" has the meaning set forth in Section 6.12.

"Product IP" means any and all (a) Product Patents, (b) Product Know-How and (c) other Intellectual Property Rights Controlled by Seller or any of its Affiliates that is related to any Compounds or Products but excluding (i) Domain Names and (ii) Trademarks.

"**Product Know-How**" means all Know-How Controlled by Seller or any of its Affiliates as of the Closing Date that (a) is used or held for use in or is necessary or useful for the Product Operations or (b) is necessary or useful for the research, Development, manufacture, or Commercialization of any Compound or Product, including, for the avoidance of doubt, all of Seller's and its Affiliates' Know-How made before the Closing Date that (i) constitutes the composition of matter of any Compound or Product, or a method of manufacturing or using any Compound or Product, or (ii) relates to a companion diagnostic for any Compound or Product.

"Product Operations" means the research, Development, manufacture, formulation, testing, use, distribution, marketing, sale, promotion, and other exploitation of any Compound or Product as conducted by Seller and its Affiliates, and any Affiliates of Seller prior to the Agreement Date, in each case at any time prior to the Closing Date (except as expressly provided elsewhere in this Agreement). "Product Operations" shall expressly exclude the subject matter of the Dropped Patents. Notwithstanding the foregoing, solely with respect to any representation by Seller pursuant to this Agreement (including but not limited to the representations set forth in Section 5.1(m)), "Product Operations" shall expressly exclude subject matter that is claimed in both the Product Patents and the Dropped Patents shall be included within "Product Operations."

"**Product Patents**" means each of the following, whether or not pending, issued, expired, withdrawn, rejected, canceled, abandoned, or closed: (a) the Patents listed on **Schedule C**, and (b) any renewals, divisions, continuations (in whole or in part), or requests for continued examination of any of such patents, certificates of invention, and patent applications, any and all patents or certificates of invention issuing thereon, and any and all reissuances, reexaminations, extensions, divisions, renewals, substitutions, confirmations, registrations, revisions, and additions of or to any of the Patents listed on **Schedule C**. The Product Patents shall expressly exclude the Dropped Patents.

"Purchaser Common Stock" means shares of Purchaser's common stock, \$0.0001 par value per share.

"Purchaser Parties" has the meaning set forth in Section 10.4(b).

"Purchaser Milestone Representative" has the meaning set forth in Section 3.3(c)(iv).

"Purchaser SEC Reports" has the meaning set forth in Section 5.2(h)(i).

"Registrable Shares" has the meaning set forth in Section 10.1(b).

"Registration Liability Cap" has the meaning set forth in Section 10.4(b).

"Qualified CMC Expenses" shall mean all costs and expenses payable to the CMOs set forth on Schedule E for CMC related work for the Products arising from purchase orders or Contracts set forth on Schedule E that are approved by Purchaser (with such approval not to be unreasonably withheld, delayed or conditioned) in an aggregate amount not to exceed \$1,000,000 (unless otherwise approved in writing by Purchaser), which purchase orders or Contracts shall be included in the Acquired Business Contracts.

"**Regulatory Approval**" means, with respect to any Product in any regulatory jurisdiction, approval from the applicable Regulatory Authority sufficient for the manufacture, distribution, use, marketing, and sale of the Product in such jurisdiction in accordance with applicable Laws.

"Regulatory Approval Application" means an application submitted to the applicable Regulatory Authority in pursuit of receiving Regulatory Approval of a Product in a country, regulatory jurisdiction or region, and all amendments and supplements thereto.

"Regulatory Authority" means any applicable Governmental Authority involved in the granting and maintenance of Regulatory Approvals in any country worldwide, or the conduct of clinical investigations, including the FDA in the United States.

"Regulatory Documentation" means all books, records, files, documents, information and correspondence of Seller or its Affiliates to the extent (a) relating to the Compounds, Products, Acquired Assets, or Assumed Liabilities, (b) used or held for use in the Product Operations, or (c) generated in the conduct of the Product Operations, including (i) all records with respect to supply sources; (ii) all market research data, market intelligence reports, statistical programs (if any) used for marketing and sales research with respect to the Products; (iii) promotional, advertising and marketing materials, sales forecasting models, medical education materials, sales training materials, web site content, and advertising and display materials relating to the Products; (iv) all records, including vendor and supplier lists, research, development and manufacturing records, sampling records, standard operating procedures and batch records, related to the manufacturing process for any Compounds or Products; (v) all data contained in laboratory notebooks to the extent relating to any Compounds or Products or relating to the biological, physiological, mechanical or formula properties of any of the foregoing, including data related to preclinical and clinical trials and investigator brochures, and all statistical programs developed (or modified in a manner to the use or function thereof) to analyze such data; (vi) all adverse experience reports and files related thereto (including source documentation) and all periodic adverse experience reports and all data contained in electronic data bases relating to periodic adverse experience reports with respect to any Compounds or Products; (vii) all analytical and quality control data relating to any Compounds or Products; (viii) all correspondence, minutes, and other communications with the FDA, any other Regulatory Authority and institutional review board or research ethics committee relating to any Compounds or product, and (ix) HIPAA authorizations collected in connection with cl

"Regulatory Filings" means any and all regulatory applications, filings, approvals and associated correspondence required or related to the Development, manufacture or Commercialization of any Compound or Product in any country or jurisdiction, including all INDs relating to the Compounds and the Products.

"Related Agreements" means the Bill of Sale, Assignment and Assumption Agreement, the Support Agreements, the CVR Agreement and any agreement, document, or instrument entered into or delivered in connection with this Agreement and the Transactions.

"Representatives" means officers, directors, employees, stockholders, partners, members, agents, attorneys, accountants, bankers, advisors and representatives.

"Requisite Holders" has the meaning set forth in the Recitals.

"Resale Registration Statement" has the meaning set forth in Section 10.1(a).

"Restriction Termination Date" has the meaning set forth in Section 10.3.

"**RINGSIDE Part B Study**" means the Phase 3/Part B double-blind, placebo-controlled part of the Study of AL102 in Patients with Progressing Desmoid Tumors (RINGSIDE Part B Study)(NCT04871282) being conducted by or on behalf of Seller.

"Rule 144" means Rule 144 as promulgated by the SEC under the Securities Act, or any similar successor rule that may be promulgated by the SEC.

"Rule 424" means Rule 424 as promulgated by the SEC under the Securities Act, or any similar successor rule that may be promulgated by the SEC.

"SEC" means the United States Securities and Exchange Commission.

"SEC Restrictions" has the meaning set forth in Section 10.3.

"Securities Act" means the Securities Act of 1933.

"Seller" has the meaning set forth in the preamble.

"Seller Common Stock" means shares of Seller's common stock, par value \$0.001 per share.

"Seller Financial Statements" has the meaning set forth in Section 5.1(x)(ii).

"Seller IP" means any and all (a) Product Patents owned by Seller or its Affiliate, (b) Product Know-How owned by Seller or its Affiliate and (c) other Intellectual Property Rights (including any Domain Names or Trademarks) owned by Seller or any of its Affiliates that is related to any Compounds or Products.

"Seller Parties" has the meaning set forth in Section 10.4(a).

"Seller Patents" means any and all Product Patents owned by Seller or its Affiliate.

"Seller SEC Reports" means each form, report, schedule, registration statement, definitive proxy statement and other document (together with all amendments thereof and supplements thereto) filed or furnished by Seller pursuant to the Securities Act or the Exchange Act with the SEC since January 1, 2023, in each case, as such documents have since the time of their filing been amended or supplemented as of the date that is two (2) Business Days prior to the Agreement Date.

"Seller Tax Liabilities" means (i) all Taxes of Seller or its Affiliates, or for which Seller or any of its Affiliates is or are liable (including as a transferee or successor, or by contract or otherwise by operation of Law), for any taxable period (to the extent not relating to the Acquired Assets, the Product Operations or the Assumed Liabilities) and for any Pre-Closing Tax Period (to the extent relating to the Acquired Assets, the Product Operations or the Assumed Liabilities) (including any such Tax of Seller or any of its Affiliates that becomes a Liability of Purchaser under any common law doctrine of de facto merger or transferee or successor liability or otherwise by operation of contract or Law), (ii) any Taxes of another Person for which Seller is liable (including under Treasury Regulation Section 1.1502-6 or any similar provision of state, local or non-U.S. applicable Law) as a result of being a member of an affiliated, consolidated, combined or unitary group for Tax purposes on or before the Closing Date or any similar provision of state, local, or foreign applicable Law, (iii) any Taxes that arise out of the transactions contemplated by this Agreement (including withholding taxes imposed on payments under this Agreement and fifty-percent of any Transfer Taxes pursuant to Section 6.5(c)), (iv) any Taxes relating to the Excluded Assets or Excluded Liabilities for any Pre-Closing Tax Period.

"SOX" has the meaning set forth in Section 5.1(x)(i).

"Specified Liabilities" has the meaning set forth in Section 2.2(b).

"Stockholder Written Consent" has the meaning set forth in Section 6.6(a).

"Straddle Period" has the meaning set forth in Section 6.5(b).

"Straddle Period Tax" has the meaning set forth in Section 6.5(b).

"Strategic Transaction" shall mean any transaction involving: (a) the sale, license, disposition or acquisition of the Acquired Assets; (b) the issuance, grant, disposition or acquisition of (i) any capital stock or other equity security of Seller or any direct or indirect subsidiary of Seller, (ii) any option, call, warrant or right (whether or not immediately exercisable) to acquire any capital stock or other equity security of Seller or any direct or indirect subsidiary of Seller, or (iii) any security, instrument or obligation that is or may become convertible into or exchangeable for any capital stock or other equity security of Seller or any direct or indirect Subsidiary of Seller; or (c) a merger, amalgamation, consolidation, share exchange, business combination, sale of substantially all the assets, reorganization, recapitalization, liquidation, dissolution or other similar transaction involving Seller or any of its direct and indirect subsidiaries; *provided, however*, that none of the following shall be deemed to constitute a "Strategic Transaction": (A) the grant of stock, stock options or other equity or equity-based compensation, or the granting or payment of cash bonuses, right to proceeds or other payments, by Seller to its employees and contractors if such grant and/or payment is made in the Ordinary Course of Business or in connection with the Transactions, (B) the issuance of stock by Seller to its employees and contractors upon the exercise of outstanding stock options, (C) the issuance of stock by Seller in connection with the exercise by the holders thereof of any of Seller's currently outstanding convertible debt and equity securities, and (D) the issuance of securities as contemplated by the right of certain existing investors of Seller to (1) lend an additional \$4.0 million to Seller on the same terms as the loan made to Seller on November 17, 2023, including warrant coverage, and (2) purchase senior convertible promissory notes of Seller up to an aggregate amount of \$1.458 million, all

"**Subsidiary**" means with respect to any Person, any other Person (a) of which the initial Person directly or indirectly owns or controls more than 50% of the voting equity interests or has the power to elect or direct the election of a majority of the members of the governing body of such Person or (b) which is required to be consolidated with such Person under GAAP.

"Tail Policy" has the meaning set forth in Section 6.10.

"Tax" or "Taxes" means any and all federal, provincial, territorial, state, municipal, local, foreign or other taxes (including imposts, rates, levies, assessments, and other charges, in each case in the nature of a tax), including all income, excise, franchise, gains, capital, real property, goods and services, transfer, value added, gross receipts, windfall profits, severance, ad valorem, personal property, production, sales, use, license, stamp, documentary stamp, mortgage recording, employment, payroll, social security, unemployment, disability, estimated or withholding taxes, and all customs and import duties, together with all interest, penalties and additions to tax (and additional amounts imposed with respect to such amounts).

#### "Tax Contest" has the meaning set forth in Section 6.5(d).

"**Tax Return**" means all U.S. federal, state, local, provincial and non-U.S. returns, declarations, claims for refunds, forms, statements, reports, schedules, information returns or similar statements or documents and any amendments thereof (including any related or supporting information or schedule attached thereto) filed or required to be filed with any applicable Governmental Authority in connection with the determination, assessment or collection of any Tax.

"Third Party" means any Person other than a Party hereto or an Affiliate of a Party hereto.

"**Trademarks**" means trademarks, trade names, corporate names, service marks, brand names, logos, trade dress, slogans, and other indicia of source or origin together with all translations, adaptations, and combinations thereof and including all goodwill associated with the foregoing and all common-law rights thereto, as well as all applications, registrations and renewals in connection therewith.

"Trading Day" means any day on which the primary market on which shares of Purchaser Common Stock are listed as open for trading.

"Transactions" means the transactions contemplated pursuant to this Agreement and the other Related Agreements.

"Transfer Agent" has the meaning set forth in Section 3.5(c).

"Transfer Taxes" has the meaning set forth in Section 6.5(c).

"Transferred Permits" has the meaning set forth in Section 5.1(h).

"U.S." means the United States of America and its territories and possessions.

"Written Consent Delivery Time" has the meaning set forth in Section 6.6(a).

#### 2.1 Acquired Assets.

(a) Purchase and Sale. Upon the terms and subject to the conditions set forth in this Agreement, Seller shall cause to be sold, transferred and conveyed to Purchaser, at the Closing, all of Seller's and its Subsidiaries' right, title and interest at the Closing in the (a) Acquired Assets, free and clear of all Liens other than Permitted Liens, and (b) the Dropped Patents. At the Closing, the sale, transfer, conveyance, assignment, and delivery of the Acquired Assets and the Dropped Patents will be effected pursuant to the Assignment and Assumption Agreement and the Bill of Sale. Notwithstanding anything to the contrary contained in this Agreement, the transfer of the Acquired Assets will not include the assumption by Purchaser of any Liability of Seller related to the Acquired Assets, unless Purchaser expressly assumes that Liability as an Assumed Liability pursuant to <u>Section 2.2</u>. "Acquired Assets" means the following properties, assets, and rights of Seller:

(i) all Seller IP together with (A) any and all goodwill symbolized thereby and associated therewith, (B) any and all rights to royalties, profits, compensation, license payments, and other payments or remuneration of any kind relating to the Acquired Assets, and (C) any and all rights to obtain renewals, reissues, reexaminations, supplemental examinations and certificates and extensions of registrations, exclusivities, or other legal protections pertaining to the Seller IP and all Patent Documents;

(ii) all rights in, to, and under the Contracts listed on Schedule 2.1(a)(ii) (including the BMS License and any additional Contracts listed on Schedule 2.1(a)(ii) as updated by Seller and delivered to Purchaser at least five Business Days prior to the anticipated Closing Date; *provided* that Purchaser shall have the right to reject inclusion of (i) any Contracts proposed to be listed on the updated Schedule 2.1(a)(ii) or (ii) any Contracts listed on Schedule 2.1(a)(ii) that were materially amended or under which material rights were waived after the Agreement Date) (collectively, the "Acquired Business Contracts");

(iii) all Inventory;

(iv) all Regulatory Filings and other Regulatory Documentation, including all copies thereof;

(v) the Transferred Permits;

(vi) all chemical or biological materials relating to the Compounds or Products, including all patient samples from clinical trials;

(vii) all rights, claims, credits (including the \$2,480,000.00 credit under the IQVIA Agreement), guaranties, warranties, indemnities, causes of action or rights of set-off, and other similar rights against Third Parties to the extent relating to or arising from the Compounds, the Products, the Acquired Assets or the Assumed Liabilities; and

(viii) Seller's rights in or to any equipment used in the development or manufacturing the Compounds or Products.

(b) Seller and Purchaser expressly agree and acknowledge that Purchaser is not acquiring any right, title, or interest in any assets that are not Acquired Assets (collectively, the "Excluded Assets").

**2.2 Assumed Liabilities.** Upon the terms and subject to the conditions of this Agreement, at the Closing, Purchaser will assume and agree to pay, perform, and discharge only those Liabilities that are Assumed Liabilities. The assumption of the Assumed Liabilities by Purchaser will be effected pursuant to the Assignment and Assumption Agreement. For purposes of this Agreement, "Assumed Liabilities" means only the following liabilities of Seller:

(a) all Liabilities of Seller and its Affiliates under the Acquired Business Contracts arising on or after the Closing Date, including Liabilities under the BMS License (but excluding (i) Liabilities resulting from any breach or non-compliance with of any such Acquired Business Contract by Seller or any of its Affiliates or (ii) Seller Tax Liabilities); and

(b) any Liability or other obligation expressly set forth on Schedule 2.2(b) in an aggregate amount not to exceed \$4,000,000 (the Liabilities identified in this clause (b), the "Specified Liabilities").

2.3 Excluded Liabilities. The Parties acknowledge and agree that Purchaser will not, and in no event will Purchaser assume or be required to pay, perform, or discharge any Liabilities other than the Assumed Liabilities, and that, as between the Parties, Seller shall remain responsible for all Excluded Liabilities.

**2.4 Assets Incapable of Transfer.** Notwithstanding anything herein to the contrary, this Agreement will not constitute an assignment or transfer of, an attempted assignment or transfer of, or an agreement to effect an assignment or transfer of, Acquired Assets that are not assignable or transferable without the consent of another Person (that has not been obtained at or prior to Closing) (the "Non-Transferable Assets"), if such assignment or transfer, attempted assignment or transfer, or agreement would constitute a breach of any Acquired Business Contract in the absence of such consent; *provided*, *however*, that, except as provided in <u>Section 4.2(a)(v)</u>, Closing shall occur notwithstanding the foregoing without any adjustment to the Consideration on account thereof. For a period of 12 months following the Closing Date, Seller will, at no cost to Purchaser, use commercially reasonable efforts to obtain the consent of such other Person to the assignment or transfer of any such Non- Transferable Asset to Purchaser in all cases in which such consent is required for such assignment or transfer. Purchaser will reasonably cooperate with Seller in its efforts to obtain such consents. To the extent any such consent cannot be obtained, Seller will, at no cost to Purchaser, use commercially reasonable efforts to Purchaser under the relevant Non- Transferable Asset; *provided*, *however*, that Seller shall not be required to undertake any work or take any action that would constitute a breach of any such Contract included in a Non-Transferable Asset. Without limiting the generality of the foregoing, the beneficial interest in and to the Acquired Assets, to the fullest extent permitted by the relevant Contract or Permit and applicable Law, will pass to Purchaser.

#### 3. CONSIDERATION

**3.1 Consideration.** In consideration for the purchase and sale of the Acquired Assets, at the Closing, upon the terms and subject to conditions set forth herein, Purchaser shall pay the Consideration. The "**Consideration**" means the sum of (a) the Closing Consideration, (b) the Milestone Payments and (c) the Assumed Liabilities; *provided*, in the case of clause (b), such payments shall be made only when and to the extent they are payable pursuant to the terms of this Agreement.

#### 3.2 Closing Consideration.

(a) At the Closing, Purchaser shall pay (or cause to be paid) to Seller an amount of immediately available funds equal to (i) \$20,000,000, *minus* (ii) the Deposit (provided the Deposit has not already been refunded to Purchaser in accordance with <u>Section 6.8</u>), *plus* (iii) the aggregate amount of documented Qualified CMC Expenses incurred and paid by Seller prior to Closing as documented to Purchaser at least two (2) Business Days prior to Closing, *minus* (iv) the aggregate amount of Excess Liabilities (collectively, the "**Closing Cash Payment**"), by wire transfer to one or more accounts provided to Purchaser by Seller no later than two (2) Business Days prior to the Closing.

(b) At the Closing, Purchaser shall issue to Seller 2,175,489 unregistered shares of Purchaser Common Stock (the "Shares", and together with the Closing Cash Payment, the "Closing Consideration").

(c) At the Closing, Purchaser shall assume and agree, on and after the Closing Date, to pay, perform and discharge promptly and fully when due the Assumed Liabilities.

#### 3.3 Milestone Consideration.

(a) Development Milestone Payments. Subject to Closing and the other terms and conditions of this Agreement, from and after the Closing, when an event set forth in the table below is achieved (each such event, a "Development Milestone Event"), Purchaser shall pay or issue (or cause to be paid or issued), as applicable, to Seller, in accordance with and subject to the terms of this Agreement, the one-time, non-refundable, non-creditable payment equal to the corresponding amount set forth in the table below (each such payment, a "Development Milestone Payment"):

Development Milestone Event	Development Milestone Payment
Public announcement of Positive Topline Data	\$10,000,000 in cash
Grant of NDA Regulatory Approval of a Product by the FDA in a	
first indication	\$17,500,000 in cash

Each of the Development Milestone Payments shall be payable or issuable only one time, for the first achievement of the corresponding Development Milestone Event, and no Development Milestone Payment would be due for subsequent or repeated achievements of the same Development Milestone Event, whether by the same Product or a different Product. Each Development Milestone Payment shall be paid or issued within 45 days after the achievement of the corresponding Development Milestone Event. In no event shall the total Development Milestone Payments exceed \$27,500,000.

(b) Net Sales Milestone Payment. Subject to Closing and the other terms and conditions of this Agreement, from and after the Closing, when an event set forth in the table below is achieved (each such event, a "Net Sales Milestone Event"), Purchaser shall pay (or cause to be paid) to Seller, in accordance with and subject to the terms of this Agreement, the one-time, non-refundable, non-creditable payment equal to the corresponding amount set forth in the table below (each such payment, a "Net Sales Milestone Payment"):

Net Sales Milestone Event	Net Sales Milestone Payment	
First calendar year in which Annual Net Sales of AL102 exceed		
\$100,000,000	\$10,000,000 in cash	

The Net Sales Milestone Payment set forth above shall be payable only one time, for the first achievement of such Net Sales Milestone Event. In no event shall the total Net Sales Milestone Payment exceed \$10,000,000. The Net Sales Milestone Payment shall be paid within 90 days after the end of the calendar year in which such Net Sales Milestone Event is achieved.

#### (c) Net Sales Reporting.

(i) Purchaser shall, within 90 days following the date of the first calendar year following the date hereof, and within 90 days following each calendar year thereafter, provide to Seller a report (each, a "Net Sales Report") setting forth, in reasonable detail, the Annual Net Sales of AL102 for each such calendar year, calculated in accordance with the definition of "Net Sales" as set forth herein; provided that Purchaser's obligation to prepare a Net Sales Report shall terminate automatically upon payment of the Net Sales Milestone Payment under this Agreement or the CVR Agreement, as applicable.

(ii) Purchaser shall keep complete and accurate, in all material respects, books and records of Net Sales of AL102 to the extent required to calculate the Net Sales Milestone Payment payable hereunder (the "Net Sales Information") and shall maintain the Net Sales Information until one year after the last day of the calendar year to which such Net Sales Information relates.

(iii) Notwithstanding anything to the contrary in this Agreement, Seller shall have 30 days after the receipt of a Net Sales Report pursuant to <u>Section 3.3(c)(i)</u> to dispute any of the calculations therein by providing written notice thereof (including a reasonably detailed basis thereof) to Purchaser (such notice, an "**Objection Notice**"). If the Seller does not deliver an Objection Notice within such 30-day period, it shall be deemed to have irrevocably consented to the applicable Net Sales Report and the calculations therein. If the Seller delivers an Objection Notice the Parties shall work in good faith to resolve such dispute. If the Parties are unable to resolve any such dispute within 30 days after Purchaser's receipt of an Objection Notice, the matters remaining in dispute shall be submitted for resolution to a nationally recognized independent accounting firm to be mutually agreed upon by Seller and Purchaser (such agreed firm, the "**Independent Expert**"). The Independent Expert's determination of Net Sales shall be within the range of values proposed by Purchaser and Seller and the Independent Expert shall make such determination in accordance with the definition of "Net Sales" and the other applicable terms of this Agreement. The Independent Expert's decision shall be final, absent manifest error, and the costs of such Independent Expert shall be borne by the Seller if the Annual Net Sales are determined by the Independent Expert to be \$100,000,000 or less and by Purchaser if the Annual Net Sales are determined by the Independent Expert's final determination. No later than 30 days after such decision and in accordance with such decision, Purchaser shall pay any amount of any Milestone Payment owing to Seller in accordance with such decision.

(iv) In the event that, following the Closing Date, Seller is wound up, liquidated or dissolved, Seller shall be entitled to designate, by written notice to Purchaser, another Person who shall be reasonably acceptable to Purchaser to enforce its rights and assume its obligations under this <u>Section 3.3</u> (the "**Purchaser Milestone Representative**"), and such Purchaser Milestone Representative shall be a third-party beneficiary of this Agreement for purposes of this <u>Section 3.3</u>. The Purchaser Milestone Representative shall agree to be bound to a customary confidentiality agreement in form and substance reasonably satisfactory to Purchaser as a condition to serving in such capacity.

(d) Purchaser Diligence. Commencing upon the Closing, Purchaser shall use Diligent Efforts to achieve the Development Milestone Events set forth in Section 3.3(a) for one Product in one indication in the U.S. Neither Purchaser nor any of its Affiliates shall take any action the primary purpose of which is to avoid the payment of any Milestone Payment. The Parties intend the express provisions of this Section 3.3(d) to govern their contractual obligations with respect to the rights and obligations under this Section 3.3(d) and to supersede any standard of efforts or implied covenant of good faith and fair dealing that might otherwise be imposed by applicable Law with respect thereto.

(e) Non-Transferable Right. The right of Seller to receive any amounts with respect to Milestone Payments (i) shall not be evidenced by a certificate or other instrument, (ii) shall not be assignable or otherwise transferable by Seller other than pursuant to a court Order, by operation of Law (including a consolidation or merger) or without consideration in connection with the dissolution, liquidation or termination of any corporation, limited liability company, partnership or other entity and (iii) does not represent any right other than the right to receive the Milestone Payments pursuant to this Agreement. Any attempted transfer of the right to any amounts with respect to any such payment by any holder thereof (other than as specifically permitted by the immediately preceding sentence) shall be null and void. Notwithstanding the foregoing, if Seller's stockholders approve a plan of dissolution in compliance with Section 6.9 and after the date that is six (6) months following the Closing Date Seller decides to make a pro rata distribution of its rights to receive all of the remaining Milestone Payments under this Section 3.3 to its stockholders, Seller shall notify Purchaser and the applicable Parties shall enter into the CVR Agreement. Seller shall distribute one CVR for each share of common stock of Seller that is outstanding as of the record date for such distribution, which will represent an aggregate right to receive, upon achievement of the applicable Development Milestone Event or Net Sales Milestone Event, the applicable Milestone Payments under this Agreement to the extent not paid to Seller prior to the date of distribution of the Parties and the right of Seller to receive any Milestone Payments under this Agreement and under the CVR Agreement (and any CVRs that are issued pursuant thereto) exceed \$37,500,000 in the aggregate.

(f) Right of Set-Off. Purchaser shall have the right, but not the obligation, to withhold from any unpaid future Milestone Payments due and payable to Seller the amount of any Excess Liabilities to the extent not already taken into account in the calculation of Closing Cash Payment pursuant to Section 3.2(a) (*i.e.*, Purchaser may withhold an aggregate amount from any Milestone Payments up to 100% of the amount of its Excess Liabilities). In order to exercise its right of Set-Off, Purchaser must deliver a notice to Seller which contains (a) a description and calculation of the amount of Excess Liabilities to be set off against the Milestone Payments, (b) supporting documentation in respect of such Excess Liabilities.

**3.4 Withholding.** Purchaser and its agents (as applicable) shall be entitled to deduct and withhold from any amounts payable or otherwise deliverable pursuant to this Agreement and the CVR Agreement such amounts as such Person is required to deduct or withhold therefrom under any applicable Laws and shall pay the amounts withheld to the appropriate Governmental Authority, *provided, however*, that Purchaser use commercially reasonable efforts, together with Seller, to reduce or eliminate the amount of any such deduction or withholding. Purchaser acknowledges that it has no knowledge as of the date hereof, pursuant to current Law, of any obligation to deduct and withhold from any amount otherwise payable pursuant to this Agreement to Seller provided that the IRS Form W-9 described in <u>Section 4.2(a)(iv)</u> is delivered by Seller. To the extent such amounts are so deducted or withheld and remitted to the appropriate Governmental Authority, Purchaser will send proof of Tax payment to Seller on a reasonable and timely basis following such Tax payment. Any amounts deducted and withheld by Purchaser or its agent from payment to Seller shall be treated for all purposes under this Agreement and the CVR Agreement as having been paid to Seller.

## 3.5 Legends.

(a) The Shares shall be placed in a restrictive class bearing a restrictive legend substantially similar to the following:

THE SHARES HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED, OR ANY STATE SECURITIES LAWS. THEY MAY NOT BE SOLD OR OFFERED FOR SALE IN THE ABSENCE OF AN EFFECTIVE REGISTRATION STATEMENT AS TO THE SECURITIES UNDER THE SECURITIES ACT AND ANY APPLICABLE STATE SECURITIES LAW OR AN EXEMPTION FROM SUCH REGISTRATION UNDER THE SECURITIES ACT. THE ISSUER OF THESE SHARES MAY REQUIRE AN OPINION OF COUNSEL REASONABLY SATISFACTORY TO THE ISSUER THAT SUCH OFFER, SALE OR OTHER TRANSFER OTHERWISE COMPLIES WITH THE SECURITIES ACT AND ANY APPLICABLE STATE SECURITIES LAWS.

Additionally, during the Lock-up Period (as defined below), the Lock-up Shares (as defined below) will also bear the restrictive legend in substantially the following form (and a stop-transfer order may be placed against transfer of the Lock-up Shares):

# THE SALE, PLEDGE, HYPOTHECATION OR TRANSFER OF THESE SECURITIES IS SUBJECT TO THE TERMS AND CONDITIONS OF AN ASSET PURCHASE AGREEMENT DATED FEBRUARY 5, 2024, BETWEEN IMMUNOME, INC. AND AYALA PHARMACEUTICALS, INC..

(b) Purchaser shall be entitled to place appropriate legends on the book entries or certificates evidencing any shares of Purchaser Common Stock to be received in the Transactions by Seller to the extent Seller may be considered an "affiliate" of Purchaser for purposes of Rules 144 and 145 under the Securities Act, reflecting the restrictions set forth in Rules 144 and 145 and to issue appropriate stop transfer instructions to the transfer agent for Purchaser Common Stock.

(c) Subject to receipt from Seller by Purchaser and Purchaser's transfer agent (the "Transfer Agent") of customary representations and other documentation reasonably acceptable to Purchaser and the Transfer Agent in connection therewith, Purchaser shall remove any legend from the book entry position evidencing the Shares issued hereunder and Purchaser will, if required by the Transfer Agent, use its commercially reasonable efforts to cause an opinion of Purchaser's counsel be provided, in a form reasonably acceptable to the Transfer Agent, to the effect that the removal of such restrictive legends in such circumstances may be effected under the Securities Act, (i) following the time the Resale Registration Statement is declared effective, or (ii) if such Shares have been sold pursuant to Rule 144 or any other applicable exemption from the registration requirements of the Securities Act. If restrictive legends are no longer required for such Shares pursuant to the foregoing, Purchaser shall, in accordance with the provisions of this Section 3.5(c) and within two (2) Trading Days of any request therefor from Seller accompanied by such customary and reasonably acceptable representations and other documentation referred to above establishing that restrictive legends are no longer required, deliver to the Transfer Agent irrevocable instructions to make a new, unlegended entry for such book entry Shares. Notwithstanding the foregoing, (A) Purchaser shall not be obligated to remove or cause to be removed the Lock-up Legend from the Lock-up Shares prior to the expiration of the Lock-up Period, and (B) promptly following the one (1) year anniversary of the Closing, Purchaser shall remove any legend from the book entry position evidencing the Shares. Seller agrees with Purchaser that Seller will only sell Shares in accordance with either the registration requirements of the Securities Act, including any applicable prospectus delivery requirements, or an exemption therefrom, and that if Shares are sold pursuant to the Resale Registration Statement, they will be sold in compliance with the plan of distribution set forth therein, and acknowledges that the removal of the restrictive legend from certificates representing Shares as set forth in this Section 3.5 is predicated upon Purchaser's reliance upon this understanding.

**3.6 Lock Up**. Seller agrees that it will hold and will not sell greater than 50% of the Shares (the "**Lock-up Shares**") (or otherwise make any short sale of, grant any option for the purchase of, or enter into any hedging or similar transaction with the same economic effect as a sale of the Lock-up Shares) until the date that is six (6) months following the Closing Date (the "**Lock-up Period**"). Notwithstanding the foregoing, this <u>Section 3.6</u> will not preclude (a) distributions of Lock-up Shares to general or limited partners, members, stockholders, Affiliates or wholly-owned subsidiaries of Seller or any investment fund or other entity controlled or managed by Seller; *provided*, in each case, that following any such transfer such Lock-up Shares will remain subject to the provisions of this <u>Section 3.6</u>, or (b) transfers pursuant to a bona fide third party tender offer for all outstanding shares of Purchaser Common Stock, merger, consolidation or other similar transaction made to all holders of Purchaser's securities involving a change of control of Purchaser (including the entering into any lock-up, voting or similar agreement pursuant to which Seller may agree to transfer, sell, tender or otherwise dispose of Lock-up Shares or other such securities in connection with such transaction, or vote any Lock-up Shares or other such securities in favor of any such transaction); *provided* that in the event that such tender offer, merger, consolidation or other such transaction is not completed, the Lock-up Shares shall remain subject to the provisions of this <u>Section 3.6</u>. Following the expiration of the Lock-up Period, Purchaser shall promptly remove, or cause to be removed, the Lock-up Legend from the Lock-up Shares.

**3.7 Orderly Market**. Seller agrees that for one (1) year following the Closing Date (the "**Orderly Trading Period**"), for any transfer of Shares, other than a pro rata distribution to the stockholders of Seller in connection with a dissolution of Seller (which for the avoidance of doubt shall not be prohibited or restricted by this <u>Section 3.7</u>), taking place on any single Trading Day that exceeds 15% of the average daily trading volume of Purchaser Common Stock on Nasdaq over the five (5) Trading Day period ending on the Trading Day immediately prior to such Trading Day, to conduct such Share transfer as a block trade or other disposition through a market participant designated by Purchaser. Notwithstanding the foregoing, this <u>Section 3.7</u> will not preclude sales pursuant to a bona fide third party tender offer for all outstanding shares of Purchaser Common Stock, merger, consolidation or other similar transaction made to all holders of Purchaser's securities involving a change of control of Purchaser (including the entering into any lock-up, voting or similar agreement pursuant to which Seller may agree to transfer, sell, tender or otherwise dispose of Shares or other such securities in connection with such transaction).

**3.8 Rule 144 Reporting.** With a view to making available the benefits of certain rules and regulations of the SEC that may permit the sale of the Closing Shares to the public without registration, the Purchaser agrees to use its commercially reasonable efforts to:

(a) make and keep adequate current public information with respect to Purchaser available in accordance with Rule 144;

(b) file with the SEC in a timely manner all reports and other documents required of Purchaser under the Securities Act and the Exchange Act at any time after it has become subject to such reporting requirements; and

(c) so long as Seller owns any Closing Shares, furnish to Seller forthwith upon written request a written statement by Purchaser as to its compliance with the reporting requirements of Rule 144, and of the Securities Act and the Exchange Act, a copy of the most recent annual or quarterly report of Seller, and such other reports and documents so filed as Seller may reasonably request in availing itself of any rule or regulation of the SEC allowing Seller to sell any such securities without registration.

# 4. CLOSING

**4.1 Closing.** Subject to the satisfaction or waiver of the conditions set forth in <u>Article 7</u> and <u>Article 8</u>, and unless this Agreement shall have been validly terminated in accordance with <u>Article 10</u>, the consummation of the Transactions, including the closing of the sale of the Acquired Assets and the assumption of the Assumed Liabilities pursuant to this Agreement (the "**Closing**") shall take place remotely via the electronic exchange of documents and signatures, on a date (no later than the third (3<sup>rd</sup>) business day after the satisfaction or waiver of the last of the conditions set forth in <u>Article 7</u> and <u>Article 8</u> to be satisfied, other than those conditions that by their nature are to be satisfied at Closing, but subject to the satisfaction or waiver of such conditions) to be agreed upon by Purchaser and Seller. For purposes of this Agreement, "**Closing Date**" means the date on which the Closing actually takes place.

# 4.2 Deliveries at Closing.

(a) Deliveries of Seller. On or before the Closing, Seller shall deliver or caused to be delivered to Purchaser the following:

(i) a duly executed counterpart of the Assignment and Assumption Agreement;

(ii) a duly executed counterpart of the Bill of Sale;

(iii) a duly executed counterpart of the IP and Patent Assignment Agreement;

(iv) a duly executed and valid IRS Form W-9 from Seller;

(v) duly executed counterparts of all approvals, consents, and waivers that are listed on <u>Schedule 4.2(a)(v)</u>, or evidence reasonably satisfactory to Purchaser that all notices listed on <u>Schedule 4.2(a)(v)</u> have been delivered, except to the extent waived by Purchaser;

(vi) evidence, reasonably satisfactory to Purchaser, that all Liens (other than Permitted Liens) relating to any of the Acquired Assets have been removed;

(vii) evidence, reasonably satisfactory to Purchaser, that the Tail Policy has been obtained;

(viii) any letter of transfer required to transfer any Regulatory Filings; and

(ix) evidence, reasonably satisfactory to Purchaser, regarding amounts owed, and credits available, under the IQVIA Agreement.

(b) Deliveries of Purchaser. On or before Closing, Purchaser shall deliver or caused to be delivered to Seller the following:

(i) a duly executed counterpart of the Assignment and Assumption Agreement;

(ii) a duly executed counterpart of the Bill of Sale;

(iii) a duly executed counterpart of the IP and Patent Assignment Agreement; and

(iv) evidence that the notice required under Section 15.4.2 of the BMS License has been duly sent to BMS.

**4.3 Possession.** At the Closing, Seller shall (a) place Purchaser in actual possession and operating control of all Acquired Assets that are tangible assets and all Patent Documents; and (b) deliver possession of all remaining Acquired Assets to Purchaser at Closing or as soon as reasonably practicable following the Closing. Without limiting the generality of the foregoing, Seller, at no additional cost, shall work with Purchaser to transfer all Acquired Assets stored in electronic form wherever stored, in an agreed upon format as soon as reasonably feasible following the Closing and shall provide to Purchaser, at no additional cost, access to all such Acquired Assets as requested by Purchaser, prior to the transfer of such electronic data to Purchaser. Further, Seller, at Seller's expense, shall make personnel reasonably available and shall support the transfer of all technology included with the Acquired Assets to Purchaser. Title to Inventory shall be conveyed by delivery of such Inventory into Purchaser's possession.

## 5. **Representations and Warranties**

**5.1 Representations and Warranties of Seller.** For purposes of this <u>Section 5.1</u>, references to "**Seller**" shall refer to Seller and each Subsidiary of Seller. Except as set forth in the disclosure schedules delivered by Seller on the Agreement Date (the "**Disclosure Schedules**"), or (b) disclosed in Seller SEC Reports other than disclosures contained in any part of any Seller SEC Reports entitled "Risk Factors," "Forward-Looking Statements," "Cautionary Statement Regarding Forward Looking Statements," or "Note Regarding Forward Looking Statements," or any other disclosures in any Seller SEC Reports are cautionary, predictive or forward looking in nature), Seller makes the following representations and warranties to Purchaser as of the Agreement Date and as of the Closing Date (except, in each case, to the extent such representations and warranties speak expressly as of a different date, and then, as of such date) as follows:

(a) Organization and Existence. Seller is a corporation duly organized, validly existing and in good standing under the laws of its jurisdiction of incorporation. Seller has the requisite corporate power and authority to own, use and operate the Acquired Assets as currently conducted. Seller is duly qualified or licensed to do business and in good standing as a foreign corporation in each jurisdiction in which the ownership of the Acquired Assets or the conduct of its business requires such qualification or license, except for those jurisdictions where the failure to be so qualified or licensed and in good standing has not had and would not reasonably be expected to have, individually or in the aggregate, a material adverse effect on the Acquired Assets or the Assumed Liabilities, taken as a whole. Seller has delivered to Purchaser accurate and complete copies of the Organizational Documents of Seller.

(b) Authority; Binding Nature of Agreements. Seller has all requisite power and authority to enter into this Agreement, the Related Agreements and the other agreements, instruments, and documents to be executed and delivered in connection herewith and therewith to which Seller is (or becomes) a party, and, subject to the approval of Seller's stockholders in accordance with Section 6.6, to consummate the Transactions. Except as contemplated by Section 6.6, the execution and delivery of this Agreement and the Related Agreements to which Seller is a party, and the consummation of the Transactions by Seller, including the sale of the Acquired Assets, have been duly authorized by all necessary corporate action, if required, and, except as contemplated by Section 6.6, no further corporate or stockholder action is required on the part of Seller to authorize this Agreement or any Related Agreements to which Seller is a party or the Transactions or for Seller to perform its obligations under this Agreement or any other Related Agreements. This Agreement has been duly executed and delivered by Seller and the other Related Agreements will be duly executed and delivered by Seller, and, assuming the due execution and delivery of this Agreement by Purchaser and of the Related Agreements by the counterparties thereto, this Agreement constitutes, and the Related Agreements when so executed and delivered will each constitute, a valid and legally binding obligation of Seller, enforceable against it in accordance with their respective terms, except as enforceability may be affected by bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium and other similar Laws relating to or affecting creditors' rights generally, and general equitable principles (whether considered in a Legal Proceeding in equity or at Law) (the "Enforceability Exception"). The Stockholder Written Consent, when executed and delivered by the Consenting Stockholders in accordance with Section 6.6, shall be signed by holders of Seller Common Stock having not less than the minimum number of votes necessary to approve the consummation of the Transactions by Seller's stockholders by written consent under applicable Law (including Section 271 of the Delaware General Corporation Law) and Seller's Organizational Documents.

(c) No Conflict. The execution and delivery of this Agreement and each of the Related Agreements to which it is a party does not, and the consummation of the Transactions will not, conflict with or result in any violation of or default under (with or without notice or lapse of time, or both) or give rise to, any payment obligation, or a right of termination, cancellation, modification or acceleration of any obligation or loss of any benefit under (any such event, a "Conflict") (i) any provision of Seller's Organizational Documents, (ii) any Acquired Business Contract, or (iii) any material Law or material Order applicable to the Acquired Assets, or any Product or Compound, except in the cases of clauses (ii) and (iii), where the Conflict, individually or in the aggregate, has not been and would not reasonably be expected to be material and adverse to the Acquired Assets, taken as a whole, or that would not reasonably be expected to prevent, materially impede or materially delay the consummation by Seller of the Transactions. Section 5.1(c) of the Disclosure Schedule sets forth all notices, consents, waivers and approvals of parties to any Acquired Business Contract that are required thereunder in connection with the Transactions or for any Acquired Business Contract to remain in full force and effect with modification from and after the Closing. The execution and delivery of this Agreement and the Related Agreements does not, and the consummation of the Transactions will not result in the creation or imposition of any Lien other than Permitted Liens on the Acquired Assets. Following the Closing, Purchaser will be permitted to exercise all of its rights under the Acquired Business Contracts without the payment of any additional amounts or consideration other than ongoing fees, royalties, or payments that Seller would otherwise have been required to pay pursuant to the terms of such Acquired Business Contracts had the Transactions not occurred.

(d) Governmental Approvals and Filing. No consent, notice, waiver, approval, order, or authorization of, or registration, declaration, or filing with, any Governmental Authority is required by, or with respect to, Seller in connection with the execution and delivery of this Agreement, the Related Agreements or any other agreement to which Seller is a party or the consummation of the Transactions, except for (i) the filings required under Section 6.6, (ii) any approvals in order to close the sale of the Acquired Assets under this Agreement, and (iii) any consent, notice, waiver, approval, order, or authorization of, or registration, declaration, or filing that has not been and would not reasonably be expected to be material and adverse to the Acquired Assets, taken as a whole, or that would not reasonably be expected to prevent, materially impede or materially delay the consummation by Seller of the Transactions.

#### (e) Taxes.

(i) All income and other material Tax Returns required to be filed by Seller or its Affiliates with respect to the Acquired Assets and Product Operations have been filed and such Tax Returns are true and correct in all material respects. All Taxes due and owing by Seller (whether or not shown on a Tax Return) attributable to the Acquired Assets or Product Operations have been paid. No power of attorney that is currently in effect has been granted by Seller with respect to the Acquired Assets or the Product Operations (other than powers of attorney granted in the ordinary course of business, such as to a payroll provider).

(ii) The Seller has (i) complied in all respects with all applicable laws relating to the payment, reporting and withholding of Taxes (including withholding of Taxes pursuant to Sections 1441, 1442, 1445, 1446, 1471, 1472, and 3406 of the Code or similar provisions under any foreign Law), (ii) withheld (within the time and in the manner prescribed by applicable Law) from employee wages or consulting compensation and paid over to the proper Governmental Authorities (or is properly holding for such timely payment) all amounts required to be so withheld and paid over under all applicable Law, including federal and state income Taxes, Federal Insurance Contribution Act, Medicare, Federal Unemployment Tax Act, relevant state income and employment tax withholding Laws, and (iii) timely filed all withholding Tax Returns, for all periods through and including the Closing, to the extent any such matter described in sub-clauses (i), (ii) or (iii) of this clause (e)(ii) relates to the Acquired Assets or the Product Operations.

(iii) There are no Liens for unpaid Taxes on the Acquired Assets (other than Liens for Taxes not yet due and payable). There is no Tax Contest pending or, to the Knowledge of Seller, threatened, that relates to the Acquired Assets.

(iv) Seller is not a "foreign person" within the meaning of Sections 1445 or 1446 of the Code. Seller is and always has been properly treated as a domestic corporation for U.S. federal and applicable state and local income Tax purposes.

(v) The Acquired Assets do not include any tax-sharing agreements, United States real property assets, or stock or other ownership interests in any corporations, partnerships, joint ventures, limited liability companies, business trusts, or other entities.

(vi) None of the Acquired Assets are (i) property required to be treated as being owned by another Person pursuant to the provisions of Section 168(f)(8) of the Internal Revenue Code of 1954, as amended and in effect immediately prior to the enactment of the Tax Reform Act of 1986, (ii) "tax-exempt use property" within the meaning of Section 168(h)(1) of the Code, (iii) "tax-exempt bond financed property" within the meaning of Section 168(g)(1)(A) of the Code, or (v) subject to a "section 467 rental agreement" as defined in Section 467 of the Code.

(vii) No jurisdiction in which Seller does not file Tax Returns in respect of the Acquired Assets, or the Product Operations, has ever asserted in writing that Seller may be required to file a Tax Return in such jurisdiction with respect to the Acquired Assets or the Product Operations. No Taxes have attached to the Acquired Assets and will become a liability of Purchaser as a result of the execution of this Agreement and purchase of the Acquired Assets.

(viii) Seller has not received any written notice from any Governmental Authority of any Tax deficiency, Tax examination, or other Tax proceeding that relates to Taxes attributable to the Product Operations or the Acquired Assets, which deficiency, examination, or proceeding has not been resolved in full. Seller has not waived any statute of limitations in respect of Taxes attributable to the Product Operations or the Acquired Assets, which waiver is currently in effect.

(f) Title. None of the Acquired Assets is subject to any Liens (including Tax-related Liens), other than Liens of the type described in clauses (a)(i) and (c) of the definition of Permitted Liens. Seller has good and marketable title to all of the Acquired Assets (whether real or personal, tangible) and will transfer to Purchaser good and marketable title to all Acquired Assets at the Closing, free and clear of any Liens other than Permitted Liens. Following the Closing, no Affiliate of Seller will have title to or rights under any of the Acquired Assets.

## (g) Compliance with Laws.

(i) Seller is in compliance in all material respects with, and since the Look-Back Date has complied in all material respects with, all Laws and Orders applicable to the Acquired Assets and the conduct of the Product Operations;

(ii) Seller, and to the Knowledge of Seller, its Affiliates and each Person acting on their behalf, is and since the Look-Back Date has been, in compliance, in all material respects, with (A) all applicable Laws relating to the privacy, data protection and security of any personal information, including HIPAA, (B) all privacy, data protection and security policies of Seller concerning patient medical records and other personal information, and (C) any contractual requirements to which Seller and its Affiliates are subject that relate to any of the foregoing. The execution, delivery and performance of this Agreement and the consummation of the Transactions will comply, in all material respects, with all applicable (1) Laws, (2) policies of Seller concerning patient medical records and other personal information, and (3) contractual requirements to which Seller or its Affiliates are subject, in each case (1) through (3) relating to privacy, data protection and security of personal information; and

(iii) Since the Look-Back Date, Seller has received no written notification or communication from any Governmental Authority (A) asserting that Seller is not in compliance with any Law or Order with respect to the Acquired Assets or (B) threatening to revoke or suspend any Transferred Permit owned or held by Seller relating to the Acquired Assets or the Development, manufacturing of Commercialization of any Products or Compounds.

(h) Permits. Section 5.1(h) of the Disclosure Schedules contains a complete list of all material Permits that constitute Acquired Assets (the "Transferred Permits"). Seller possesses all such Transferred Permits, and each such Transferred Permit is validly and presently in effect (and the continuing validity and effectiveness of such Transferred Permit will not be affected by the consummation of the Closing), and Seller is not in default (with or without notice or lapse of time, or both) under any Transferred Permit in any material respect. There are no Legal Proceedings pending, nor to the Knowledge of Seller, threatened, that seek the revocation, cancellation, suspension, failure to renew or materially adverse modification of any such Transferred Permit. Since the Look-Back Date, all required filings with respect to each such Transferred Permit have been timely made and all required applications for renewal thereof have been timely filed.

## (i) Material Contracts.

(i) Section 5.1(i)(i) of the Disclosure Schedules sets forth a true and accurate list of each Contract (other than purchase orders issued by Seller to a Third Party that are ancillary to another written Contract with the same Third Party and that do not constitute an Assumed Liability) in effect as of the date of this Agreement to which Seller is a party or which was entered into by or on behalf of Seller, or by which any of the Acquired Assets is bound in the following categories (the "Material Contracts"):

(A) any Contract establishing a joint venture or collaboration, co-promotion or like arrangement, or involving a sharing with another Person of profits, losses, costs, royalties, milestone payments, or Liabilities of Seller relating to the Acquired Assets or the Development, manufacture, or Commercialization of any Compound or Product, including the conduct of any clinical trials;

(B) any Contract containing covenants prohibiting or limiting the right to compete or engage in any aspect of the Product Operations or prohibiting or restricting Seller's ability to conduct the Product Operations with any Person or in any geographical area;

(C) any Contract granting most favored nation or exclusive rights relating to any Compound or Product to any other

Person;

(D) any Contract pursuant to which Seller has obtained or granted any Intellectual Property Rights included in the Acquired Assets (or that would have been included in the Acquired Assets but for such Contract), including any covenant not to enforce or assert, including any existing license agreement relating to any Compound or Product or the Product Operations and each other Contract under which Seller is a licensor or licensee of any Intellectual Property Rights relating to any Compound or Product or the Product Operations other than any of the following entered into in the Ordinary Course of Business and, in each case, that are not Acquired Business Contracts and are deemed Excluded Liabilities: (i) Nondisclosure Agreements; (ii) services agreements containing non-exclusive licenses to Intellectual Property Rights included in the Acquired Assets for the sole purpose of a service provider performing services for or on behalf of Seller; (iii) agreements with clinical investigators and clinical sites for the conduct of a clinical study, which study is complete or substantially complete at the relevant clinical sites as of the date of this Agreement; (iv) licenses to commercially available software or cloud or software as a service agreements; and (v) assignment agreements with employees, including proprietary information and invention assignment agreements with employees;

Product or any Product IP;

(E) any Contract under which Seller pays or receives milestone or royalty payments relating to any Compound or

(F) any Contract relating to the creation of Liens on any Acquired Assets or the guarantee of the payment of Liabilities or performance of obligations of any other Person by Seller relating to any Compound or Product or any Acquired Assets;

(G) any Contract entered into by Seller or any of its Affiliates in settlement of any Legal Proceeding or other dispute relating to the Acquired Assets or the Product Operations, including the conduct of any clinical trials;

(H) any Contract that limits Seller's ability to make generally available any versions of any Compound or Product

developed by or for Seller;

(I) any Contract for the research or Development of any Compound or Product, other than any of the following entered into in the Ordinary Course of Business and, in each case that are not Acquired Business Contracts and are deemed Excluded Liabilities: (i) standalone indemnity arrangements with clinical trial sites or clinical trial investigators; (ii) powers of attorney, letters of delegation, declarations and similar instruments executed by Seller in connection with the regulatory and ethics committee submissions and data processing activities for clinical studies outside of the U.S. (other than local representative agreements and legal representative agreements; (iv) licenses to commercially available software or cloud or software as a service agreements; and (v) service agreements with quality assurance auditors, meeting planners, Third Parties providing meeting support services, and non-physician advisory board participants (i.e. nurse advisors);

(J) any Contract for the development, manufacture, supply, packaging, labeling, distribution, analytical testing, or storage of the active pharmaceutical ingredients and other raw materials for any Compound or Product, and related quality agreements;

(K) any Contract for the ongoing or planned analytical testing or storage of biological specimens collected from subjects participating in clinical trials of any Compound or Product;

Product;

(L) any Contract for the distribution, promotion, marketing, reselling or other Commercialization of any Compound or

(M) any Contract for the maintenance of the safety database for any Compound or Product, and any safety data exchange agreements or pharmacovigilance agreements related to any Compound or Product;

(N) any Contract with any Governmental Authority relating to any Compound or Product or any of the Acquired Assets, other than clinical trial agreements and related ancillary agreements with public institutions; and

(O) any other Contract that is material to the Development, manufacture or sale of any Compound or Product, in each case, as currently conducted by Seller, other than any Contract relating to (i) real property, (ii) employees, or employee compensation or benefit matters, including any Employee Benefit Plan, (iii) indebtedness, other than indebtedness associated with any Lien on any Acquired Asset, (iv) general administration expenses, or (v) insurance.

(ii) All of the Material Contracts are valid and binding agreements of Seller, enforceable in accordance with their terms, subject to the Enforceability Exception. Other than Material Contracts entered into on behalf of Seller, Seller has made available or delivered to Purchaser a correct and complete copy of each written Material Contract. Seller is not in material breach or material default of any of the Material Contracts or Nondisclosure Agreements, and no event has occurred that with notice or lapse of time, or both, would constitute a material default of such Material Contract and no event has occurred that with notice or lapse of time, or both, would constitute a material default of such Material Contract and no event has occurred that with notice or lapse of time, or both, would constitute a material default of such Material Contract or Nondisclosure Agreement. No party has repudiated in writing or, to the Knowledge of Seller, otherwise provided notice of its intention to repudiate any provision of a Material Contract or Nondisclosure Agreement. Seller has not given to or received from any other Person any written, or to the Knowledge of Seller other, notice regarding any material violation or breach of, or default under, any Material Contract or Nondisclosure Agreement.

(iii) As of the date hereof, (A) the Liabilities of Seller relating to the IQVIA Agreement are \$3,800,000, which amount is net of all deposits (including the deposit referenced in clause (B)), advances and prepayments, and (B) Seller has a credit available under the IQVIA Agreement in an amount equal to \$2,480,000.

(iv) Section 5.1(i)(iv) of the Disclosure Schedules sets forth an accurate and complete list of all outstanding accrued trade payables under the Acquired Business Contracts as of the date hereof.

(j) Inventory. Section 5.1(j) of the Disclosure Schedule lists Seller's clinical supply Inventory as of January 30, 2024. All of the Inventory has been manufactured, stored and otherwise maintained in accordance with Seller's manufacturing practices and with applicable Laws, in each case, in all material respects, and is in usable condition for the Development of any Product or Compound as intended, subject to its shelf life. Other than as set forth on Section 5.1(j) of the Disclosure Schedule, the Inventory is in a quantity with a shelf-life and with available back-up supply that is sufficient for Seller to conduct and complete the RINGSIDE Part B Study in the Ordinary Course of Business.

#### (k) Legal Proceedings.

(i) As of the Agreement Date, there are no Legal Proceedings pending against or, to the Knowledge of Seller, threatened, pertaining to the Product Operations, any Compound or Product, or any of the Acquired Assets, and there is no investigation (internal or external), audit or other Legal Proceeding pending or, to the Knowledge of Seller, threatened, pertaining to the Product Operations, any Compound or Product, or any of the Acquired Assets or any of the executive officers or directors of Seller, in each case, in relation to any Compound or Product or any of the Acquired Assets, by or before any Governmental Authority.

(ii) There is no Order to which Seller is subject or that is pending or, to the Knowledge of Seller, threatened that relates to the Acquired Assets or the Assumed Liabilities.

# (I) Regulatory Compliance.

(i) Since the Look-Back Date, Seller has not been in violation of, and is not the subject of any Legal Proceeding with respect to the violation of, any Law or Order, and has not received any FDA Form 483, "warning letters", or "untitled letters", or other similar Governmental Authority notice of inspectional observations or deficiencies relating to the Product Operations. Since the Look-Back Date, no Compound or Product has been subject to any import detention or refusal by the FDA or other similar Regulatory Authority or any safety alert issued by the FDA or other similar Regulatory Authority. No Legal Proceeding is pending, or to the Knowledge of Seller, threatened, with respect to any violation of any Law or Order by Seller, or to the Knowledge of Seller, and Seller is and since the Look-Back Date, has been in compliance in all material respects with all Laws and Orders, in each case pertaining to the Product Operations. Since the Look-Back Date, Seller has not received any notice of any such Legal Proceeding or any Liability on the part of Seller to undertake or to bear all or any portion of the cost of any Product Operations remedial action of any nature. Seller has filed, maintained or submitted all material reports, documents, forms, notices, applications, records, claims, submissions and supplements or amendments were materially complete and accurate on the date filed (or were corrected or supplemented by a subsequent submission).

(ii) Except to the extent not required to be conducted in accordance with GMP, all manufacturing operations and the manufacture of any Products and Compounds by, or on behalf of, Seller are being conducted and since the Look-Back Date, have been conducted, in material compliance with applicable Laws and Orders and in accordance with GMP. The processes used to produce the Compounds and Products are adequate to ensure that the Compounds and Products will conform to the specifications established therefor at the time of production. Since the Look-Back Date, Seller has not received any material written complaints about the Compounds or Products. Since the Look-Back Date, Seller has not conducted any recoveries or recalls of Products or Compounds.

(iii) Since the Look-Back Date, all preclinical testing and clinical trials in respect of the Product Operations conducted by or on behalf of Seller have been conducted in all material respects in accordance with experimental protocols, procedures, and controls, as well as pursuant to applicable Laws (including GCPs and GLPs) and Orders, as applicable. Since the Look-Back Date, no clinical trial in respect of any Product or Compound has been placed on clinical hold, terminated, materially delayed, limited or suspended prior to completion by the FDA or any similar Regulatory Authority, or by any institutional review board that has or has had jurisdiction over such clinical trial, and neither the FDA nor any other applicable Regulatory Authority, nor any institutional review board that has or has had jurisdiction over such clinical trial has ordered or commenced, or, to the Knowledge of Seller, threatened to initiate, any such action or alleged any violation of any Law in connection with any such clinical trial. Seller has provided Purchaser with true and correct copies of final study reports of all clinical trials in respect of any Products or Compounds that were completed prior to the Agreement Date.

(iv) All human clinical trials conducted by or on behalf of Seller that are intended to be submitted to Governmental Authorities to support regulatory approval of the Compounds and Products are being conducted in compliance in all material respects with applicable GCP regulations and guidance, and all applicable Laws relating to protection of human subjects, including those contained in 21 CFR Parts 50, 54, 56 and 312, and any comparable state and foreign Laws. All required approvals and authorizations for clinical trials to proceed have been obtained from an appropriate institutional review board, and informed consent, in material compliance with applicable Laws, has been obtained from all subjects enrolled in the studies.

(v) Since the Look-Back Date, Seller has complied in all material respects with all adverse event reporting requirements applicable to the Products and Compounds. Section 5.1(l)(v) of the Disclosure Schedules identifies all serious adverse events and suspected adverse reactions that occurred during any clinical trial in respect of the Product Operations, as well as any serious and unexpected suspected adverse reactions that were the subject of an IND safety report.

(vi) Seller is not the subject of any pending or, to the Knowledge of Seller, threatened investigation by any Governmental Authority in respect of Seller or the Product Operations, including by the FDA pursuant to its "Fraud, Untrue Statements of Material Facts, Bribery, and Illegal Gratuities" Final Policy set forth in 56 Fed. Reg. 46191 (September 10, 1991) and any amendments thereto or any other Governmental Authority that has jurisdiction over the operations of Seller under any similar policy. Neither Seller nor, to the Knowledge of Seller, any of its officers, employees, or agents acting for Seller, has since the Look-Back Date committed any act, made any statement or failed to make any statement, relating to the Product Operations that would reasonably be expected to provide a basis for the FDA to invoke its policy with respect to "Fraud, Untrue Statements of Material Facts, Bribery, and Illegal Gratuities" set forth in 56 Fed. Reg. 46191 (September 10, 1991) and any amendments thereto. Since the Look-Back Date, neither Seller nor any current or, to the Knowledge of Seller, former officer or employee, or, to the Knowledge of Seller, agent of Seller with respect to the Product Operations has been convicted of any crime or engaged in any conduct that would reasonably be expected to result in exclusion under 42 U.S.C. Section 1320a-7 or any similar state law or regulation or been debarred by the FDA under Article 306 of the FDC Act, 21 U.S.C. §335a(a) or (b), or any similar foreign or local law, rule, or regulation.

(vii) The Compounds and Products are being and have been researched, Developed, manufactured, labeled, stored, and tested in compliance in all material respects with all applicable requirements under the FDC Act, the Public Health Service Act, and their applicable implementing regulations, and all comparable state and foreign Laws in those jurisdictions outside the United States in which either (A) human clinical trials involving the Compounds and Products have been or are being conducted by Seller or (B) the Compounds and Products have been or are being manufactured by or for Seller.

(viii) Neither Seller nor, to the Knowledge of Seller, any representative agent, contract manufacturing organization, contract research organization, clinical investigator or clinical trial site acting on behalf of Seller nor any of its licensors, licensees or assignees of Product IP has received any written notice that the FDA or any other Regulatory Authority has initiated, or threatened to initiate, any action to suspend any clinical trial sponsored by Seller with respect to the Compounds and Products, or to recall or suspend the manufacture of the Compounds and Products.

(ix) Seller has made available to Purchaser copies of any and all written notices of inspectional observations, establishment inspection reports, and any other documents received by Seller since the Look-Back Date and prior to the date of this Agreement from the FDA or comparable foreign Regulatory Authorities that identify lack of compliance with Laws of the FDA or comparable foreign Regulatory Authorities with respect to Product Operations.

(x) There are no proceedings pending or, to the Knowledge of Seller, threatened, relating to the Product Operations with respect to a violation by Seller or by any representative, agent, contract manufacturing organization, contract research organization, clinical investigator or clinical trial site acting on behalf of Seller, of the FDC Act, FDA regulations adopted thereunder, the Controlled Substances Act, federal or state whistleblower statutes, or any other Law promulgated by any other Governmental Authorities.

## (m) Intellectual Property.

(i) Section 5.1(m)(i) of the Disclosure Schedules identifies (A) all registered Product IP, (B) all applications for the registration of Product IP, and (C) all licenses, sublicenses, and other Contracts (excluding any agreement with an employee or contractor of Seller or its Affiliate in the Ordinary Course of Business that provides for the general assignment of inventions made in the course of employment by or providing services to Seller or its Affiliate) (whether royalty bearing or non-royalty bearing) under which Seller is granted rights by others in any Product IP.

(ii) Section 5.1(m)(ii) of the Disclosure Schedules identifies all License Grants in the Product IP, other than non-exclusive License Grants included in services agreements entered into in the Ordinary Course of Business for the sole purpose of a service provider performing services for or on behalf of Seller ("Non-Scheduled License Grants") and confidentiality agreements entered into in the Ordinary Course of Business.

(iii) Except for the rights identified in Section 5.1(m)(i)(C) of the Disclosure Schedules and subject to the License Grants identified in Section 5.1(m)(ii) of the Disclosure Schedules, Seller owns and possesses all right, title, and interest in and to, free and clear of all Liens (other than Permitted Liens), all of the Seller IP. There are no Intellectual Property Rights and there is no Know-How owned or licensed (including by sublicense, covenant-not-to-assert, or any similar right or authorization) that are necessary to the Product Operations other than the Product IP. All previously due fees associated with maintaining any Seller IP or other Product IP for which Seller is responsible for filing, prosecution or maintenance have been paid in full in a timely manner to the proper Governmental Authority, except where Seller has decided, to the extent described in Section 5.1(m)(iii) of the Disclosure Schedules, abandon or cancel such Product IP (including any Intellectual Property Rights that would have been Product IP but for such abandonment or cancelation). Upon the Closing, all of the Product IP shall be available for use by Purchaser on terms and conditions substantially identical to those under which Seller owned or used the Product IP immediately prior to the Closing. To the Knowledge of Seller, no Product IP is subject to any outstanding consent or Order restricting the use thereof.

(iv) Seller has not received any written claim by any Third Party and, to Seller's Knowledge, there is no claim threatened against Seller contesting the validity, enforceability, or ownership of any Product IP. The validity or enforceability of registered Product IP or rights of Seller to use Product IP has not been challenged in any jurisdiction. Each item of registered Product IP is valid, subsisting, in full force and effect and, to the Knowledge of Seller, enforceable; and has not been cancelled, expired, or abandoned except where Seller has decided to abandon or cancel such Product IP to the extent described in Section 4.1(m)(iii) of the Disclosure Schedules. No claim is pending or, to Seller's Knowledge, threatened challenging Seller's right to any Product IP or right to use any Product IP. No claim is pending or, to Seller's Knowledge, threatened to the effect that any registered Product IP is, or upon consummation of the Transactions will be, invalid or unenforceable.

(v) To the Knowledge of Seller, Seller has not infringed, misappropriated, diluted, violated, or otherwise conflicted with any Intellectual Property Rights of any Third Party in conducting the Product Operations. To the Knowledge of Seller, the Product Operations, including the use of the Compounds and Products, the manufacture, use, sale, and importation of the Compounds and Products, the possession, use, disclosure, copying, or distribution of any information, data, or other tangible or intangible assets in the possession of Seller regarding any Compound or Product included in the Acquired Assets, and the possession or use by Seller of the Product IP, has not and does not; and, if possessed and used in the same manner as Seller has in conducting the Product Operations, will not infringe, misappropriate, dilute, violate, or otherwise conflict with any Intellectual Property Rights of any other Person. Seller has not received any written notice of any claim (including by an offer to license any Intellectual Property Rights) and Seller has not received any written claim and, to Seller's Knowledge, there is no threatened claim against Seller asserting that the Product Operations, or the manufacture, use, sale, and importation of any Compound or Product, may infringe, misappropriate, dilute, violate, or otherwise conflict with the Intellectual Property Rights of any Person.

(vi) To Seller's Knowledge, none of the Product IP is being infringed by any Third Party other than by the License Grants as set forth on Section 5.1(m)(ii) of the Disclosure Schedules, the Non-Scheduled License Grants or confidentiality agreements entered into in the Ordinary Course of Business. Seller has not given any notice to any Person asserting infringement, misappropriation, dilution, violation or conflict by any such Person of any of the Product IP. There are no pending, or, to Seller's Knowledge, threatened or potential, claims of infringement, misappropriation, dilution, conflict, or violation of the Product IP by any Person.

(vii) Seller has obtained from all individuals who participated in any respect in the invention or authorship of any Seller IP effective written assignments of all ownership rights of such individuals in such Seller IP, and, to Seller's Knowledge, no Person who as of the Agreement Date claims to be an inventor of an invention claimed in any registered Seller IP is not identified as an inventor of such invention in the filed patent documents for such registered Seller IP.

(viii) Seller and its Affiliates have taken all reasonable measures to protect the secrecy, confidentiality, and value of all Product Know-How that constitutes trade secrets under applicable Laws (including requiring all employees, consultants, and independent contractors to execute agreements requiring all such employees, consultants, and independent contractors to maintain the confidentiality of such Product Know-How) and to Seller's Knowledge, such Product Know-How has not been used by, or disclosed to, any Third Party except pursuant to such confidentiality agreements and there has not been a breach by any party to such confidentiality agreements.

(ix) Neither Seller nor any of its Affiliates has entered into a government funding relationship that would result in any payment obligations to any Governmental Authority or any rights to any Product or Acquired Asset residing in any Governmental Authority and neither the Product nor the Product Operations are subject to overriding obligations to the United States federal government as set forth in Public Law 96 517 (35 U.S.C. § 200 -204), or any similar obligations under the applicable Laws of any state or other country.

1 0 11

(x) To Seller's Knowledge, the Product Patents constitute all Patents covering any Compounds or Products that are Controlled

by Seller.

(xi) Seller and its Affiliates have the exclusive right to use all data generated in the course of, or as a result of, any clinical trial or other testing in humans conducted by or on behalf of Seller in respect of any of the Compounds or Products, except for (A) non-exclusive research licenses granted in the Ordinary Course of Business to a Third Party university or institution that generated such data for non-commercial, medical and academic purposes, or to Third Party service providers or vendors solely to enable the performance of services on Seller's behalf, or (B) data that constitutes or is included in the medical records of any individual.

(xii) Seller is not bound by any Contract to indemnify, defend, hold harmless, or reimburse any other Person with respect to, or otherwise assumed or agreed to discharge or otherwise take responsibility for, any existing intellectual property infringement, misappropriation, or similar claim.

(n) No Brokers. No broker, finder, or investment banker is entitled to any brokerage commission, finder's fee or similar payment in connection with the Transactions based upon arrangements made by or on behalf of Seller.

(o) Affiliate Transactions. Neither Seller nor any of its Affiliates, nor, to the Knowledge of Seller, any current director, officer, employee or any beneficial owner of 5% or more of Seller's outstanding capital stock of Seller, (i) has any direct or indirect financial interest (excluding in their capacity as an employee or stockholder of Seller) (A) in, or is a director or officer of, any Person that is a material client, customer, supplier, lessor, lessee, debtor, creditor or competitor of Seller in respect of the Acquired Business Contracts or (B) in any material property, asset, or right that is owned or used by or on behalf of Seller exclusively in the conduct of the Product Operations or (ii) is, or since the Look-Back Date has been, a party to any Acquired Business Contract.

(p) Certain Product Operations Activities. Since the Look-Back Date, neither Seller nor any of its officers, directors, and to the Knowledge of Seller, its employees, agents or representatives, or any Affiliate of or any Person associated with or acting for or on behalf of Seller, has directly or indirectly, acting for or on behalf of Seller, in each case, in connection with the Product Operations:

(i) used any funds for unlawful contributions, gifts, or entertainment or other unlawful payments, including having made or attempted to make any improper contribution or gift, bribe, rebate, payoff, influence payment, kickback, or other improper payment to any Person, private or public, regardless of what form, whether in money, property, or services to (A) obtain favorable treatment for business or Contracts secured, (B) pay for favorable treatment for business or Contracts secured, or (C) obtain special concessions or for special concessions already obtained, in each of clauses (A), (B), and (C) in violation of any requirement of applicable Law;

(ii) made or attempted to make any such contribution or gift, bribe, rebate, payoff, influence payment, kickback, or other improper payment in violation of any applicable written policy of Seller;

(iii) established or maintained any fund or asset for the purpose of making any such contribution or gift, bribe, rebate, payoff, influence payment, kickback, or other improper payment in violation of any applicable Law or applicable written policy of Seller and which Seller or any of its officers, directors, or employees has willfully failed to record in the Regulatory Documentation. To the extent required by applicable Law, Seller has established and maintains a compliance program and reasonable internal controls and procedures that, for all periods prior to the Closing, were appropriate to satisfy the requirements of applicable Anti-Corruption Laws; or

(iv) consummated any transaction, made any payment, entered into any contract or arrangement or taken any other action in violation of Section 1128B(b) of the U.S. Social Security Act.

(q) No Government Officials. To the Knowledge of Seller, no officer or director of Seller is or since the Look-Back Date has been a foreign or domestic government official or employee or a candidate for any foreign or domestic political office.

(r) Books and Records. Since the Look-Back Date, Seller has made and kept its books and records, including the Regulatory Documentation and the Patent Documents, which, in reasonable detail, accurately and fairly reflect the activities of the Product Operations in all material respects. Since the Look-Back Date, Seller has not engaged in any material transaction, maintained any bank account, or used any corporate funds in connection with the Product Operations, except as reflected in its normally maintained books and records.

(s) Solvency. Seller currently owns property that has a fair saleable value greater than the amounts required to pay its debts (including a reasonable estimate of the amount of all contingent liabilities). No transfer of property is being made and no obligation is being incurred in connection with the Transactions with the intent to hinder, delay or defraud either present or future creditors of Seller.

(t) Sufficiency of Acquired Assets. The Acquired Assets are sufficient for the continued conduct of the Product Operations by Purchaser after the Closing Date in substantially the same manner as currently conducted by Seller and constitute all of the material rights, property, and assets necessary to conduct the Product Operations as currently conducted by Seller, in each case, other than employees, contractors, real estate and Employee Benefit Plans necessary to conduct the Product Operations.

(u) No Transfer of Employees. Unless Purchaser and Seller otherwise agree in writing, no employees of Seller will be transferred to Purchaser in connection with the Transactions.

(v) Absence of Changes. Since December 31, 2023, (i) there has not been any condition, change, effect, event occurrence, state of facts or development that, individually or in the aggregate, has resulted in or would be reasonably expected to result in a Material Adverse Effect, (ii) there has not been any material loss, damage or destruction to, or any material interruption in the use of, any Acquired Assets, and (iii) Seller has not taken any action that would have required the consent of Purchaser pursuant to Section 6.3 if taken after the Agreement Date.

(w) Insurance. Seller has the insurance of the types and in the amounts set forth in Section 5.1(w) of the Disclosure Schedules (the "Insurance Policies") in respect of the Acquired Assets. The Insurance Policies are in full force and effect and all premiums due and payable under such Insurance Policies have been paid on a timely basis. As of the Agreement Date, there is no material claim relating to the Acquired Assets pending under the Insurance Policies as to which coverage has been questioned, denied or disputed by the underwriters of such policies. Seller is in compliance in all material respects with the terms of such policies. To Sellers' Knowledge as of the Agreement Date, there is no threatened termination of, or material premium increase with respect to, any of such policies.

# (x) SEC Reports; Financial Statements.

(i) Since January 1, 2023, Seller has timely filed or furnished each of the Seller SEC Reports. As of their respective dates, after giving effect to any amendments or supplements thereto, the Seller SEC Reports (A) complied as to form in all material respects with the requirements of the Securities Act and the Exchange Act, as applicable, and, to the extent applicable, Sarbanes-Oxley Act of 2002 ("SOX"), and (B) did not contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading, *provided*, *however*, that no representation is made as to the accuracy of any financial projections or forward-looking statements or the completeness of any information filed or furnished by Seller to the SEC solely for the purposes of complying with Regulation FD promulgated under the Exchange Act. As of the Agreement Date, there are no material outstanding or unresolved comments in comment letters from the SEC staff with respect to any of the Seller SEC Reports.

(ii) The audited financial statements and unaudited interim financial statements (including, in each case, the notes, if any, thereto) included in the Seller SEC Reports (the "Seller Financial Statements") complied in all material respects with the published rules and regulations of the SEC with respect thereto in effect at the time of filing or furnishing the applicable Seller SEC Report, were prepared in accordance with GAAP, applied on a consistent basis during the periods involved (except as may be indicated therein or in the notes thereto and except with respect to unaudited statements as permitted by the SEC on Form 8-K, Form 10-Q or any successor or like form under the Exchange Act) and fairly present in accordance with GAAP (subject, in the case of the unaudited interim financial statements, to normal, recurring year end audit adjustments that would not be, individually or in the aggregate, materially adverse to Purchaser) in all material respects the financial position of Seller, as of the respective dates thereof, and the consolidated results of their operations and cash flows for the respective periods then ended.

(y) Opinion of Financial Advisor. Prior to the execution of this Agreement, the Board of Directors of Seller has received an opinion from Alliance Global Partners (the "Opinion") to the effect that, as of the date of the Opinion and subject to certain assumptions, qualifications, limitations and other matters set forth therein, the Consideration to be received by Seller in the Transaction is fair, from a financial point of view, to Seller, and such opinion has not been withdrawn, revoked or modified as of the date of this Agreement. Promptly following the execution of this Agreement, Seller shall deliver or make available to Purchaser a copy of the Opinion for informational purposes only.

(z) HSR Act. Seller (or if different, the Acquired Person as defined in the HSR Act regulations) does not satisfy the higher size of person jurisdictional threshold under the HSR Act.

#### (aa) Investment Representations.

(i) Seller acknowledges and understands that its investment in the Shares involves substantial risk and, if and when issued by Purchaser in accordance with this Agreement, (A) will not be registered for sale under the Securities Act or any other applicable securities Laws, (B) will not be traded on any national exchanges as of the date of this Agreement, and (C) may not be offered, sold or otherwise transferred except in compliance with the registration requirements of the Securities Act and any other applicable securities Laws or pursuant to an exemption therefrom, and in each case in compliance with the conditions set forth in this Agreement.

(ii) Seller acknowledges and understands that it is acquiring the Shares for its own account, for investment purposes only and not with a view toward, or for sale in connection with, any distribution thereof, or with any present intention of distributing or selling any Shares, in each case, in violation of the federal securities laws or any other applicable Law.

(iii) Seller represents that it is an "Accredited Investor" as that term is defined in Rule 501 of Regulation D promulgated under

the Securities Act.

(iv) Seller understands and agrees that the Shares issuable pursuant to this Agreement may not be transferred, sold, offered for sale, pledged, hypothecated or otherwise disposed of without registration under the Securities Act and any other provision of applicable United States federal, United States state, or other Law or pursuant to an applicable exemption therefrom.

(v) Seller acknowledges and agrees that the book entries or certificates evidencing the Shares shall bear a restrictive legend substantially similar to the legend set forth in Section 3.5(a).

(vi) Seller represents that it has received all the information that it considers necessary or appropriate for deciding whether to acquire Shares and has had the opportunity to ask questions and receive answers from Purchaser regarding the Shares and the business, properties, prospects, and financial condition of Purchaser and the terms and conditions of the transactions contemplated hereby and to obtain such additional information (to the extent Purchaser possessed such information or could acquire it without unreasonable effort or expense) necessary to verify the accuracy of any information furnished to Seller or to which Seller had access. Seller represents that it is a sophisticated and experienced investor and is capable of evaluating, to its satisfaction, the accounting, tax, financial, legal and other risks associated with Transactions, and has retained outside legal counsel and has had the opportunity to consult with his, her or its accounting, tax, financial and legal advisors to be able to evaluate the risks involved in the Transactions and to make an informed investment decision with respect to the Transactions.

(bb) Disclaimer of Warranties. Except as set forth in this Section 5.1 (as qualified or modified by the Disclosure Schedules), (i) the Acquired Assets are being sold on an "as is, where is" basis as of the Closing and (ii) none of Seller, its Affiliates or any of their respective Representatives have made, or shall be deemed to have made, any other representation or warranty, express or implied, at law or in equity, in respect of the Acquired Assets or the Assumed Liabilities, including with respect to (A) merchantability or fitness for any particular purpose, (B) the Product Operations after the Closing, (C) the probable success or profitability of the Product Operations after the Closing or (D) the accuracy or completeness of any (1) projections, predictions, forecasts, estimates, plans or budgets of future revenues, expenses or expenditures, future results of operations (or any component thereof), future cash flows (or any component thereof) or future financial condition (or any component thereof) that may be contained or referred to in the Disclosure Schedules or elsewhere, or (2) information, documents or materials regarding the business of Seller, the Acquired Assets or the Assumed Liabilities, including any information furnished or made available to Purchaser, its Affiliates or their respective Representatives in any confidential information memorandum or presentation or material for the purposes of enabling Purchaser to evaluate the consummation of the Transactions. Any such other representations or warranties are hereby expressly disclaimed. Except in the case of Fraud, none of Seller, its Affiliates or any of their respective Representatives will have or be subject to any liability or obligation to Purchaser or any other Person resulting from the distribution to Purchaser, its Affiliates or their respective Representatives of any of the Transactions.

**5.2 Representations and Warranties of Purchaser.** Purchaser makes the following representations and warranties to Seller as of the Agreement Date and as of the Closing Date (except, in each case, to the extent such representations and warranties speak expressly as of a different date, and then, as of such date) as follows:

(a) Organization and Existence. Purchaser is a corporation, duly organized, validly existing and in good standing under the laws of the state of its jurisdiction, with full power and authority to own, lease, and operate its business and properties and to carry on its business as and where such properties and assets are now owned or leased and such business is now conducted.

(b) Authority and Approval. Purchaser has the power to enter into this Agreement and each of the Related Agreements to which it is to be a party and to perform its obligations thereunder. The execution, delivery, and performance by Purchaser of this Agreement and the Related Agreements to which it is to be a party, and the consummation by Purchaser of the Transactions, have been duly authorized by all required action on the part of Purchaser. This Agreement has been duly executed and delivered by Purchaser and, when executed and delivered by Purchaser, the Related Agreements to which Purchaser is to be a party will have been duly executed and delivered by Purchaser. This Agreement has been duly executed and delivered by Purchaser. This Agreement has been duly executed and delivered by Purchaser, and, assuming the due execution and delivered by Purchaser and the other Related Agreements will be duly executed and delivered by Purchaser, and, assuming the due execution and delivery of this Agreement by Seller and of the Related Agreements by the counterparties thereto, this Agreement constitutes, and the Related Agreements when so executed and delivered will each constitute, a valid and legally binding obligation of Purchaser, enforceable against it in accordance with their respective terms, except as enforceability may be affected by the Enforceability Exception.

(c) No Conflict. The execution and delivery by Purchaser of this Agreement and each of the Related Agreements to which it is to be a party, and Purchaser's compliance with the terms and conditions hereof and thereof, and the consummation by Purchaser of the Transactions, do not and will not (i) conflict with any of, or require any consent of any Person that has not been obtained under, Purchaser's Organizational Documents, (ii) violate any provision of, or require any consent, authorization, or approval under, any Law or any Order applicable to Purchaser, (iii) conflict with, result in a breach of, constitute a default under (whether with or without notice or the lapse of time or both), accelerate or permit the acceleration of the performance required by, or require any consent, authorization, or approval under, any material Contract to which Purchaser is a party or by which it is bound or to which any of its assets or property is subject, or (iv) result in the creation of any Lien upon the assets or property of Purchaser, except in the case of the immediately preceding clauses (ii), (iii) and (iv), as would not reasonably be expected to have a material adverse effect on Purchaser to consummate the Transactions.

(d) Governmental Approvals and Filing. No consent, authorization, approval, or action of, filing with, notice to, or exemption from any Governmental Authority on the part of Purchaser is required in connection with the execution, delivery and performance of this Agreement or any Related Agreements to which Purchaser is to be a party or the consummation of the Transactions, except for any other consent, approval or action where the failure to obtain any such consent, approval, or action, to make any such filing, to give any such notice or obtain any such exemption would not be reasonably expected to (A) have a material adverse effect on Purchaser or (B) materially adversely affect the validity or enforceability against Purchaser of this Agreement or such Related Agreements or materially adversely affect the ability of Purchaser to consummate the Transactions.

(e) No Brokers. No broker, finder, or investment banker is entitled to any brokerage commission, finder's fee, or similar payment in connection with the Transactions based upon arrangements made by or on behalf of Purchaser.

(f) Legal Proceedings. As of the Agreement Date, there are no Legal Proceedings pending, or to the knowledge of Purchaser, threatened, against Purchaser that would reasonably be expected to prevent or delay the ability of Purchaser to enter into and perform its obligations under this Agreement, the Related Agreements, or the Transactions. There is no Order to which Purchaser is subject or that is pending or, to the knowledge of Purchaser, threatened that would reasonably be expected to prevent or delay the ability of Purchaser to enter into and perform its obligations under this Agreement, threatened that would reasonably be expected to prevent or delay the ability of Purchaser to enter into and perform its obligations under this Agreement, the Related Agreements or the Transactions.

(g) Availability of Funds; Solvency. Purchaser will have available sufficient cash to enable it to pay the cash portions of the Closing Consideration as required pursuant to <u>Section 3.2</u> and the cash portions of any Milestone Payments as required pursuant to <u>Section 3.3(a)</u>. Purchaser is solvent and currently: (i) is able to pay its debts as they become due; (ii) owns property that has a fair saleable value greater than the amounts required to pay its debts (including a reasonable estimate of the amount of all contingent liabilities); and (iii) has adequate capital to carry on its business. No transfer of property is being made and no obligation is being incurred in connection with the Transactions with the intent to hinder, delay or defraud either present or future creditors of Purchaser.

## (h) SEC Reports, Financial Statements.

(i) Purchaser has timely filed or furnished each form, report, schedule, registration statement, definitive proxy statement and other document (together with all amendments thereof and supplements thereto) required to be filed or furnished by Purchaser pursuant to the Securities Act or the Exchange Act with the SEC since January 1, 2023 (as such documents have since the time of their filing been amended or supplemented, the "**Purchaser SEC Reports**"). As of their respective dates, after giving effect to any amendments or supplements thereto, the Purchaser SEC Reports (A) complied as to form in all material respects with the requirements of the Securities Act and the Exchange Act, as applicable, and, to the extent applicable, SOX, and (B) did not contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading, *provided, however*, that no representation is made as to the accuracy of any financial projections or forward-looking statements or the completeness of any information filed or furnished by Purchaser to the SEC solely for the purposes of complying with Regulation FD promulgated under the Exchange Act. As of the Agreement Date, there are no material outstanding or unresolved comments in comment letters from the SEC staff with respect to any of the Purchaser SEC Reports.

(ii) The audited financial statements and unaudited interim financial statements (including, in each case, the notes, if any, thereto) included in the Purchaser SEC Reports (the "**Purchaser Financial Statements**") complied in all material respects with the published rules and regulations of the SEC with respect thereto in effect at the time of filing or furnishing the applicable Purchaser SEC Report, were prepared in accordance with GAAP, applied on a consistent basis during the periods involved (except as may be indicated therein or in the notes thereto and except with respect to unaudited statements as permitted by the SEC on Form 8-K, Form 10-Q or any successor or like form under the Exchange Act) and fairly present (subject, in the case of the unaudited interim financial statements, to normal, recurring year end audit adjustments that would not be, individually or in the aggregate, materially adverse to Purchaser) in all material respects the financial position of Purchaser, as of the respective dates thereof, and the consolidated results of their operations and cash flows for the respective periods then ended.

(iii) Purchaser maintains a system of internal control over financial reporting (as defined in Rules 13a–15(f) and 15d–15(f) of the Exchange Act) sufficient to provide reasonable assurances regarding the reliability of financial reporting. Purchaser (A) maintains disclosure controls and procedures (as defined in Rules 13a–15(e) and 15d–15(e) of the Exchange Act) to provide reasonable assurance that all information required to be disclosed by Purchaser in the reports that it files or submits under the Exchange Act is recorded, processed, summarized and reported within the time periods specified in the SEC's rules and forms and is accumulated and communicated to Purchaser's management as appropriate to allow timely decisions regarding required disclosure, and (B) has disclosed, based on its most recent evaluation of internal control over financial reporting, to Purchaser's outside auditors and the audit committee of the board of directors of Purchaser (1) all significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting that are reasonably likely to adversely affect Purchaser's ability to record, process, summarize and report financial information and (2) any fraud, whether or not material, that involves management or other employees who have a significant role in Purchaser's internal control over financial reporting, and the information described in the foregoing clauses (1) and (2) has been disclosed to Seller prior to the Agreement Date. Neither Purchaser nor, to the knowledge of Purchaser, Purchaser's independent registered accountant has received or otherwise had or obtained knowledge of any material complaint, allegation, assertion or claim, whether written or oral, in each case, regarding deficient accounting or auditing practices, procedures, methodologies or methods of Purchaser or its internal accounting controls or any material inaccuracy in the Purchaser Financial Statements.

(i) No Purchaser Vote Required. No vote or other action of the stockholders of Purchaser is required by applicable Law, the certificate of incorporation or bylaws of Purchaser or otherwise in order for Purchaser to consummate the Transactions.

(k) Valid Issuance. The Shares, when issued as provided in and pursuant to the terms of this Agreement, will be duly authorized and validly issued, fully paid and nonassessable, and (other than restrictions under applicable securities Laws or as described in <u>Section 3.5</u>) will be free of restrictions on transfer.

(I) HSR Act. Purchaser (or if different, the Acquiring Person, as defined in the HSR Act regulations) does not satisfy the higher size of person jurisdictional threshold under the HSR Act.

(m) Independent Investigation; Acknowledgements and Confirmations. Purchaser has conducted to its satisfaction its own independent review and analysis of, and based thereon has formed an independent judgment concerning, the Transactions, the Acquired Assets and the Assumed Liabilities. In entering into this Agreement, Purchaser has relied solely upon its own review and analysis and the specific representations and warranties of Seller expressly set forth in Section 5.1 (as qualified or modified by the Disclosure Schedules) and not on any other representations, warranties, statements or omissions (whether by another Person or Seller). Without prejudicing any claim for Fraud, Purchaser acknowledges and confirms that, except for the representations and warranties expressly set forth in Section 5.1 (as qualified or modified by the Disclosure Schedules), none of Seller, its Affiliates or any of their respective Representatives have made, or shall be deemed to have made, and Purchaser has not relied on, is not relying on and hereby disclaims reliance upon, any other representation or warranty, express or implied, at law or in equity, in respect of the Acquired Assets or the Assumed Liabilities, including with respect to (a) merchantability or fitness for any particular purpose, (b) the operation of the business to which the Acquired Assets relate after the Closing, (c) the probable success or profitability of the Product Operations after the Closing or (d) the accuracy or completeness of any Evaluation Material or the appropriateness or suitability of such information for the purposes of enabling Purchaser to evaluate the consummation of the Transactions. Except in the case of Fraud, none of Seller, its Affiliates or any of their respective Representatives will have or be subject to any liability to Purchaser or any other Person resulting from the distribution to Purchaser, its Affiliates or their respective Representatives of, or Purchaser's use of or reliance on, any Evaluation Material in expectation of the Transa

(n) Disclaimer of Warranties. Except as set forth in this Section 5.2 (as qualified or modified by the Disclosure Schedules), none of Purchaser, its Affiliates or any of their respective Representatives have made, or shall be deemed to have made, any other representation or warranty, express or implied, at law or in equity, in respect of the Transactions, and any such other representations or warranties are hereby expressly disclaimed.

## 6. Additional Agreements

**6.1 Public Disclosures.** The initial press release relating to the Transactions shall be a joint press release, the text of which has been agreed to by each of Purchaser and Seller. Each of Purchaser and Seller shall allow the other Party reasonable time to review and comment on such joint press release. Thereafter, no Party shall issue any press release related to the Transactions unless such press release (a) has been approved by the other Party (which approval shall not be unreasonably withheld, conditioned or delayed) or (b) is required to be issued by such Party under applicable Laws based upon advice of counsel. Notwithstanding the foregoing, each Party may make any public statement in response to questions by the press, analysts, investors or those attending industry conferences or financial analyst calls, or issue press releases, so long as any such public statement or press release is consistent with prior public disclosures or public statements approved pursuant to this <u>Section 5.1</u> and which do not reveal nonpublic information about the other Party. If the filing of this Agreement with the SEC or any stock exchange on which securities issued by a Party or its Affiliate are traded is required by applicable Laws, the Parties shall coordinate in advance with each other in connection with such filing and each Party will use reasonable efforts to seek confidential treatment for the terms proposed to be redacted; *provided* that each Party will ultimately retain control over what information to disclose to any Governmental Authority or any stock exchange as the case may be.

**6.2 Confidentiality.** Seller agrees that following the Closing Date it shall hold in strict confidence, unless compelled to disclose by judicial or administrative process or by other Laws (and then only following reasonable prior written notice to Purchaser, to the extent practicable, so that Purchaser shall have an opportunity to object (and Seller shall reasonably cooperate with Purchaser in objecting to any such compulsion)), all confidential information of Purchaser or the Acquired Assets to which they had access prior to the Closing and will not release or disclose such confidential information to any other Person, except to their auditors, attorneys, financial advisors and other consultants, agents and advisors who need to know such information in connection with Seller's business (provided that Seller takes reasonable steps to ensure that each such Person maintain the confidentiality required hereunder); *provided* that the foregoing obligations shall not apply to any such information, or any information that the recipient of such information in connection whom Seller is authorized to release or disclose such information, or any information that the recipient of such information independently develops or discovers after the Closing without reference to the disclosed information or breach hereof.

**6.3 Conduct Pending the Closing.** From the Agreement Date until the earlier of the Closing Date or the termination of this Agreement pursuant to <u>Section 11.1</u>, except (i) as otherwise contemplated, permitted, or required by this Agreement, (ii) as required by Law, (iii) as set forth on Section 6.3 of the Disclosure Schedule or (iv) to the extent that Purchaser otherwise consents in writing, which consent shall not be unreasonably withheld, conditioned or delayed, (x) Seller shall cause the Acquired Assets to be operated in the Ordinary Course of Business in all material respects and (y) Seller shall not:

(a) sell, lease, transfer, or otherwise dispose of any of the Acquired Assets;

(b) grant or suffer to exist any Lien (other than Permitted Liens) on any of the Acquired Assets;

(c) commence any material Legal Proceeding, or settle, pay, discharge, or satisfy any Legal Proceeding, where such commencement, settlement, payment, discharge, or satisfaction would impose any restrictions or limitations upon the Acquired Assets following the Closing;

(d) terminate, extend, or modify any Acquired Business Contract or Material Contract, other than to the extent explicitly contemplated or provided by this Agreement or any Related Agreement, or enter into any Contract in respect of the Acquired Assets or the Assumed Liabilities that, if in effect on the Agreement Date, would be a Material Contract;

(e) waive or release any right of material value, in each case related to any Acquired Assets or any Assumed Liabilities;

(f) fail to keep in force and effect, or allow to lapse, any insurance policy in respect of the Acquired Assets comparable in amount and scope of coverage to that maintained as of the Agreement Date;

(g) correspond, communicate or consult with the FDA or similar Governmental Authority, in each case with respect to the Compounds or the Products, other than (x) any immaterial communication in the Ordinary Course of Business, or (y) any such correspondence, communication or consultation required by applicable Law in connection with an adverse event;

(h) merge, combine or consolidate itself with any other Person or adopt a plan of complete or partial liquidation, dissolution, consolidation, restructuring, recapitalization or other reorganization, or file a certificate of dissolution in respect of Seller or any of its Subsidiaries with the Secretary of State of the State of Delaware;

(i) make any distributions to its stockholders or declare or pay any dividends on shares of Seller's capital stock; or

(j) agree in writing to do any of the foregoing.

## 6.4 Further Assurances and Cooperation; Non-Compete.

(a) Further Assurances. Upon the terms and subject to the conditions set forth in this Agreement, each of the Parties agrees to use its commercially reasonable efforts to take, or cause to be taken, all actions, and to do, or cause to be done, and to assist and cooperate with the other Parties in doing, all things necessary, proper, or advisable to consummate and make effective, in the most expeditious manner practicable, the Transactions, including using commercially reasonable efforts to accomplish the following: (i) the obtaining of all necessary actions or non-actions, waivers, consents, and approvals from Governmental Authorities and the making of all necessary registrations and filings (including filings with Governmental Authorities, if any) and recordings of license grants; (ii) the obtaining of all necessary consents, approvals, or waivers from Third Parties required in accordance with the transfer of any Acquired Asset from Seller to Purchaser pursuant to Section 2.1; (iii) the execution and delivery of any additional instruments necessary to consummate the Transactions, and to fully carry out the purposes of, this Agreement and the Related Agreements; and (iv) the identification and delivery of all Acquired Assets not previously identified and delivered.

(b) Cooperation. If, in order to properly prepare any documents or reports required to be filed with any Governmental Authority, it is necessary that either Purchaser or Seller be furnished with additional information, documents, or records relating to the Product Operations, the Acquired Assets, the Excluded Liabilities, or the Assumed Liabilities, and such information, documents, or records are in the possession or control of the other Party, such other Party will use its commercially reasonable efforts to furnish or make available such information, documents, or records (or copies thereof) at the recipient's reasonable request and at recipient's cost and expense; *provided, further*, that the Parties agree that any Party receiving such materials shall be subject to the same obligations of confidentiality described in <u>Section 6.2</u>.

(c) Non-Compete. Seller agrees that, from the Agreement Date until five (5) years following the Closing Date, it shall not, and shall cause its controlled Affiliates, successors, and assigns to not, directly or indirectly, without Purchaser's prior written consent, (i) Develop, test, manufacture, commercialize, or otherwise exploit any program, pharmaceutical product candidate or product that targets gamma secretase or that is intended to treat desmoid tumors (a "Competing Product"), anywhere in the world, (ii) own or have the right to acquire, or manage, operate or control, any Person engaged in, or otherwise participate in the research, Development, testing, manufacture, Commercialization, or other exploitation of any Competing Product anywhere in the world, or (iii) otherwise knowingly assist or enable any Third Party to research, Develop, test, manufacture, commercialize, or otherwise exploit any Competing Product anywhere in the world. Seller acknowledges that the time, geographic, and scope limitations of its obligations under this Section 6.4(c) are fair and reasonable in all respects. If any term of this provision is invalid or unenforceable in any jurisdiction, it shall not affect the validity or enforceability of the remaining terms and provisions hereof or the validity or enforceability of the offending term or provision in any other situation or in any other jurisdiction. Further, the Parties agree to replace such invalid or unenforceable term or provision with a valid and enforceable term or provision that will achieve, to the extent possible, the economic, business and other purposes of such invalid or unenforceable term. In addition, Seller will not waive or release any employee from any Contract between Seller and its employees that binds its employees to customary non-compete obligations consistent with this Section 6.4(c) and applicable Law.

# 6.5 Tax Matters.

(a) Allocation. Purchaser and Seller acknowledge and agree that the purchase and sale of the Acquired Assets will be treated for all Tax purposes as a taxable asset purchase. The Consideration (including any Assumed Liabilities, to the extent properly taken into account for income Tax purposes) shall be allocated for all Tax purposes consistently with Section 1060 of the Code. Following the Closing Date, Purchaser shall prepare a draft allocation schedule (the "Allocation Schedule"). Purchaser shall provide the draft Allocation Schedule to Seller within 120 days after the Closing Date, and Seller shall have 30 days thereafter to review and comment on such Allocation Schedule. Purchaser shall consider in good faith any such comments of the Seller, and revise the draft Allocation Schedule accordingly to the extent required to comply with Section 1060 of the Code. Each Party agrees to prepare and file all applicable Tax Returns in a manner consistent with the Allocation Schedule.

(b) Straddle Period Taxes. In the case of any real or personal property Taxes (or other Taxes imposed on a periodic basis) (each, a "Straddle Period Tax") attributable to the Acquired Assets that are reported on a Tax Return covering a period beginning on or before the Closing Date and ending thereafter (such period, a "Straddle Period"), any such Straddle Period Taxes shall be prorated between Purchaser and Seller on a per diem basis. The Party required by applicable Law to pay any such Straddle Period Tax (the "Paying Party") shall file the Tax Return related to such Straddle Period Tax within the time period and in the manner required by applicable Law and shall timely pay such Straddle Period Tax. To the extent any such payment exceeds the obligation of the Paying Party hereunder, the Paying Party shall provide the other Party (the "Non-Paying Party") with notice of the amount of such Straddle Period Taxes, and within 10 days of receipt of such notice of payment, the Non-Paying Party shall reimburse the Paying Party for the Non-Paying Party's share of such Straddle Period Taxes.

(c) Transfer Taxes. All transfer, documentary, sales, use, stamp, value added, goods and services, excise, registration and other similar Taxes, and all conveyance fees, recording charges and other fees and charges (including any penalties and interest) incurred in connection with consummation of the Transactions ("Transfer Taxes") shall be borne 50% by Purchaser and 50% by Seller. The Party required under applicable Laws to file a Tax Return with respect to Transfer Taxes will, at its own expense, prepare or cause to be prepared, and file, or cause to be filed, all necessary Tax Returns and other documentation with respect to such Transfer Taxes, and if required by applicable Law, the Parties will, and will cause their Affiliates to, join in the execution of any such Tax Returns. The Parties agree to reasonably cooperate to minimize any Transfer Taxes to the extent permitted by applicable Law.

(d) Cooperation. To the extent relevant to the Product Operations or the Acquired Assets, each Party shall (i) provide the other with such assistance as may reasonably be requested in connection with the preparation of any Tax Return, claim for refund, and the conduct of any audit or other examination by any Governmental Authority or in connection with judicial or administrative proceedings relating to any Liability for Taxes and (ii) provide the other with all records or other information in its possession that is reasonably requested and may be relevant to the preparation of any Tax Returns, claim for refund, or the conduct of any audit, examination, or other proceeding (including any judicial or administrative proceeding) relating to Taxes (each, a "Tax Contest"). Such cooperation shall include obtaining and providing appropriate forms, retaining and providing records and information that are reasonably relevant to any such Tax Return, claim for refund, or Tax Contest, and making employees available on a mutually convenient basis to provide additional information of any materials provide hereunder. Notwithstanding the foregoing, no Party shall be required to provide or provide access to any income Tax Returns or related workpapers, other than Tax Returns relating to the Acquired Assets or Product Operations. For the avoidance of any doubt, this Section 6.5(d) shall only apply to cooperation solely related to the Acquired Assets and the Product Operations, and no party shall be required to provide any information whatsoever otherwise.

#### 6.6 Seller Stockholder Approval.

(a) Immediately following the execution and delivery of this Agreement (i) Seller shall accept any notices of conversion or notices of exercise delivered by any Requisite Holders in respect of outstanding convertible notes or warrants, as applicable, and issue any shares of Seller Common Stock in respect thereof as soon as practicable thereafter, (ii) in lieu of calling a meeting of Seller's stockholders, Seller shall submit the Stockholder Written Consent, in the form attached hereto as **Exhibit F** (the "**Stockholder Written Consent**"), to the Requisite Holders. No later than 11:59 pm Eastern Time on the date that is three (3) Business Days following the Agreement Date (the time that the Stockholder Written Consent is obtained, the "Written Consent Delivery Time"), Seller shall obtain the Stockholder Written Consent, duly executed and delivered by each Consenting Stockholder, in accordance with the Delaware General Corporation Law and Seller's Organizational Documents. The Parties agree and acknowledge that the Stockholder Written Consent shall be void and of no further effect if this Agreement is terminated in accordance with the terms and conditions hereof.

(b) Within ten (10) Business Days after the date of this Agreement, Seller shall file with the SEC a preliminary information statement in accordance with Regulation 14C promulgated under of the Exchange Act (such preliminary information statement and any revised or definitive information statement, the "Information Statement") relating to the Stockholder Written Consent. Purchaser shall reasonably cooperate with Seller in the preparation of the preliminary Information Statement, the definitive Information Statement and any amendments or supplements thereto and shall promptly furnish to Seller the information relating to Purchaser required by the Exchange Act for inclusion therein. Prior to filing with the SEC, Seller shall provide Purchaser and its counsel a reasonable opportunity to review and comment on the Information Statement and shall consider in good faith for inclusion in the Information Statement any comments made by Purchaser or its counsel. Seller shall use reasonable best efforts to respond as promptly as practicable to any comments of the SEC with respect to the Information Statement and to cause the Information Statement in definitive form to be mailed to the holders of shares of Seller Common Stock entitled thereto as promptly as reasonably practicable after (i) the tenth (10th) calendar day after the initial filing of the preliminary Information Statement with the SEC if by such date the SEC has not informed Purchaser that it intends to review the Information Statement or (ii) if the SEC has, by the tenth (10th) calendar day after the filing of the initial preliminary Information Statement with the SEC, informed Seller that it intends to review the Information Statement, the date on which the SEC confirms that it has no further comments on the Information Statement. Seller shall notify Purchaser promptly of (and in any event no more than one (1) Business Day after) the receipt of any comments from the SEC or its staff and of any request by the SEC or its staff for any amendments or supplements to the preliminary Information Statement or the definitive Information Statement, and if required, Seller shall mail to the holders of shares of Seller Common Stock entitled thereto, as promptly as reasonably practicable, such amendment or supplement. Prior to filing with the SEC, Seller shall provide Purchaser and its counsel a reasonable opportunity to review and comment on any such amendments or supplements to the Information Statement and shall reasonably consider in good faith for inclusion in any amendments or supplements any comments made by Purchaser or its counsel. If at any time prior to the Closing any event shall occur, or fact or information shall be discovered, that should be set forth in an amendment or supplement to the Information Statement so that such document would not include any misstatement of a material fact or omit to state any material fact necessary to make the statements therein, in light of the circumstances under which they are made, not misleading, the Party that discovers such information shall as promptly as practicable notify the other Party and Seller shall prepare and file with the SEC such amendment or supplement, in consultation with and subject to review by Purchaser and its counsel as promptly as practicable and, to the extent required by Law, cause such amendment or supplement to be disseminated to the holders of shares of Seller Common Stock entitled thereto. Notwithstanding the foregoing, in the event that this Agreement is terminated in accordance with the terms and conditions hereof, Purchaser shall not be required, after the date of termination, to prepare, file and mail the Information Statement pursuant to this Section 6.6(b).

### 6.7 Non-Solicitation.

(a) From the Agreement Date until the earlier of the Closing Date or the termination of this Agreement pursuant to <u>Section 11.1</u>, Seller shall not, and shall cause its Representatives not to, directly or indirectly: (a) solicit, seek, initiate or knowingly encourage, respond to (other than to communicate that Seller is subject to an exclusivity obligation and not permitted to respond further as a result) or facilitate the making of any inquiry, expression of interest, proposal or offer that constitutes, or is reasonably likely to lead to or encourage the initiation or submission of, any expression of interest, inquiry, proposal or offer from any Person (other than Purchaser) relating to a possible Strategic Transaction; (b) participate in, maintain or continue any discussions or negotiations or enter into any agreement with, provide any non-public information to, or provide access to the properties, books or records of Seller or any of its direct or indirect subsidiaries to any Person (other than Purchaser) relating to or in connection with a possible Strategic Transaction; or (c) agree to, accept, recommend or endorse, or publicly propose or announce any intention or desire to agree to, accept, recommend or endorse, any proposal or offer from any Person (other than Purchaser) relating to a possible Strategic Transaction; *provided, however*, that in each case, Seller shall be permitted to respond to any unsolicited inquiry from any Person (other than Purchaser), solely for the purposes of informing such Person that Seller is subject to an exclusivity obligation and not permitted to respond further as a result.

(b) From the Agreement Date until the earlier of the Closing Date or the termination of this Agreement pursuant to <u>Section 11.1</u>, Seller shall promptly provide Purchaser with an oral and a written description of any expression of interest, inquiry, proposal or offer relating to a possible Strategic Transaction (including the material terms and conditions of any such expression of interest, inquiry, proposal or offer and the identity of the third party) that is received by Seller or by any of Seller's Representatives from any Person (other than Purchaser). From the Agreement Date until the earlier of the Closing Date or the termination of this Agreement pursuant to <u>Section 11.1</u>, Seller shall not release any third party from, or waive any provision of, any confidentiality agreement to which Seller is a party.

(c) It is understood that any violation of the restrictions set forth above by any of the Representatives of Seller shall be deemed to be a breach of this Section 6.7 by Seller.

**6.8 Deposit.** Each of Purchaser and Seller acknowledge and agree that the Deposit previously made by Purchaser pursuant to the Exclusivity Agreement shall be non-refundable, except for the following: (a) if Seller fails to obtain Support Agreements from each of the Requisite Holders; (b) if Seller fails to obtain the fully executed and delivered Stockholder Written Consents in accordance with <u>Section 6.6(a)</u> three (3) Business Days following the Agreement Date; (c) (i) if this Agreement is terminated by Purchaser pursuant to <u>Section 11.1(c)</u> in respect of any material breach or failure respect any of Seller's representations, warranties, covenants, or agreements contained in this Agreement, or (ii) if this Agreement is terminated by Purchaser pursuant to <u>Section 11.1(e)</u> and such termination is due to Seller's failure to obtain an approval, consent or waiver listed on <u>Schedule 4.2(a)(y)</u>; or (d) if Seller or any of its Representatives breach in any material respect the provisions of <u>Section 6.7</u>. Upon the occurrence of any event described in (a) through (c) above, Seller shall refund the Deposit to Purchaser within three (3) Business Days of the date on which Purchaser delivers written notice to Seller demanding repayment of the Deposit.

## 6.9 Restrictions on Dissolution and Winding Up.

(a) For a period of six (6) months following the Closing Date, without Purchaser's prior written consent (which consent shall not be unreasonably withheld, conditioned or delayed), (a) Seller shall not make any distributions to its stockholders or declare or pay any dividends on shares of Seller's capital stock, (b) Seller shall not be wound up, liquidated or dissolved, or initiate any procedures or processes to be wound up, liquidated or dissolved, (c) Seller shall not file or cause to be filed on its behalf a petition in bankruptcy under any provisions of federal or state bankruptcy or insolvency Law or consent to the entry of an order for relief (or any other order with similar effect) under any involuntary bankruptcy proceedings (or any other similar proceedings under state law) commenced against Seller, and (d) Seller shall not seek or cause to be sought on its behalf, nor shall Seller consent to, the appointment of a trustee, receiver, liquidator, assignee for the benefit of creditors or other Person or official with similar duties with respect to the Seller or its assets.

(b) Notwithstanding anything to the contrary in Section 6.9(a), Seller may seek the approval of its stockholders to give the board of directors of Seller the discretion to file, at a time chosen by the board of directors of Seller but no earlier than the date that is six (6) months following the Closing Date, a certificate of dissolution with the Secretary of State of the State of Delaware in accordance with Section 275 of the Delaware General Corporation Law.

**6.10 Tail Insurance**. Prior to the Closing, Seller shall obtain and deliver to Purchaser evidence of an insurance tail policy or similar coverage reasonably satisfactory to Purchaser (the "**Tail Policy**") that (a) covers claims made in respect of product liability or clinical trial liability for activities conducted by Seller prior to Closing and such other coverage for incurred but unreported claims arising on or prior to Closing related to Seller's conduct of its business relating to the Acquired Assets prior to the Closing as requested by Purchaser, (b) has an extended reporting period from the Closing through the fifth (5<sup>th</sup>) anniversary of the Closing Date and (c) is consistent, both in terms of coverage and limits, with the Insurance Policies.

**6.11 Transition Services**. For a period of one (1) month following the Closing, Seller shall provide the transition services set forth on **Schedule 6.11**.

**6.12 Employee Matters**. The Parties acknowledge that employees, advisors, and independent contractors of Seller who have been involved in research and development related to the Products and the Compounds ("**Product Information**"), may have information necessary to enable technology transfer to Purchaser following the Closing (the "**Key Personnel**"). Purchaser anticipates that it may enter into consulting arrangements with some Key Personnel. The Parties acknowledge that Key Personnel may have entered into employment agreements or other written agreements with Seller or its Affiliates, that contain (a) obligations of confidentiality, non-use and non-disclosure with respect to any information related to the Acquired Assets and/or the Assumed Liabilities, or (b) non-compete obligations with respect to the Acquired Assets and/or the Assumed Liabilities ("**Key Personnel** Agreements"). Seller hereby consents and agrees that Purchaser may offer consulting or other arrangements to one or more Key Personnel and Seller agrees that it will not assert that any such consulting arrangement violates the Key Personnel Agreements. Seller hereby waives any rights under any Key Personnel Agreements solely with respect to activities set forth in this <u>Section 6.12</u>.

#### 7. CONDITIONS PRECEDENT TO OBLIGATIONS OF PURCHASER.

The obligation of Purchaser to effect the Closing and consummate the Transactions is subject to the satisfaction (or waiver by Purchaser), at or prior to the Closing, of each of the following conditions:

**7.1 Accuracy of Representations and Warranties.** The Fundamental Representations shall be true and correct in all respects, in each case as of the Agreement Date and as of the Closing Date, with the same effect as though made as of the Closing Date (provided that the accuracy of representations and warranties that by their terms speak as of a specified date will be determined as of such date). All other representations and warranties of Seller set forth in this Agreement (without giving effect to any limitation as to "materiality" or "Material Adverse Effect" or any similar limitation contained in this Agreement) shall be true and correct, in each case as of the Agreement Date and as of the Closing Date (provided that the accuracy of representations and warranties that by their terms speak as of a specified as of the Agreement Date and as of the Closing Date, with the same effect as though made as of the Closing Date (provided that the accuracy of representations and warranties that by their terms speak as of a specified date will be determined as of such date), except where the failure of the representations and warranties of Seller set forth in this Agreement to be so true and correct would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect.

7.2 Performance of Covenants. Seller shall have performed and complied with, in all material respects, all of its covenants contained in this Agreement at or before the Closing (to the extent that such covenants require performance by Seller at or before the Closing).

**7.3 No Restraints.** No temporary restraining order, preliminary or permanent injunction, or other Order preventing the consummation of any of the Transactions shall have been issued by any court of competent jurisdiction and remain in effect, and no material Law shall have been enacted that makes consummation of any of the Transactions illegal.

7.4 No Material Adverse Effect. There shall not have occurred and be continuing a Material Adverse Effect since the Agreement Date.

**7.5** No Governmental Litigation. There shall not be pending or threatened in writing before any court of competent jurisdiction or other Governmental Authority any Legal Proceeding (a) to which a Governmental Authority is a party, and (b) that would or would reasonably be expected to (i) restrain, enjoin, prevent, prohibit or make illegal the consummation of the Transactions, or (ii) prohibit or limit the ownership or operation by Purchaser, its Affiliates of the Acquired Assets or the Assumed Liabilities, or (iii) compel Purchaser or its Affiliates to dispose of, hold separate or license any material portion of the business or assets (including the Acquired Assets) of Purchaser or its Subsidiaries, as a result of the Transactions.

**7.6 Closing Certificate.** Seller shall have delivered to Purchaser a certificate to the effect that each of the conditions specified above in <u>Sections</u> <u>7.1, 7.2</u> and <u>7.4</u> is satisfied in all respects as of the Closing.

7.7 Seller Closing Deliverables. Seller shall have executed and delivered to Purchaser (or caused one or more of its Affiliates to execute and deliver to Purchaser, as applicable) each of the documents and materials contemplated to be delivered by <u>Section 4.2(a)</u>.

**7.8 Seller Information Statement.** At least twenty (20) calendar days shall have elapsed from the date the definitive Information Statement was mailed to Seller's stockholders in accordance with Rule 14c-2 promulgated under the Exchange Act.

# 8. CONDITIONS PRECEDENT TO OBLIGATION OF SELLER

The obligation of Seller to effect the Closing and consummate the Transactions is subject to the satisfaction (or waiver by Seller), at or prior to the Closing, of the following conditions:

**8.1 Accuracy of Representations.** The representations and warranties of Purchaser set forth in this Agreement (without giving effect to any limitation as to "materiality" or "Material Adverse Effect" or any similar limitation contained in this Agreement) shall be true and correct in all material respects, except where the failure of the representations and warranties of Purchaser set forth in this Agreement to be so true and correct would not reasonably be expected to, individually or in the aggregate, prevent or materially delay the consummation of any of the Transactions.

**8.2 Performance of Covenants.** Purchaser shall have performed and complied with, in all material respects, all of its covenants contained in this Agreement at or before the Closing (to the extent that such covenants require performance by Purchaser at or before the Closing).

**8.3** No Restraints. No temporary restraining order, preliminary or permanent injunction, or other Order preventing the consummation of any of the Transactions by Seller shall have been issued by any court of competent jurisdiction and remain in effect, and no material Law shall have been enacted that makes consummation of any of the Transactions illegal.

**8.4 Closing Certificate.** Purchaser shall have delivered to Seller a certificate to the effect that each of the conditions specified above in <u>Sections</u> <u>8.1</u> and <u>8.2</u> is satisfied in all respects.

**8.5 Purchaser Closing Deliverables.** Purchaser shall have executed and delivered to Seller (or caused one or more of its Affiliates to execute and deliver to Seller, as applicable) each of the documents and materials contemplated to be delivered by <u>Section 4.2(b)</u>.

#### 9. SURVIVAL.

**9.1 Survival of Representations; Warranties and Covenants** The Parties, intending to modify any applicable statute of limitations, hereby acknowledge and agree that the representations, warranties, covenants and agreements contained in this Agreement or in any certificate or other document delivered pursuant to this Agreement shall terminate upon the earlier of (a) the Closing and (b) the termination of this Agreement pursuant to <u>Article 10</u>; <u>provided</u>, <u>however</u>, that the covenants and agreements that explicitly contemplate performance after the Closing shall survive the Closing in accordance with their terms until the earlier of (i) the date on which they have been fully performed or expire in accordance with this Agreement and (ii) 60 days following the expiration of any applicable statute of limitations.

# 10. **REGISTRATION RIGHTS.**

10.1 Definitions. For the purpose of this Section 10:

(a) the term "Resale Registration Statement" shall mean any registration statement required to be filed by <u>Section 10.2</u>, and shall include any preliminary prospectus, final prospectus, exhibit or amendment included in or relating to such registration statements; and

(b) the term "**Registrable Shares**" means the Shares; provided, however, that a security shall cease to be a Registrable Share upon the earliest to occur of the following: (i) a Resale Registration Statement registering such security under the Securities Act has been declared or becomes effective and such security has been sold or otherwise transferred by the holder thereof pursuant to and in a manner contemplated by such effective Resale Registration Statement, (ii) such security is sold pursuant to Rule 144 under circumstances in which any legend borne by such security relating to restrictions on transferability thereof, under the Security Act or otherwise, is removed by Purchaser, (iii) the first date such security is eligible to be sold pursuant to Rule 144 without any limitation as to volume of sales, holding period and without the holder complying with any method of sale requirements or notice requirements under Rule 144, or (iv) such security shall cease to be outstanding following its issuance. Notwithstanding the foregoing, no Shares shall be Registrable Shares following the third (3<sup>rd</sup>) anniversary of the date on which the Mandatory Registration Statement is declared effective.

#### **10.2 Registration Procedures.**

(a) Purchaser shall use commercially reasonable efforts to file a Resale Registration Statement (the "Mandatory Registration Statement") with the SEC on or before the date that is seven (7) days following the earlier of (i) April 1, 2024 and (ii) the date Purchaser files its annual report on Form 10-K for the year ended December 31, 2023 (such earlier date, the "Filing Date") to register all of the Registrable Shares on Form S-3 under the Securities Act (providing for shelf registration of such Registrable Shares under SEC Rule 415) (or if Purchaser is not then eligible to use Form S-3, Form S-1); *provided*, that Purchaser's obligation to file a Resale Registration Statement (including the Mandatory Registration Statement) is contingent upon Seller furnishing in writing to Purchaser such information regarding Seller, the securities of Purchaser held by Seller and the intended method of disposition of such Shares, which shall be limited to non-underwritten public offerings and the pro rata distributions of the Shares to the stockholders of Seller in connection with the dissolution of Seller, as shall be requested by Purchaser to effect the registration of such Shares in compliance with applicable securities laws and which information shall be requested by Purchaser from Seller at least five (5) Trading Days prior to the anticipated filing date of the Resale Registration Statement.

(b) Purchaser shall use commercially reasonable efforts to promptly cause such Mandatory Registration Statement to be declared effective as soon as practicable after the filing thereof, but no later than the earlier of (i) 60 calendar days after the filing thereof in the event the SEC reviews and has written comments to the Resale Registration Statement and (ii) the fifth (5th) Trading Day after the date Purchaser is notified (orally or in writing, whichever is earlier) by the SEC that the Resale Registration Statement will not be "reviewed" or will not be subject to further review (the earlier of (i) and (ii), the "Effectiveness Deadline"); *provided*, that if such deadline falls on a Saturday, Sunday or other day that the SEC is closed for business, the Effectiveness Deadline shall be extended to the next Trading Day.

(c) Not less than two (2) Trading Days prior to the filing of a Resale Registration Statement or any related prospectus or any amendment or supplement thereto, Purchaser shall furnish via email to Seller copies of all such documents proposed to be filed (other than any document that is incorporated or deemed to be incorporated by reference therein) for review by Seller. Purchaser shall reflect in each such document when so filed with the SEC such comments regarding Seller and the plan of distribution as Seller may reasonably and promptly propose no later than two (2) Trading Days after Seller has been so furnished with copies of such documents as aforesaid.

(d) Purchaser shall promptly prepare and file with the SEC such amendments and supplements to such Resale Registration Statements and the prospectus used in connection therewith as shall be necessary to keep such Resale Registration Statements continuously effective and free from any material misstatement or omission to state a material fact therein for so long as such Shares remain Registrable Shares, subject to Purchaser's right to suspend such registration pursuant to <u>Section 10.5</u>.

(e) Purchaser shall furnish to Seller such number of copies of prospectuses in conformity with the requirements of the Securities Act as Seller may reasonably request, in order to facilitate the public sale or other disposition of all or any of the Registrable Shares by Seller.

(f) Upon notification by the SEC that that the Resale Registration Statement has been declared effective by the SEC, Purchaser shall, if required, file the final prospectus under Rule 424 within the applicable time period prescribed by Rule 424.

(g) For so long as the Shares remain Registrable Shares, Purchaser shall advise Seller:

(i) within one (1) Trading Day of the effectiveness of the Resale Registration Statement or any post-effective amendments

thereto;

(ii) within five (5) Trading Days of any request by the SEC for amendments to the Resale Registration Statement or amendments to the prospectus or for additional information relating thereto;

(iii) within one (1) Trading Day of the issuance by the SEC of any stop order suspending the effectiveness of the Resale Registration Statement or the initiation of any proceedings for such purpose;

(iv) within one (1) Trading Day of the suspension of the qualification of the Registrable Shares for sale in any jurisdiction, or the initiation of any proceeding for any of the preceding purposes; and

(v) within one (1) Trading Day of the existence of any fact and the happening of any event that makes any statement of a material fact made in the Resale Registration Statement, the prospectus and amendment or supplement thereto, or any document incorporated by reference therein, untrue, or that requires the making of any additions to or changes in the Resale Registration Statement or the prospectus in order to make the statements therein not misleading.

(h) Purchaser shall cause all Registrable Shares to be listed on each securities exchange, if any, on which equity securities by Purchaser are then listed.

(i) Purchaser shall bear all expenses in connection with the procedures in paragraphs (a) through (i) of this <u>Section 10.2</u> and the registration of the Registrable Shares on such Resale Registration Statement.

**10.3 Rule 415; Cutback**. If the SEC prevents Purchaser from including any or all of the Registrable Shares in a Resale Registration Statement due to limitations on the use of Rule 415 under the Securities Act or requires Seller to be named as an "underwriter," Purchaser shall use its commercially reasonable efforts to persuade the SEC that the offering contemplated by the Resale Registration Statement is a valid secondary offering and not an offering "by or on behalf of the issuer" as defined in Rule 415 and that Seller is not an "underwriter." In the event that, despite Purchaser's commercially reasonable efforts and compliance with the terms of this <u>Section 10.3</u>, the SEC refuses to alter its position, Purchaser shall (i) remove from the Resale Registration Statement such portion of the Registrable Shares (the "**Cut Back Shares**") and/or (ii) agree to such restrictions and limitations on the registration and resale of the Registrable Shares as the SEC may require to assure Purchaser's compliance with the requirements of Rule 415 (collectively, the "**SEC Restrictions**"); *provided, however*, that Purchaser shall not agree to name Seller as an "underwriter" in such Registration Statement without the prior written consent of Seller. Seller acknowledges that it shall not have suffered any Losses as to any Cut Back Shares until the date that is five (5) Trading Days following the date that Purchaser is eligible to bring effective the registration of such Cut Back Shares in accordance with any SEC Restrictions (such date, the "**Restriction Termination Date**" of such Cut Back Shares). From and after the Restriction Termination Date applicable to any Cut Back Shares, all of the provisions of this <u>Section 10</u> shall again be applicable to such Cut Back Shares; *provided, however*, that the Filing Deadline for the Resale Registration Statement including such Cut Back Shares shall be ten (10) Trading Days after such Restriction Termination Date, and Purchaser shall use commercially reasonable efforts to cause such Res

#### 10.4 Registration Right Indemnification.

(a) Purchaser agrees to indemnify and hold harmless Seller and its Affiliates and Representatives (each, a "Seller Party" and collectively the "Seller Parties") to the fullest extent permitted by applicable Law, from and against any losses, claims, damages or liabilities (collectively, "Losses") to which they may become subject (under the Securities Act or otherwise) insofar as such Losses (or actions or proceedings in respect thereof) arise out of, or are based upon, any material breach of this Section 10 by Purchaser or any untrue statement or alleged untrue statement of a material fact contained in the Resale Registration Statement or any omission or alleged omission to state therein a material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not misleading or arise out of any failure by Purchaser to fulfill any undertaking included in the Resale Registration Statement and Purchaser will, as incurred, reimburse the Seller Parties for any legal or other expenses reasonably incurred in investigating, defending or preparing to defend any such action, proceeding or claim; provided, however; that Purchaser shall not be liable in any such case to the extent that such Loss arises out of, or is based upon: (i) an untrue statement or omission or alleged untrue statement or omission made in such Resale Registration Statement based on written information furnished to Purchaser by or on behalf of Seller specifically for use in preparation of the Resale Registration Statement; or (ii) any breach of this Section 10 by Seller; provided further, however, that Purchaser shall not be liable to any Seller Party (or any partner, member, officer, director or controlling Person of Seller) to the extent that any such Loss is caused by an untrue statement or omission or alleged untrue statement or omission made in any preliminary prospectus if either (A) (1) Seller failed to send or deliver a copy of the final prospectus with or prior to, or such Seller failed to confirm that a final prospectus was deemed to be delivered prior to (in accordance with Rule 172 of the Securities Act), the delivery of written confirmation of the sale by Seller to the Person asserting the claim from which such Loss resulted and (2) the final prospectus corrected such untrue statement or omission, (B) (1) such untrue statement or omission is corrected in an amendment or supplement to the prospectus and (2) having previously been furnished by or on behalf of Purchaser with copies of the prospectus as so amended or supplemented or notified by Purchaser that such amended or supplemented prospectus has been filed with the SEC, in accordance with Rule 172 of the Securities Act, Seller thereafter fails to deliver such prospectus as so amended or supplemented, with or prior to or Seller fails to confirm that the prospectus as so amended or supplemented was deemed to be delivered prior to (in accordance with Rule 172 of the Securities Act), the delivery of written confirmation of the sale by Seller to the Person asserting the claim from which such Loss resulted or (C) Seller sold Registrable Shares in violation of its covenants contained in Section 3.6.

(b) Seller agrees to indemnify and hold harmless Purchaser and its Affiliates and Representatives (each, a "Purchaser Party" and collectively the "Purchaser Parties") from and against any Losses to which the Purchaser Parties may become subject (under the Securities Act or otherwise), insofar as such Losses (or actions or proceedings in respect thereof) arise out of, or are based upon, any material breach of this Agreement by Seller or untrue statement or alleged untrue statement of a material fact contained in the Resale Registration Statement (or any omission or alleged omission to state therein a material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not misleading), but only to the extent such untrue statement or omission or alleged untrue statement, and Seller will reimburse each Purchaser Party for any legal or other expenses reasonably incurred in investigating, defending or preparing to defend any such action, proceeding or claim; *provided, however*, that in no event shall any indemnity under this <u>Section 10.4(b)</u> be greater in amount than the dollar amount of the net proceeds received by Seller upon its sale of the Registrable Shares included in the Resale Registration Statement giving rise to such indemnification obligation (such amount, the "**Registration Liability Cap**").

(c) Promptly after receipt by any indemnified Person of a notice of a claim or the beginning of any action in respect of which indemnity is to be sought against an indemnifying Person pursuant to this <u>Section 10.4</u>, such indemnified Person shall notify the indemnifying Person in writing of such claim or of the commencement of such action, and, subject to the provisions hereinafter stated, in case any such action shall be brought against an indemnified Person and such indemnifying Person shall have been notified thereof, such indemnifying Person shall be entitled to participate therein, and, to the extent that it shall wish, to assume the defense thereof, with counsel reasonably satisfactory to such indemnifying Person shall not be liable to such indemnified Person for any legal expenses subsequently incurred by such indemnified Person in connection with the defense thereof; *provided, however*, that if there exists or shall exist a conflict of interest that would make it inappropriate, based on the opinion of counsel, for the same counsel to represent both the indemnifying Person and such indemnifying Person; provided, further, that no indemnifying Person shall be responsible for the fees and expense of more than one separate counsel for all indemnified parties. The indemnifying party shall not settle an action without the consent of the indemnified party, which consent shall not be unreasonably withheld, conditioned or delayed; provided that no consent shall be required if such settlement contains an unconditional release of the indemnified party from all liability arising out of such action or claim and does not include a statement as to or an admission of fault, culpability or failure to act by or on behalf of any indemnified party.

(d) If the indemnification provided for in this <u>Section 10.4</u> is held by a court of competent jurisdiction to be unavailable to an indemnified party with respect to any Losses, the indemnifying party, in lieu of indemnifying such indemnified party thereunder, shall to the extent permitted by applicable Law contribute to the amount paid or payable by such indemnified party as a result of such loss, claim, damage or liability in such proportion as is appropriate to reflect the relative fault of the indemnifying party on the one hand and of the indemnified party on the other, as well as any other relevant equitable considerations; *provided*, that in no event shall any contribution by Seller hereunder be greater than the Registration Liability Cap.

(e) For the avoidance of doubt, nothing contained in this Section 10.4 relate to a claim for Losses by an Indemnified Party pursuant to

**10.5 Prospectus Suspension**. Seller acknowledges that there may be times when Purchaser must suspend the use of the prospectus forming a part of the Resale Registration Statement until such time as an amendment to the Resale Registration Statement has been filed by Purchaser and declared effective by the SEC, or until such time as Purchaser has filed an appropriate report with the SEC pursuant to the Exchange Act. Seller hereby covenants that it will not sell any Registrable Shares pursuant to said prospectus during the period commencing at the time at which Purchaser gives Seller notice of the suspension of the use of said prospectus and ending at the time Purchaser gives Seller notice that Seller may thereafter effect sales pursuant to said prospectus; *provided*, (a) that such suspension periods shall in no event exceed (i) on more than three occasions, a period of more than thirty (30) consecutive Trading Days or (ii) more than an aggregate total of sixty (60) Trading Days, in each case in any 360-day period, and (b) the board of directors of Purchaser has reasonably determined that, in order for such Resale Registration Statement not to contain a material misstatement or omission, an amendment thereto would be needed to include information that would at that time not otherwise be required in a current, quarterly or annual report under the Exchange Act.

## 11. TERMINATION

Section 9.

**11.1 Termination of the Agreement.** This Agreement may be terminated, and the Transactions may be abandoned, by written notice delivered by the terminating Party to the other Party (other than in the case of <u>Section 11.1(a)</u>) at any time prior to the Closing:

(a) by the mutual written agreement of Purchaser and Seller;

(b) by either Purchaser or Seller if any Governmental Authority of competent jurisdiction issues an Order permanently restraining, enjoining, or otherwise prohibiting the consummation of the Transactions, and such Order becomes final and non-appealable; *provided*, however, that the right to terminate this Agreement under this <u>Section 11.1(b)</u> shall not be available to a Party whose failure to perform its covenants or agreements contained in this Agreement has been the cause of or has resulted in the imposition of such Order or the failure of such Order to be resisted, resolved, or lifted;

(c) by Purchaser if (i) there exists a breach of any representation or warranty of Seller contained in this Agreement such that the Closing condition set forth in Section 7.1 would not be satisfied or (ii) Seller shall have breached any of the covenants or agreements contained in this Agreement to be complied with by Seller such that the Closing condition set forth in Section 7.2 would not be satisfied; *provided*, that (A) Purchaser shall not be entitled to terminate this Agreement pursuant to this Section 11.1(c) unless, in the case of the immediately preceding clauses (i) or (ii), such breach is not cured by Seller within the shorter of 30 days after Seller receives written notice of such breach from Purchaser or by the Outside Date; and (B) Purchaser shall not be entitled to terminate this Agreement pursuant to this Section 11.1(c) if, at the time of such termination, Purchaser is in breach of any representation, warranty, covenant or other agreement contained in this Agreement in a manner such that the conditions to Closing set forth in Section 8.1 or Section 8.2, as applicable, would not have been satisfied;

(d) by Seller if (i) there exists a breach of any representation or warranty of Purchaser contained in this Agreement such that the Closing condition set forth in Section 8.1 would not be satisfied or (ii) Purchaser shall have breached any of the covenants or agreements contained in this Agreement to be complied with by Purchaser such that the Closing condition set forth in Section 8.2 would not be satisfied; *provided*, that (A) Seller shall not be entitled to terminate this Agreement pursuant to this Section 11.1(d) unless, in the case of the immediately preceding clauses (i) or (ii), such breach is not cured by Purchaser within the shorter of 30 days after Purchaser receives written notice of such breach from Seller or by the Outside Date; and (B) Seller shall not be entitled to terminate this Agreement pursuant to this Section 11.1(d) if, at the time of such termination, Seller is in breach of any representation, warranty, covenant or other agreement contained in this Agreement in a manner such that the conditions to Closing set forth in Section 7.2, as applicable, would not have been satisfied;

(e) by either Party by written notice to the other if the consummation of the Transactions contemplated hereby shall not have occurred on or before the date that is six months following the Agreement Date (the "Outside Date"); *provided, that* the terminating Party shall only be entitled to exercise such right of termination if the terminating Party is not then in breach of any representation, warranty, covenant or other agreement contained in this Agreement such that the conditions to Closing set forth in <u>Article 7</u> or <u>Article 8</u>, as applicable, would not have been satisfied; or

(f) by Purchaser if Seller fails to deliver, by 11:59 p.m. Eastern Time on the date that is three (3) Business Days after Agreement Date, the Stockholder Written Consent executed and delivered by each of the Consenting Stockholders.

**11.2 Effect of Termination.** In the event of termination by either Party pursuant to <u>Section 11.1</u>, written notice thereof will forthwith be given to the other Party and the Transactions will be terminated, without further action by either Party. If the Transactions are terminated as provided herein, this Agreement shall become null and void and have no further force and effect and all obligations of the Parties under this Agreement shall terminate and there shall be no liability of any Party to any other Party, except that nothing herein will relieve or release any Party from liability arising from any breach by such Party of this Agreement prior to any such termination. Notwithstanding the foregoing, the following Sections of this Agreement shall remain in full force and effect following termination of this Agreement: <u>Section 6.1</u>, <u>Section 6.2</u>, <u>Section 6.8</u>, this <u>Section 11.2</u>, and <u>Article 12</u>.

#### 12. MISCELLANEOUS

12.1 Governing Law; Jurisdiction. This Agreement shall be governed by and construed and enforced under the substantive laws of the State of Delaware, excluding any conflicts or choice of law rule or principle that might otherwise make this Agreement subject to the substantive law of another jurisdiction. Purchaser and Seller each hereby irrevocably: (i) consents to submit itself in any suit, action or proceeding arising out of or related to this Agreement or any of the transactions contemplated hereby to the exclusive personal jurisdiction of the Court of Chancery of the State of Delaware (or, only if such court declines to accept jurisdiction over a particular matter, any state or federal court within the New Castle County, Delaware); (ii) agrees that it will not attempt to defeat or deny such personal jurisdiction by motion or other request for leave from such court; and (iii) agrees that it will not bring any action arising out of or related to this Agreement or any of the transactions contemplated hereby in any court other than any such court.

**12.2 Specific Performance.** The Parties agree that irreparable damage for which monetary damages, even if available, would not be an adequate remedy, would occur in the event that the Parties do not perform their obligations under the provisions of this Agreement in accordance with its specified terms or otherwise breach such provisions. The Parties acknowledge and agree that (i) the Parties shall be entitled to an injunction or injunctions, specific performance, or other equitable relief, to prevent breaches of this Agreement and to enforce specifically the terms and provisions hereof in the courts described in <u>Section 12.1</u> without proof of damages or otherwise, this being in addition to any other remedy to which they are entitled under this Agreement and (ii) the right of specific performance is an integral part of the Transactions and without that right, neither Purchaser nor Seller would have entered into this Agreement. Each of the Parties agrees that it will not oppose the granting of an injunction, specific performance and other equitable relief on the basis that the other Parties have an adequate remedy at law or an award of specific performance is not an appropriate remedy for any reason at law or equity. The Parties acknowledge and agree that any Party seeking an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the terms and provisions of this Agreement in accordance with this <u>Section 12.2</u> shall not be required to provide any bond or other security in connection with any such order or injunction.

**12.3 WAIVER OF JURY TRIAL.** EACH OF PURCHASER AND SELLER HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES ALL RIGHT TO TRIAL BY JURY IN ANY ACTION, PROCEEDING OR COUNTERCLAIM (WHETHER BASED ON CONTRACT, TORT OR OTHERWISE) ARISING OUT OF OR RELATING TO THIS AGREEMENT OR THE ACTIONS OF PURCHASER AND SELLER IN THE NEGOTIATION, ADMINISTRATION, PERFORMANCE AND ENFORCEMENT HEREOF.

12.4 Entire Agreement; Severability. This Agreement, together with the Disclosure Schedules, Related Agreements, all Exhibits and Schedules hereto and thereto constitute the entire agreement, and supersede all prior agreements and understandings, both written and oral, between the Parties with respect to the subject matter hereof and thereof. If any term, condition or other provision of this Agreement is found to be invalid, illegal or incapable of being enforced by virtue of any rule of law, public policy or court determination, all other terms, conditions and provisions of this Agreement shall nevertheless remain in full force and effect. If the final determination of any arbitration process or final judgment of a court of competent jurisdiction, in each case, to the extent in accordance with the terms of this Agreement, declares that any term or provision hereof is invalid or unenforceable, the arbitrators or court making the determination of invalidity or unenforceability shall have the power to limit the term or provision, to delete specific words or phrases, or to replace any invalid or unenforceable term or provision with a term or provision that is valid and enforceable and that comes closest to expressing the intention of the invalid or unenforceable term or provision, and this Agreement shall be enforceable as so modified.

12.5 Incorporation by Reference. The Disclosure Schedules, the Schedules and Exhibits, and the documents referenced herein and therein constitute integral parts of this Agreement and are hereby incorporated by reference herein.

12.6 Amendments and Waivers. This Agreement may not be amended or modified, nor may compliance with any condition or covenant set forth herein be waived, except by a writing duly and validly executed by Purchaser and Seller or, in the case of a waiver, the Party waiving compliance. Except as specifically set forth herein to the contrary, no delay or omission by either Party in exercising any right or power occurring upon any noncompliance or default by the other Party with respect to any of the terms of this Agreement shall impair any such right or power or be construed to be a waiver thereof. A waiver by either party of any of the covenants, conditions or agreements to be performed by the other shall not be construed to be a waiver of any succeeding breach thereof or of any other covenant, condition or agreement herein contained.

**12.7 Notices.** Any notice or other communication required or permitted to be delivered to any party under this Agreement shall be in writing and shall be deemed properly delivered, given and received (a) upon receipt when delivered by hand, (b) upon transmission, if sent by electronic transmission (in each case with receipt verified by electronic confirmation) or (c) one (1) Business Day after being sent by courier or express delivery service, *provided* that in each case the notice or other communication is sent to the address set forth beneath the name of such party below (or to such other address as such party shall have specified in a written notice given to the other Parties hereto):

(1) If to Seller, to:

Ayala Pharmaceuticals, Inc. 9 Deer Park Drive, Suite K-1 Monmouth Junction, NJ 08852, USA Attention: Kenneth A. Berlin e-mail: [\*\*\*]

with required copies to:

Morgan, Lewis & Bockius LLP 101 Park Avenue New York, NY 10178 Attention: Robert W. Dickey Email: [\*\*\*]

(2) If to Purchaser, to:

Immunome, Inc. 18912 N. Creek Parkway Bothell, WA 98011 Attention: Clay Siegall, Ph.D. e-mail: [\*\*\*]

with required copies to:

Immunome, Inc. 665 Stockton Drive, Suite 300 Exton, PA 19341 Attention: Sandra Stoneman e-mail: [\*\*\*]

Cooley LLP 10265 Science Center Drive San Diego, California 92121 Attention: Barbara Borden e-mail: [\*\*\*]

**12.8 No Assignment; Binding Effect.** This Agreement is not assignable by any Party without the prior written consent of the other Party; *provided, however*, for the avoidance of doubt, Purchaser may, with notice to Seller but without Seller's consent, (a) at any time, sell, assign, contribute, or otherwise transfer this Agreement and its rights and obligations hereunder in whole or in part to an Affiliate, (b) assign all or any part of its rights or obligations hereunder to any Person (whether or not an Affiliate of Purchaser) in connection with a merger or consolidation of Purchaser or the sale of all or substantially all of Purchaser's operations or assets or substantially all of Purchaser's assets that relate to the Acquired Assets and the Assumed Liabilities, and (c) grant or permit any Lien or assignment to any Person (whether or not an Affiliate of Purchaser to which any rights under this Agreement have been assigned or sublicensed) from time to time, in each case of clauses (a), (b), or (c) above, without Purchaser being relieved of any of its obligations hereunder. This Agreement will be binding upon and will inure to the benefit of the Parties hereto and their respective successors and permitted assigns.

**12.9 Third Person Beneficiaries.** This Agreement shall not benefit or create any right or cause of action in or on behalf of any Person other than the Parties and their respective successors and permitted assigns and nothing herein, whether express or implied, is intended to or shall confer upon any other Person any legal or equitable right, benefit or remedy of any nature whatsoever under or by reason of this Agreement.

12.10 Relationship of the Parties. Nothing contained in this Agreement shall be deemed or construed to create a partnership, joint venture, employment, franchise, agency or fiduciary relationship between Purchaser and Seller. Neither party shall have any express or implied right or authority to assume or create any obligations on behalf of or in the name of the other party or to bind the other party to any contract, agreement or undertaking with any Third Party.

**12.11 Headings; Interpretation.** The headings in this Agreement are intended solely for convenience of reference and will be given no effect in the construction or interpretation of this Agreement. Unless the context otherwise requires, the singular includes the plural, and the plural includes the singular. Whenever the words "include", "includes", or "including" are used in this Agreement, they are deemed to be followed by the words "without limitation". Unless otherwise specified, references in this Agreement to any Article shall include all Sections, subsections, and paragraphs in such Article, references to any Section shall include all subsections and paragraphs in such Section, and references in this Agreement to any subsection shall include all paragraphs in such subsection. The word "or" means "and/or" unless the context dictates otherwise because the subjects of the conjunction are mutually exclusive. The words "herein", "hereof", and "hereunder" and other words of similar import refer to this Agreement as a whole and not to any particular Section or other subdivision. All references to days in this Agreement mean calendar days, unless otherwise specified. Ambiguities and uncertainties in this Agreement, if any, shall not be interpreted against either Party, irrespective of which Party may be deemed to have caused the ambiguity or uncertainty to exist. All references herein to "immediately available funds" or "\$" shall be deemed to be references to the lawful money of the United States. Unless the context otherwise requires, references herein to a statute means such statute as amended from time to time and includes any successor legislation thereto and any regulations promulgated thereunder.

**12.12 Counterparts; Signatures.** This Agreement may be executed in one or more counterparts, all of which will be considered one and the same agreement and will become effective when one or more counterparts have been signed by each of the Parties and delivered to the other Party. This Agreement may be executed by facsimile signature or by an electronic scan delivered by electronic mail.

12.13 Expenses. Except as otherwise expressly provided in this Agreement, whether or not the Transactions are consummated, each Party hereto will pay its own costs and expenses incurred incident to its negotiation and preparation of this Agreement and the Related Agreements and to its performance and compliance with all agreements and conditions contained herein and therein on its part to be performed or complied with, including the fees, expenses and disbursements of its counsel and accountants.

[Signature Page to Follow]

IN WITNESS WHEREOF, the Parties, intending legally to be bound, have caused this Asset Purchase Agreement to be duly executed and delivered as of the Agreement Date.

IMMUNOME, INC.				
By:	/s/ Clay Siegall			
Print Name:	Clay Siegal, Ph.D.			
Title:	President and Chief Executive Officer			

IN WITNESS WHEREOF, the Parties, intending legally to be bound, have caused this Asset Purchase Agreement to be duly executed and delivered as of the Agreement Date.

Ayala Pharmaceuticals, Inc.					
By:	/s/ Kenneth Berlin				
Print Name:	Kenneth A. Berlin				
Title:	President and Chief Executive Officer				

#### <u>Exhibit A</u>

## Support Agreement

[\*\*\*]

[Exhibit A - Form of Support Agreement]

## <u>Exhibit B</u>

## Assignment and Assumption Agreement

[\*\*\*]

[Exhibit B – Assignment and Assumption Agreement]

<u>Exhibit C</u>

Bill of Sale

[\*\*\*]

[Exhibit C – Bill of Sale]

## <u>Exhibit D</u>

## **CVR Agreement**

[\*\*\*]

[Exhibit D – CVR Agreement]

#### <u>Exhibit E</u>

## IP and Patent Assignment Agreement

[\*\*\*]

[Exhibit E – IP and Patent Assignment Agreement]

#### <u>Exhibit F</u>

#### **Stockholder Written Consent**

[\*\*\*]

[Exhibit F – Stockholder Written Consent]

Schedule A

[\*\*\*]

[Schedule A – Support Agreement Signatories]

## <u>Schedule B</u>

## Compounds

[\*\*\*]

[Schedule B – Compounds]

Schedule C

#### **Product Patents**

[\*\*\*]

[Schedule C – Product Patents]

Schedule C-1

## **Dropped Patents**

[\*\*\*]

[Schedule C-1 – Dropped Patents]

## <u>Schedule D</u>

## **Knowledge Parties**

[\*\*\*]

[Schedule D – Knowledge Parties]

## Schedule 2.1(a)(ii)

## **Acquired Business Contracts**

[\*\*\*]

[Schedule 2.1(a)(ii) – Acquired Business Contracts]

## Schedule 2.2(b)

## **Specified Liabilities**

[\*\*\*]

[Schedule 2.2(b) – Assumed Liabilities]

## <u>Schedule E</u>

## Qualified CMC Expenses

[\*\*\*]

[Schedule E – Qualified CMC Expenses]

## Schedule 4.2(a)(v)

## **Closing Third Party Consents/Notices**

[\*\*\*]

[Schedule 4.2(a)(v) – Closing Third Party Consents/Notices]

#### Schedule 6.11

#### **Transition Services**

[\*\*\*]

[Schedule 6.11 – Transition Services]

#### FORM OF SUPPORT AGREEMENT

This SUPPORT AGREEMENT (this "Agreement") is made as of February 5, 2024, by and among AYALA PHARMACEUTICALS, INC., a Delaware corporation (the "Company"), IMMUNOME, INC., a Delaware corporation ("Purchaser"), and the Person or Persons set forth on the signature page hereto ("Stockholder"). Capitalized terms used but not otherwise defined herein shall have the respective meanings ascribed to such terms in the Purchase Agreement.

A. As of the date hereof, Stockholder is the holder of the number of shares of common stock of the Company, par value \$0.001 per share ("*Common Stock*"), set forth on <u>Schedule I</u> (all shares of Common Stock owned by Stockholder, or hereafter issued to or otherwise acquired, whether beneficially or of record, or with respect to which Stockholder otherwise acquires sole or shared voting power (including by proxy), whether by exercise of the Company's warrants or options or otherwise including, without limitation, by gift, succession, in the event of a stock split or as a dividend or distribution of any shares of Common Stock, and including the shares of Common Stock set forth on <u>Schedule I</u>, being referred to herein as the "*Subject Shares*");

**B.** The Company and Purchaser propose to enter into an Asset Purchase Agreement to be dated as of the date hereof (the "*Purchase Agreement*"), which provides, among other things, for the purchase by Purchaser of the Acquired Assets, including the assumption of the Assumed Liabilities, all on the terms and subject to the conditions set forth in the Purchase Agreement; and

C. As a condition to its willingness to enter into the Purchase Agreement, Purchaser has required that Stockholder, and as an inducement and in consideration therefor, Stockholder (in Stockholder's capacity as a holder of the Subject Shares) has agreed to, enter into this Agreement.

**NOW, THEREFORE**, in consideration of the foregoing and the respective representations, warranties, covenants and agreements set forth below and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto, intending to be legally bound, do hereby agree as follows:

#### ARTICLE I VOTING AGREEMENT; TRANSFER RESTRICTIONS; GRANT OF PROXY

Stockholder hereby covenants and agrees that:

1.1. Voting of Subject Shares. No later than 11:59 pm Eastern Time on the date that is three (3) Business Days following the date hereof, Stockholder shall duly execute and deliver (or cause to be duly executed and delivered) to the Company the irrevocable Stockholder Written Consent attached to the Purchase Agreement as Exhibit E, which shall represent Stockholder's vote of all Subject Shares with respect to the matters contemplated thereby. Additionally, from and after the date hereof, at every meeting of the holders of Common Stock (the "Company Stockholders"), however called, and at every adjournment or postponement thereof (or pursuant to any written consent if the Company Stockholders act by written consent in lieu of a meeting, including, for the avoidance of doubt, the Stockholder Written Consent attached to the Purchase Agreement as Exhibit E), Stockholder shall, or shall cause the holder of record on any applicable record date to, be present (in person or by proxy) and to vote or cause to be voted the Subject Shares (a) in favor of adopting and approving the Purchase Agreement, the Transactions, the matters contemplated by the Stockholder Written Consent, and the other actions contemplated by the Purchase Agreement and Related Agreements, (b) against approval of any proposal or agreement that would reasonably be expected to result in the conditions set forth in Section 7 of the Purchase Agreement not to be satisfied on or before the Outside Date, (c) against approval of any proposal made in opposition to, or in competition with, the Purchase Agreement or the consummation of the Transactions, and (d) against the following actions (other than the Transactions): (i) any Strategic Transaction; (ii) any amendment to the Company's certificate of incorporation or bylaws; (iii) any material change in the capitalization of the Company or the Company's corporate structure; (iv) any sale, lease, license or transfer of any Acquired Assets; (v) any change in a majority of the Company's board of directors; and (vi) any other action which would be reasonably likely to impede, interfere with, delay, prevent or adversely affect the Transactions or this Agreement. Except as provided under this Section 1.1 and under Section 1.2 below, Stockholder shall retain at all times the right to vote the Subject Shares in Stockholder's sole discretion and without any other limitation on those matters other than those set forth in this Section 1.1 that are at any time or from time to time presented for consideration to the Company Stockholders.

-1-

1.2. Transfer of Subject Shares; No Inconsistent Arrangements. Except as provided hereunder or under the Purchase Agreement, from and after the date hereof until this Agreement is terminated pursuant to Section 4.2, Stockholder shall not, directly or indirectly, (a) create any Lien other than restrictions imposed by Law or pursuant to this Agreement on any Subject Shares to the extent such Lien would adversely impact the Stockholder's ability to comply with the terms of this Agreement; (b) transfer, sell, assign (directly or indirectly), pledge, exchange, gift, grant, place in trust or otherwise dispose of (including, without limitation, by the creation of a Lien pursuant to clause (a) hereto), or offer to do any of the foregoing (collectively, "Transfer"), or enter into any Contract with respect to any Transfer of, any of the Subject Shares; (d) deposit or permit the deposit of the Subject Shares into a voting trust or enter into a voting agreement or arrangement with respect to the Subject Shares; or (e) take any action that would reasonably be expected to make any representation or warranty of Stockholder contained herein untrue or incorrect or have the effect of restricting Stockholder's legal power, authority and right to vote all of the Subject Shares or would otherwise have the effect of preventing or disabling Stockholder from performing Stockholder's obligations hereunder. Any action taken in violation of the foregoing sentence shall be null and void *ab initio*.

1.3. **Documentation and Information**. Stockholder shall permit and hereby authorizes the Company and Purchaser to publish and disclose in all documents and schedules filed with the SEC and in any press release or other disclosure document that the Company or Purchaser reasonably determines to be necessary or advisable in connection with the Transactions, Stockholder's identity and ownership of the Subject Shares and the nature of Stockholder's commitments and obligations under this Agreement.

1.4. Irrevocable Proxy. Stockholder hereby revokes (or agrees to cause to be revoked) any proxies that Stockholder has heretofore granted with respect to the Subject Shares. Stockholder hereby irrevocably appoints Purchaser as attorney-in-fact and proxy for and on behalf of Stockholder, for and in the name, place and stead of Stockholder, to: (a) attend any and all meetings of the Company Stockholders, (b) vote, express consent or dissent or issue instructions to the record holder to vote the Subject Shares in accordance with the provisions of Section 1.1 at any and all meetings of the Company Stockholders or in connection with any action sought to be taken by written consent of the Company Stockholders without a meeting, including with respect to the Stockholder Written Consent, and (c) grant or withhold, or issue instructions to the record holder to grant or withhold, consistent with the provisions of Section 1.1, all written consents with respect to the Subject Shares at any and all meetings of the Company Stockholders or in connection with any action sought to be taken by written consent of the Company Stockholders without a meeting, including with respect to the Stockholder Written Consent. Purchaser agrees not to exercise the proxy granted herein for any purpose other than the purposes described in this Agreement. The foregoing proxy shall be deemed to be a proxy coupled with an interest, is irrevocable (and as such shall survive and not be affected by the death, incapacity, mental illness or insanity of Stockholder, as applicable) until the termination of this Agreement and shall not be terminated by operation of law or upon the occurrence of any other event other than the termination of this Agreement pursuant to Section 4.2. Stockholder authorizes such attorney-in-fact and proxy to substitute any other Person to act hereunder, to revoke any substitution and to file this proxy and any substitution or revocation with the Secretary of the Company. Stockholder hereby affirms that the proxy set forth in this Section 1.4 is given in connection with and granted in consideration of and as an inducement to Purchaser to enter into the Purchase Agreement and that such proxy is given to secure the obligations of Stockholder under Section 1.1. The proxy set forth in this Section 1.4 is executed and intended to be irrevocable, subject, however, to its automatic termination upon the termination of this Agreement pursuant to Section 4.2. With respect to any Subject Shares that are owned beneficially by Stockholder but are not held of record by Stockholder (other than shares beneficially owned by Stockholder that are held in the name of a bank, broker or nominee), Stockholder shall take all action necessary to cause the record holder of such Subject Shares to grant the irrevocable proxy and take all other actions provided for in this Section 1.4 with respect to such Subject Shares.

1.5. <u>No Ownership Interest</u>. Nothing contained in this Agreement will be deemed to vest in Purchaser any direct or indirect ownership or incidents of ownership of or with respect to any of the Subject Shares. All rights, ownership and economic benefits of and relating to the Subject Shares will remain and belong to Stockholder, and Purchaser will have no authority to manage, direct, superintend, restrict, regulate, govern or administer any of the policies or operations of the Company or exercise any power or authority to direct Stockholder in the voting of any of the Subject Shares, except as otherwise expressly provided herein with respect to the Subject Shares and except as otherwise expressly provided in the Purchase Agreement.

-2-

1.6. <u>Waivers</u>. In connection with the Transactions, Stockholder hereby expressly agrees that Stockholder will not bring, commence, institute, maintain, prosecute, participate in or voluntarily aid any action, claim, suit or cause of action, in law or in equity, in any court or before any Governmental Authority, which (i) challenges the validity of or seeks to enjoin the operation of any provision of this Agreement or the Stockholder Written Consent or (ii) alleges that the execution and delivery of this Agreement or the Stockholder Written Consent by Stockholder, or the approval of the Purchase Agreement by the board of directors of the Company (the "*Company Board*"), breaches any fiduciary duty of the Company Board or any member thereof; *provided* that Stockholder may defend against, contest or settle any such action, claim, suit or cause of action brought against Stockholder that relates solely to Stockholder's capacity as a director, officer or securityholder of the Company.

1.7. No Solicitation of Transactions. Stockholder hereby agrees that Stockholder shall not, directly or indirectly: (a) solicit, seek, initiate or knowingly encourage, respond to (other than to communicate that the Company is subject to an exclusivity obligation and not permitted to respond further as a result) or facilitate the making of any inquiry, expression of interest, proposal or offer that constitutes or is reasonably likely to lead to or encourage the initiation or submission of, any expression of interest, inquiry, proposal or offer from any Person (other than Purchaser) relating to a possible Strategic Transaction; (b) participate in, maintain or continue any discussions or negotiations or enter into any agreement with, provide any non-public information to, or provide access to the properties, books or records of the Company or any of its direct or indirect subsidiaries to any Person (other than Purchaser) relating to or in connection with a Strategic Transaction; (c) agree to, accept, recommend or endorse, any proposal or offer form any Person (other than Purchaser) relating to a possible Strategic Transaction; *provided, however*, that in each case, Stockholder shall be permitted to respond to any unsolicited inquiry form any Person (other than Purchaser), solely for purposes of informing such Person that the Company is subject to an exclusivity obligation and not permitted to respond further as a result.

#### ARTICLE II REPRESENTATIONS AND WARRANTIES OF STOCKHOLDER

Stockholder hereby represents and warrants to Purchaser that:

2.1. **Organization; Authorization; Binding Agreement**. To the extent Stockholder is an entity, Stockholder is duly incorporated or organized, as applicable, validly existing and in good standing under the laws of its jurisdiction of incorporation or organization. Stockholder has full legal capacity and all necessary power, right and authority to execute and deliver this Agreement and to perform Stockholder's obligations hereunder and to consummate the transactions contemplated hereby. The execution and delivery of this Agreement, performance by Stockholder of its obligations hereunder and the consummation of the transactions contemplated hereby by Stockholder have been duly authorized by all necessary action on the part of Stockholder are necessary to authorize this Agreement, or to consummate the transactions contemplated hereby. This Agreement has been duly and validly executed and delivered by Stockholder, and constitutes a legal, valid and binding obligation of Stockholder enforceable against Stockholder in accordance with its terms, subject to bankruptcy, insolvency, fraudulent transfer, reorganization, moratorium or similar Laws of general application affecting or relating to the enforcement of creditors rights generally, and subject to equitable principles of general applicability, whether considered in a proceeding at Law or in equity. If Stockholder is an individual, Stockholder has the legal capacity to execute and deliver this Agreement, to perform Stockholder's obligations hereunder and to consummate the transactions contemplated hereby.

2.2. **Ownership of Subject Shares**; **Total Shares**. Stockholder is the record or beneficial owner of the Subject Shares and has good and marketable title to the Subject Shares free and clear of any Liens (including any restriction on the right to vote the Subject Shares or Transfer any of the Subject Shares), except (a) as provided hereunder, (b) pursuant to any applicable restrictions on transfer under the Securities Act, (c) subject to any risk of forfeiture or repurchase rights of the Company with respect to any Common Stock granted to Stockholder under an employee benefit plan of the Company and (d) as provided in the bylaws of the Company. The Subject Shares listed on <u>Schedule I</u> constitute all of the Common Stock, or rights to acquire Common Stock pursuant to an employee benefit plan of the Company, no Person has any contractual or other right or obligation to purchase or otherwise acquire the any of the Subject Shares. For purposes of this Agreement "*beneficial ownership*" shall be interpreted as defined in Rule 13d-3 under the Exchange Act; *provided* that for purposes of determining beneficial ownership, a Person shall be deemed to be the beneficial owner of any securities that may be acquired by such Person pursuant to any Contract or upon the exercise of conversion rights, exchange rights, warrants or options, or otherwise (irrespective of whether the right to acquire such securities is exercisable immediately or only after the passage of time, including the passage of time in excess of 60 days, the satisfaction of any conditions, the occurrence of any event or any combination of the foregoing).

-3-

2.3. <u>Voting Power</u>. Stockholder has full voting power, with respect to the Subject Shares, and full power of disposition, full power to issue instructions with respect to the matters set forth herein and full power to agree to all of the matters set forth in this Agreement, in each case, with respect to all of the Subject Shares. None of the Subject Shares are subject to any proxy, voting trust or other agreement or arrangement with respect to the voting of the Subject Shares, except as provided hereunder.

2.4. <u>Reliance</u>. Stockholder has had the opportunity to review the Purchase Agreement, including <u>Section 3</u> (Consideration), and this Agreement with counsel of Stockholder's own choosing. Stockholder has had an opportunity to review with its own tax advisors the tax consequences of the Transactions and the transactions contemplated by the Purchase Agreement. Stockholder understands that it must rely solely on its advisors and not on any statements or representations made by Purchaser, the Company or any of their respective agents or representatives. Stockholder understands that such Stockholder (and not Purchaser or the Company) shall be responsible for such Stockholder's tax liability that may arise as a result of the Transactions. Stockholder understands and acknowledges that Purchaser is entering into the Purchase Agreement in reliance upon Stockholder's execution, delivery and performance of this Agreement.

2.5. <u>Absence of Litigation</u>. With respect to Stockholder, as of the date hereof, there is no Action pending against, or, to the knowledge of Stockholder, threatened against, Stockholder or any of Stockholder's properties or assets (including the Subject Shares) that could reasonably be expected to prevent, delay or impair the ability of Stockholder to perform its obligations hereunder or to consummate the transactions contemplated hereby.

2.6. <u>Non-Contravention</u>. The execution and delivery of this Agreement by Stockholder and the performance of the transactions contemplated by this Agreement by Stockholder does not and will not violate, conflict with, or result in a breach of: (a) if the Stockholder is an entity, the organizational documents of Stockholder, (b) any applicable Law or any injunction, judgment, order, decree, ruling, charge, or other restriction of any Governmental Authority to which Stockholder is subject or (c) any Contract to which Stockholder is a party or is bound or to which the Subject Shares are subject, such that it could reasonably be expected to prevent, delay or impair the ability of Stockholder to perform its obligations hereunder or to consummate the transactions contemplated hereby.

#### ARTICLE III REPRESENTATIONS AND WARRANTIES OF PURCHASER AND THE COMPANY

Each of Purchaser and the Company, with respect to itself only, represents and warrants to Stockholder that:

3.1. <u>Organization; Authorization</u>. Each of Purchaser and the Company is a corporation duly incorporated, validly existing and in good standing under the laws of the State of Delaware. The consummation of the transactions contemplated hereby is within Purchaser's and the Company's corporate powers and has been duly authorized by all necessary corporate actions on the part of Purchaser (except, in the case of the Company, for the receipt of the Stockholder Written Consent as contemplated by the Purchase Agreement). Each of Purchaser and the Company has full power and authority to execute, deliver and perform this Agreement.

3.2. **Binding Agreement**. This Agreement has been duly authorized, executed and delivered by each of Purchaser and the Company and constitutes a valid and binding obligation of each of Purchaser and the Company enforceable against each of the Purchaser and the Company in accordance with its terms, subject to the Enforceability Exceptions.

#### ARTICLE IV MISCELLANEOUS

4.1. <u>Notices</u>. All notices, requests and other communications to either party hereunder shall be in writing (including electronic mail) and shall be given, (a) if to Purchaser or the Company, in accordance with the provisions of the Purchase Agreement and (b) if to Stockholder, to the Stockholder's address or electronic mail address set forth on a signature page hereto, or to such other address or electronic mail address as Stockholder may hereafter specify in writing in accordance with this <u>Section 4.1</u> to Purchaser.

#### -4-

4.2. <u>Termination</u>. This Agreement shall terminate automatically, without any notice or other action by any Person, upon the earlier of (a) the termination of the Purchase Agreement in accordance with its terms, (b) the Closing and (c) if, at any time after the date hereof, the Purchase Agreement is amended, without the prior written consent of the Stockholder, in a manner that affects the economic terms of the Purchase Agreement in a manner that is adverse to the Company or its stockholders. Upon termination of this Agreement, neither party shall have any further obligations or liabilities under this Agreement; *provided, however*, that (i) nothing set forth in this Section 4.2 shall relieve either party from liability for any breach of this Agreement prior to termination hereof, and (ii) the provisions of Section 1.6 (Waivers) and this Article IV shall survive any termination of this Agreement.

4.3. <u>Confidentiality</u>. Except to the extent required by applicable Law, Stockholder shall hold any non-public information regarding this Agreement, the Purchase Agreement and the Transactions in strict confidence and shall not divulge any such information to any third person until Purchaser and the Company have publicly disclosed their entry into the Purchase Agreement and Purchaser's entry into this Agreement; <u>provided</u>, <u>however</u>, that Stockholder may disclose such information (a) to its attorneys, accountants, consultants, trustees, beneficiaries and other representatives (<u>provided</u> that such representatives are subject to confidentiality obligations at least as restrictive as those contained herein), and (b) to any Affiliate, partner, member, stockholder, parent or subsidiary of Stockholder, <u>provided</u>, in each case, that Stockholder informs the Person receiving the information that such information is confidential and such Person agrees in writing to abide by the terms of this <u>Section 4.3</u>. Neither Stockholder nor any of its Affiliates (other than the Company, whose actions shall be governed by the Purchase Agreement) shall issue or cause the publication of any press release or other public announcement with respect to this Agreement, the Transactions, the Purchase Agreement or the other transactions contemplated hereby or thereby without the prior written consent of the Company and Purchaser, except as may be required by applicable Law, in which circumstance such announcing party shall use reasonable best efforts to consult with the Company and Purchaser prior to such announcement.

4.4. <u>Amendments and Waivers</u>. This Agreement may not be amended or modified, nor may compliance with any condition or covenant set forth herein be waived, except by a writing duly and validly executed by Purchaser, the Company and Stockholder or, in the case of a waiver, the party waiving compliance. Except as specifically set forth herein to the contrary, no delay or omission by either party in exercising any right or power occurring upon any noncompliance or default by the other party with respect to any of the terms of this Agreement shall impair any such right or power or be construed to be a waiver thereof. A waiver by either party of any of the covenants, conditions or agreements to be performed by the other shall not be construed to be a waiver of any succeeding breach thereof or of any other covenant, condition or agreement herein contained.

4.5. No Assignment; Binding Effect; Third Person Beneficiaries. This Agreement is not assignable by any party without the prior written consent of the other parties hereto; *provided, however*, for the avoidance of doubt, Purchaser may, with notice to Stockholder but without Stockholder's consent, (a) at any time, sell, assign, contribute, or otherwise transfer this Agreement and its rights and obligations hereunder in whole or in part to an Affiliate, (b) assign all or any part of its rights or obligations hereunder to any Person (whether or not an Affiliate of Purchaser) in connection with a merger or consolidation of Purchaser or the sale of all or substantially all of Purchaser's operations or assets or substantially all of Purchaser's assets that relate to the Acquired Assets and the Assumed Liabilities, and (c) grant or permit any Lien or assignment to any Person (whether or not an Affiliate of Purchaser) in connection with a financing for Purchaser (or an Affiliate of Purchaser to which any rights under this Agreement have been assigned or sublicensed) from time to time, in each case of clauses (a), (b), or (c) above, without Purchaser being relieved of any of its obligations hereunder. This Agreement will be binding upon and will inure to the benefit of the parties hereto and their respective successors and permitted assigns. No provision of this Agreement is intended to confer any rights, benefits, remedies, obligations or liabilities hereunder upon any Person other than the parties hereto and their respective successors and permitted assigns.

4.6. **Governing Law; Jurisdiction**. This Agreement shall be governed by and construed and enforced under the substantive laws of the State of Delaware, excluding any conflicts or choice of law rule or principle that might otherwise make this Agreement subject to the substantive law of another jurisdiction. Purchaser, the Company and Stockholder each hereby irrevocably: (a) consents to submit itself in any suit, action or proceeding arising out of or related to this Agreement or any of the transactions contemplated hereby to the exclusive personal jurisdiction of the Court of Chancery of the State of Delaware (or, only if such court declines to accept jurisdiction over a particular matter, any state or federal court within the New Castle County, Delaware); (b) agrees that it will not attempt to defeat or deny such personal jurisdiction by motion or other request for leave from such court; and (c) agrees that it will not bring any action arising out of or related to this Agreement or any of the transactions contemplated hereby in any court other than any such court.

-5-

# 4.7. **WAIVER OF JURY TRIAL.** EACH OF PURCHASER, THE COMPANY AND STOCKHOLDER HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES ALL RIGHT TO TRIAL BY JURY IN ANY ACTION, PROCEEDING OR COUNTERCLAIM (WHETHER BASED ON CONTRACT, TORT OR OTHERWISE) ARISING OUT OF OR RELATING TO THIS AGREEMENT OR THE ACTIONS OF PURCHASER, THE COMPANY AND STOCKHOLDER IN THE NEGOTIATION, ADMINISTRATION, PERFORMANCE AND ENFORCEMENT HEREOF.

4.8. <u>Counterparts</u>. This Agreement may be executed in one or more counterparts, all of which will be considered one and the same agreement and will become effective when one or more counterparts have been signed by each of the parties and delivered to the other party. This Agreement may be executed by facsimile signature or by an electronic scan delivered by electronic mail.

4.9. Entire Agreement. This Agreement, including the schedules and exhibits attached hereto, constitutes the entire agreement and supersedes all prior agreements and understandings, both written and oral, among or between any of the parties with respect to the subject matter hereof and thereof.

4.10. <u>Severability</u>. If any term, condition or other provision of this Agreement is found to be invalid, illegal or incapable of being enforced by virtue of any rule of law, public policy or court determination, all other terms, conditions and provisions of this Agreement shall nevertheless remain in full force and effect. If the final determination of any arbitration process or final judgment of a court of competent jurisdiction, in each case, to the extent in accordance with the terms of this Agreement, declares that any term or provision hereof is invalid or unenforceable, the arbitrators or court making the determination of invalidity or unenforceability shall have the power to limit the term or provision, to delete specific words or phrases, or to replace any invalid or unenforceable term or provision with a term or provision that is valid and enforceable and that comes closest to expressing the intention of the invalid or unenforceable term or provision, and this Agreement shall be enforceable as so modified.

4.11. **Specific Performance**. Any and all remedies herein expressly conferred upon a party will be deemed cumulative with and not exclusive of any other remedy conferred hereby, or by law or equity upon such party, and the exercise by a party of any one remedy will not preclude the exercise of any other remedy. The parties hereto agree that irreparable damage would occur in the event that any of the provisions of this Agreement were not performed in accordance with their specific terms or were otherwise breached. It is accordingly agreed that the parties shall be entitled to an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the terms and provisions hereof in any Delaware Court, this being in addition to any other remedy to which they are entitled at law or in equity, and each of the parties hereto waives any bond, surety or other security that might be required of any other party with respect thereto.

4.12. <u>Headings; Interpretation</u>. The headings in this Agreement are intended solely for convenience of reference and will be given no effect in the construction or interpretation of this Agreement. Unless the context otherwise requires, the singular includes the plural, and the plural includes the singular. Whenever the words "include", "includes", or "including" are used in this Agreement, they are deemed to be followed by the words "without limitation". Unless otherwise specified, references in this Agreement to any Article shall include all Sections, subsections, and paragraphs in such Article, references to any Section shall include all subsections and paragraphs in such Section, and references in this Agreement to any subsection shall include all paragraphs in such subsection. The word "or" means "and/or" unless the context dictates otherwise because the subjects of the conjunction are mutually exclusive. The words "herein", "hereof", and "hereunder" and other words of similar import refer to this Agreement as a whole and not to any particular Section or other subdivision. All references to days in this Agreement mean calendar days, unless otherwise specified. Ambiguities and uncertainties in this Agreement, if any, shall not be interpreted against either party, irrespective of which party may be deemed to have caused the ambiguity or uncertainty to exist. Unless the context otherwise requires, references herein to a statute means such statute as amended from time to time and includes any successor legislation thereto and any regulations promulgated thereunder.

4.13. <u>Further Assurances</u>. Each of the parties hereto will execute and deliver, or cause to be executed and delivered, all further documents and instruments and use their respective reasonable best efforts to take, or cause to be taken, all actions and to do, or cause to be done, all things necessary under applicable Law to perform their respective obligations as expressly set forth under this Agreement.

4.14. <u>Capacity as Stockholder</u>. Stockholder signs this Agreement solely in Stockholder's capacity as a holder of Common Stock, and not in Stockholder's capacity as a director, officer or employee of the Company or in Stockholder's capacity as a trustee or fiduciary of any employee benefit plan or trust. Notwithstanding anything herein to the contrary, nothing herein shall in any way restrict a director or officer of the Company in the exercise of his or her fiduciary duties as a director or officer of the Company or in his or her capacity as a trustee or fiduciary of any employee benefit plan or trust or prevent or be construed to create any obligation on the part of any director or officer of the Company or any trustee or fiduciary of any employee benefit plan or trust from taking any action in his or her capacity as such director, officer, trustee or fiduciary.

4.15. <u>No Agreement Until Executed</u>. Irrespective of negotiations among the parties or the exchanging of drafts of this Agreement, this Agreement shall not constitute or be deemed to evidence a contract, agreement, arrangement or understanding between the parties hereto unless and until (a) the Company board of directors has approved the Transactions, (b) the Purchase Agreement is executed by all parties thereto and (c) this Agreement is executed by all parties hereto.

[Signature pages follow]

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed as of the date first written above.

IMMUNOME, INC.

By: Name: Title:

[Signature Page to Support Agreement]

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed as of the date first written above.

Ayala Pharmaceuticals, Inc.

By:		
Name:		
Title:		

[Signature Page to Support Agreement]

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed as of the date first written above.

#### STOCKHOLDER

(Print Name of Stockholder)

(Signature)

(Name and Title of Signatory, if Signing on Behalf of an Entity)

Address for Notices:

Email:\_\_

[Signature Page to Support Agreement]

<u>Schedule I</u>

**Company Options** 

#### Immunome To Acquire AL102, A Phase 3 Asset for the Treatment of Desmoid Tumors, From Ayala Pharmaceuticals

- AL102 is a small molecule gamma secretase inhibitor with a differentiated clinical profile -

BOTHELL, WA and MONMOUTH JUNCTION, NJ, Feb. 6, 2024 – <u>Immunome</u>, Inc. (Nasdaq: IMNM), a biotechnology company dedicated to developing first-in-class and best-in-class targeted cancer therapies, today announced that it entered into a definitive asset purchase agreement with Ayala Pharmaceuticals, Inc. (OTCQX: ADXS), a clinical-stage oncology company, to acquire AL102 and related drug candidate AL101 from Ayala. Based on the terms of the agreement, Immunome will pay Ayala \$20 million in cash and \$30 million in Immunome common stock (valued at 30-day VWAP as of February 1, 2024) at the closing and will pay up to an additional \$37.5 million in development and commercial milestone payments. Completion of the transaction is subject to customary conditions including Ayala obtaining the requisite stockholder approval.

AL102 is an investigational small molecule gamma secretase inhibitor currently being evaluated in the randomized Phase 3 RINGSIDE international trial for the treatment of desmoid tumors – a debilitating soft tissue malignancy. AL102 is a potential once-daily oral treatment for desmoid tumors. Data from clinical trials have shown AL102 may be more effective in treating desmoid tumors than OGSIVEO<sup>TM</sup> (nirogacestat), which recently became the first treatment approved for desmoid tumors by the U.S. Food and Drug Administration (FDA) in November 2023.

The Phase 2 portion of the RINGSIDE study demonstrated clinically meaningful anti-tumor activity across multiple parameters. The data showed high objective response rates, including 75% of evaluable patients (and 64% of intent-to-treat patients) in the 1.2 mg once daily arm, the dose being evaluated in the current Phase 3 trial. Furthermore, patients receiving the 1.2 mg daily dose experienced a median best reduction in tumor volume of 88% as measured by serial MRI exams, as of the previously reported July 5, 2023 data cut-off, and also demonstrated improvement in other radiographic parameters.

"Immunome is committed to advancing therapies with best-in-class potential. We are especially optimistic about the rapidity and depth of the tumor responses observed in the Phase 2 portion of RINGSIDE," said Clay B. Siegall, Ph.D., President and Chief Executive Officer of Immunome. "AL102 will complement our existing portfolio of targeted cancer agents that are approaching Phase 1 trials. As we complete the work required to advance AL102 to NDA submission, our goal is to bring clinical benefit to an underserved patient population while generating substantial value for stockholders. We also plan to investigate other populations of cancer patients that could benefit from treatment with AL102."

Ken Berlin, CEO of Ayala Pharmaceuticals, stated, "We are pleased that this exciting molecule will be advanced by Immunome and its team of highlyexperienced drug developers which, we believe, maximizes the ability to conclude the RINGSIDE study and bring our lead asset to commercialization. I am proud of the work Ayala has done over the last several years to assemble compelling clinical data and reach this point."

It is estimated that as many as 1,650 people in the U.S. are diagnosed annually with desmoid tumors, also known as aggressive fibromatosis, and the population prevalence is much higher. Desmoid tumors, which occur predominantly in young adults, can cause significant pain and disability. Available non-pharmacologic therapies such as surgery or radiation are often ineffective and can lead to long term complications, and unapproved targeted therapies lack long-term potency. In November 2023, AL102 was granted Orphan Drug Designation by the FDA.

A.G.P./Alliance Global Partners is acting as strategic advisor to Ayala Pharmaceuticals, Inc. in connection with the transaction.

#### About Immunome, Inc.

Immunome is a biotechnology company dedicated to developing first-in-class and best-in-class targeted cancer therapies. Our portfolio pursues each target with a modality appropriate to its biology, including immunotherapies, radioligand therapies and ADCs. We believe that pursuing underexplored targets with appropriate drug modalities leads to transformative therapies. Our proprietary memory B cell hybridoma technology allows for the rapid screening and functional characterization of novel antibodies and targets.

For more information, visit www.immunome.com or follow us on Twitter and LinkedIn.

#### About Ayala Pharmaceuticals, Inc.

Ayala Pharmaceuticals, Inc. is a clinical-stage oncology company primarily focused on developing and commercializing small molecule therapeutics for people living with rare tumors and aggressive cancers. The Company's lead candidates under development are the oral gamma secretase inhibitor, AL102, for desmoid tumors, and aspacytarabine (BST-236), a novel proprietary anti-metabolite for first line treatment in unfit acute myeloid leukemia (AML). AL102 has received Fast Track Designation from the U.S. FDA and is currently in the Phase 3 segment of a pivotal study for patients with desmoid tumors (RINGSIDE). For more information, visit <u>www.ayalapharma.com</u>.

#### **Cautionary Statement Regarding Forward-Looking Statements**

Certain statements contained in this communication regarding matters that are not historical facts, are forward-looking statements within the meaning of Section 21E of the Securities and Exchange Act of 1934, as amended, and the Private Securities Litigation Reform Act of 1995 (the "PSLRA"). We use words such as "may," "could," "potential," "will," "plan," "believe," "goal," "optimistic," and similar expressions to identify these forward-looking statements that are intended to be covered by the safe-harbor provisions of the PSLRA. These forward-looking statements include, but are not limited to, statements regarding Immunome's expectation that the purchase of assets from Ayala will close and, if closed, will complement Immunome's development pipeline, as well as assessments of the clinical efficacy and potential commercial success of the AL102 program; Immunome plans to evaluate other populations of cancer patients that may benefit from AL102; the expected benefits of the Ayala transaction; and other statements regarding management's intentions, plans, beliefs, expectations or forecasts for the future. No forward-looking statement can be guaranteed, and actual results may differ materially from those projected. Such forward-looking statements are based on Immunome's expectations and involve risks and uncertainties; consequently, actual results may differ materially from those expressed or implied in the statements due to a number of factors, including, but not limited to, the risk that the transaction with Ayala will not be completed; potential litigation relating to the proposed transaction that could be instituted against Immunome, Ayala or their respective directors; possible disruptions from the proposed transaction that could harm Immunome's and/or Ayala's respective businesses; Immunome's ability to grow and successfully execute on its business plan, including the development and commercialization of its pipeline; changes in the applicable laws or regulations; the possibility that Immunome may be adversely affected by other economic, business, and/or competitive factors; the risk that regulatory approvals for Immunome's programs and product candidates are not obtained, are delayed or are subject to unanticipated conditions; the risk that pre-clinical data may not be predictive of clinical data; the risk that interim results of a clinical trial do not necessarily predict final results; potential delays in the commencement, enrollment and completion of clinical trials and the reporting of data therefrom; the complexity of numerous regulatory and legal requirements that Immunome needs to comply with to operate its business; the reliance on Immunome's management; the prior experience and successes of the Immunome's management team not being indicative of any future success; uncertainties related to Immunome's capital requirements and Immunome's expected cash runway; the failure to obtain, adequately protect, maintain or enforce Immunome's intellectual property rights; and other risks and uncertainties indicated from time to time described in Immunome's Annual Report on Form 10-K for the year ended December 31, 2022 filed with Securities and Exchange Commission ("SEC") on March 16, 2023, Immunome's Quarterly Report on Form 10-Q for the quarter ended September 30, 2023 filed with the SEC on November 9, 2023, and in Immunome's other filings with the SEC. Immunome cautions that the foregoing list of factors is not exclusive and not to place undue reliance upon any forward-looking statements which speak only as of the date made. Moreover, Immunome operates in a very competitive and rapidly changing environment. New risks emerge from time to time. Except as required by law, Immunome does not undertake any obligation to update publicly any forward-looking statements for any reason after the date of this press release to conform these statements to actual results or to changes in their expectations.

#### **Contacts:**

Immunome: Max Rosett Interim Chief Financial Officer investors@immunome.com

Ayala Pharmaceuticals: +1-857-444-0553 info@ayalapharma.com

Ayala Media: Tim McCarthy LifeSci Advisors, LLC <u>tim@lifesciadvisors.com</u> 917-679-9282